THE ROLE OF INHIBITORY KAPPA B KINASE ALPHA (IKKα) IN THE REGULATION OF NUCLEAR FACTOR KAPPA B SIGNALLING IN ENDOTHELIAL CELLS

By

Ка Но Но

A thesis presented in fulfilment of the requirements for the degree of Doctor of Philosophy

February 2012

Strathclyde Institute for Pharmacy and Biomedical Sciences

Division of Physiology and Pharmacology

University of Strathclyde, Glasgow, UK

"This thesis is the result of the author's original research. It has been composed by the author and has not been previously submitted for examination which has led to the award of a degree.

The copyright of this thesis belongs to the author under the terms of the United Kingdom Copyright Acts as qualified by University of Strathclyde Regulation 3.50. Due acknowledgment must always be made use of any material contained in, or derived from, this thesis."

Signed:

Date:

ABSTRACT

Atherosclerosis is one of the most common cardiovascular diseases in Western Countries. An important feature of the disease is the initiation of atherosclerotic lesions including monocyte recruitment into the vessel wall (Cominacini et al., 2000). This involves the expression a number of adhesion molecules including E-selectin, ICAM-1 and VCAM-1. In endothelial cells, ICAM-1 and VCAM-1 expression is regulated by a number of defined signalling pathways including the transcription factor Nuclear factor Kappa B (NF κ B). NF κ B activation is regulated by α and β isoforms of the Inhibitory Kappa B kinases (IKK). Whilst IKK β regulates the canonical pathway involving phosphorylation and degradation of I κ B- α and the translocation of p65 NF κ B to the nucleus, accumulated evidence implicates that IKK α regulates p65 dependent gene expressions and plays a distinctive role in non-canonical NF κ B pathway by mediating processing of p100 NF κ B precursor p100. In this PhD thesis, I have examined the effect of adenoviral (Adv.) mediated IKK α and β blockade or siRNA IKK α on NF κ B activation and adhesion molecule expression in endothelial cells.

In human umbilical vein endothelial cell, over-expression of Adv. DN-IKK β attenuated TNF- α induced I κ B- α degradation and p65 phosphorylation. In contrast, Adv. DN-IKK α had no inhibitory effect of either responses. Furthermore, either Adv. DN-IKK α or siRNA IKK α reduced TNF- α induced NF κ B reporter activity and also p65-dependent expression of the adhesion molecules, ICAM-1 and VCAM-1. Taken together, these findings suggest IKK α may regulate NF κ B activity and gene expression without being involved in the up-stream canonical NF κ B pathway. Moreover, blocking IKK α but not β inhibited TNF- α induced activation of the non-canonical pathway exemplified by p100 processing and the formation of p52. Novel selective IKK α inhibitors, SU1007 and SU1010, generated in house, were employed in the study to confirm the role of IKK α in endothelial cells. Similarly, both SU compounds did not affect TNF- α induced I κ B- α degradation and phosphorylation of p65 but inhibited NF κ B reporter activity and ICAM-1 and VCAM-1 expression. In addition, SU 1007 and 1010 also showed an inhibitory effect on the non-canonical pathway by blocking both p100 processing and the expression of p100 itself.

These findings indicate IKK α plays a key role in the regulation of the non-canonical NF κ B pathway and regulates the canonical NF κ B pathway via its nuclear activity. These data

suggest that IKK α inhibition may be a useful approach for the future treatment of cardiovascular disease.

PUBLICATIONS

Original Research Articles

Ka Ho Ho, Rebecca Scott, Sophia Hatziieremia, Kathryn McIntosh, Louise Young, Alan Harvey, Judith Huggan, Nahoum Anthony, Stuart Lang, David Breen, Murray Robertson, Oliver Sutcliffe, Simon Mackay, Colin Suckling, Joanne Edwards, Andrew Paul and Robin Plevin* (2012). Novel compounds with selectivity for inhibitory Kappa B kinase 1 regulate inflammatory gene expression in human endothelial cells. (In Preparation)

Gamble, C., McIntosh, K., Scott, R., **Ho, K. H**., Plevin, R. & Paul, A. 2012. Inhibitory kappa B kinases as targets for pharmacological regulation. Br J Pharmacol, 165, 802-19.

Poster Presentation

Ka Ho Ho, Andrew Paul and Robin Plevin (2011). The Role Of Inhibitory Kappa B Kinase Alpha (IKK α) In Regulation Of Human Endothelial Cell Signalling and Adhesion Molecule Expression. The 6th European Meeting for Vascular Biology and Medicine, Krakow, Poland.

Ka Ho Ho, Andrew Paul and Robin Plevin (2010). The Role Of Inhibitory Kappa B Kinases In The Regulation Of Human Endothelial Cells Function. First BHF-sponsored Summer School in Vascular Biology, Bristol, UK.

Ka Ho Ho, Andrew Paul and Robin Plevin (2010). The Role Of Inhibitory Kappa B Kinases In The Regulation Of Human Endothelial Cells Function. University of Strathclyde Research Day, University of Strathclyde, Glasgow, UK.

ACKNOWLEDGEMENTS

Indeed, PHD degree is a difficult degree which is always implicated with a lot of stress and over-time working, but there are also joys and excitement too. I would like to thank my supervisors Prof Robin Plevin and Dr Andy Paul who offered me an opportunity to do a PhD in University of Strathclyde. Prof Plevin is probably the most patient and professional teacher I ever have in my life. Not only he was not be mad about me when making too much 2% BSA in my first western blot experiment, he also like to spend time for students for discussion the project direction and also presentation and poster preparation. The labortory is organised and I think the weekly lab meeting is useful for improving my presentation skill.

The project eventually went successful, here I would like to say thank you to Robin who guided me to a right path of the project and taught me to become a professional scientist. I would also like to say thank you to my second supervisor Dr Andy Paul who always gave me useful advices to the project, so that I have become more confident and organised under their supervision.

I also want to say thank you to everyone who shared the hard time and good time with me in the lab. Katy and Margret gave me lots of advices when I first start my PHD. Their professional and enthusiastic attitude in science showed me a good model of being a PHD student and post doc. I will never forget about the boys: Sameer, Muhannad and Ahmed. They spent most of their time in lab working hard, but they also managed to make jokes and fooling around. Thanks to them, the lab was full of shouting, laughing, singing and joys.

Thank you Juliane and Carly who always gave me advices of my project and also the future career plans. Thank you Emma who is the only one who discovered my dancing talent –my acrylamide dance. Furthermore, going to the conference meeting in Poland with Robin and Emma was fun and exciting. We managed to go hiking and even got loss without missing the conference meeting introduction. In addition, I really appreciated that Thikryat, Sulaiman, Emma and Norsyahida who shared the lab duty with us and made the laboratory organised and tidy. For my last two years PHD study, Yuen has helped me a lot in laboratory works and sharing advices and tons of ideas with me. I would like to say thank for her kindness and the lab life is much easier with her help. I especially want to thank Katy and Tamara who spent

their time for going through the thesis with me who gave me many good advices and useful corrections. Also, thank you Tamara, it was nice to work together during the long horrible writing period.

Thank you to my parents and sister who allowed me to call them whenever I feel like to talk to, I cannot count how many times they were waken up in the middle of the night to listen me whining about works.

In the end, I would like to say thank you to you all once again, I had a wonderful time in this three years PhD period.

P.S. Thanks to the hair dresser who styling my hair in the last three year, it was nice and spiky.

ABBREVIATIONS

Adv.	Adenovirus
ANOVA	Analysis of variance
AKT	Serine/threonine-protein kinase
APS	Ammonium persulphate
Ang-II	Angiotensin-II
ATF-2	Activating transcription factor
BAFF	B cell activation factor
BAD	Bcl-2 assoicated death promoter
BLC	B lymphocyte chemoattractant
BSA	Bovine serum albumin
Bcl-2	B-cell lymphoma-2
BMK1	Big MAPK-1
CBP	CRE binding protein
CCL	C-C motif ligands
CHIP	Chromatin Immunoprecipitation
CXCL-12	Chemokine (C-X-C motif) ligand
cGMP	Cyclin guanosine monophosphate
CDK4	Cyclin-dependent kinase 4
COX	Cyclooxygenase
DN	Dominant negative
DTT	Dithiothreitol
eNOS	Endothelial nitric oxide synthase

ECL	Enhanced Chemiluminescence
ELC	EBI-1-ligand chemokine
EGF	Epidermal growth factor
EGFR	Epidermal growth factor receptor
Elk-1	ETS-like transcription factor 1
ERK	Extracellular regulated kinase
ET-1	Endothelin-1
FCS	Foetal calf serum
FGF	Fibroblast growth factor
GPCRs	G-protein coupled receptors
GSK3	Glycogen synthase kinase-3
H_2O_2	Hydrogen peroxide
HEK	Human embryonic kidney
HLH	Helix-loop-helix
HUVEC	Human umbilical vein endothelial cells
ICAM-1	Intracellular adhesion molecule-1
IL-1β	Interleukin-1 beta
IL-1R	Interleukin-1 receptor
ΙκΒ-α	Inhibitory kappa B alpha
IKK	Inhibitory kappa B kinase
JNK	c-Jun N-terminal kinase
kDa	kilo-Dalton
KDIF	Keratinocyte differentiation-inducing factor

LDL	Low density lipoprotein
Luc	Luciferase
LTα1β2	Lymphotoxin alpha 1 beta 2
MMP	Matrix metalloproteinases
МАРК	Mitogen activated protein kinase
МАРКК	Mitogen activated protein kinase kinase
МАРККК	Mitogen activated protein kinase kinase kinase
MEK	MAP (Mitogen-Activated Protein) Kinase/ERK (Extracellular Signal-Regulated Kinase) Kinase
MEF	Mouse Embryonic Fibroblast
ΝFκB	Nuclear factor kappa B
NEMO	NFκB essential modulator
NIK	NFκB inducing kinase
NLS	Nuclear localisation sequence
NO	Nitric oxide
PAGE	Polyacrylamide gel electrophoresis
PBS	Phosphate buffered saline
PDTC	Pyrrolidine dithiocarbamate
PMA	Phorbol 12-myristate 13-acetate
PP2A	Protein phosphatase 2A
PTEN	Protein Tyrosine Phosphatase
PI3K	Phosphoinositide-3 kinase
PIP ₂	Phosphatidylinositol 4,5-bisphosphate

PIP ₃	Phosphatidylinositol-3,4,5-trisphosphate
PGI ₂	Prostacyclin
РКС	Protein kinase C
SCC	Squamous cell carcinomas
SLC	Secondary lymphoid tissue chemokine
SDS	Sodium dodecyl sulphate
SP-1	Specific protein-1
siRNA	Small interfering RNA
shRNA	Short hairpin interfering RNA
RANK	Receptor activator of NFkB
TAB-2	TAK-1 binding protein 2
TAK-1	Transforming growth factor β -activated kinase 1
TEMED	N,N,N',N'-tetramethylenediamine
TGFR	Transforming growth factor β recepttr
TLR	Toll like receptor
TNF-α	Tumour necrosis factor alpha
TNFR	TNF-receptor
TRAF	TNF receptor-associated factor
UBF	Upstream binding factor
VCAM-1	Vascular cell adhesion molecule 1
VEGF	Vascular endothelial growth factor
XIAP	X-linked inhibitor of apoptosis protein

Content

Chapter 1 General Introduction

1.1	Physiology of Endothelium	2
1.2	Dysfunction of Endothelium	3
1.3	Signalling mechanisms in Endothelium	4
1.3.1	Mitogen-Activated Protein Kinase (MAPK) Signalling Pathway	5
1.3.2	The Phosphoinositide-3 Kinase Signalling Pathway	8
1.3.3	The Nuclear Factor Kappa B Signalling Pathway	11
1.3.3.1	Structure of NFkB	11
1.3.3.4	Genes Regulated by NFkB	14
1.3.3.5	Introduction to the NFkB Signalling Pathway	15
1.3.3.6	Structure of IkB	15
1.3.3.6.1	Inhibitory Kappa B-alpha (ΙκΒ-α)	18
1.3.3.6.2	Inhibitory Kappa B NFκB2 (p100/p52)	18
1.3.3.7	Identification of Inhibitory Kappa B kinase	19
1.3.3.8	The Canonical (classical) NFκB Pathway	20
1.3.3.8.1	The role of IKK β in the regulation of the canonical NF κ B Pathway	21
1.3.3.8.2	The role of IKK α in the regulation of the canonical NF κ B Pathway	23
1.3.3.8.3	The role of IKK γ in the regulation of the canonical NF κ B Pathway	25
1.3.3.9	The up-stream activation of the Non-Canonical NFKB Pathway	26
1.3.4	The Role of IKK α in the regulation of the Non-Canonical	

	NFκB Pathway	27
1.4	The Role of the Non-Canonical NF κ B Pathway in the regulation	
	of inflammation	28
1.5	NFκB and inflammatory diseases	30
1.6	Atherosclerosis	32
1.6.1	Development of atherosclerosis	32
1.6.2	Adhesion molecules and the development of atherosclerosis	33
1.6.3	NFκB and atherosclerosis	34
1.6.4	Oxidised LDL and NFKB	36
1.6.5	Atherosclerosis development and endothelial dysfunction and apoptosis	38
1.7	Angiogenesis	39
1.7.1	Introduction to Angiogenesis	39
1.7.2	NFkB and Angiogenesis	39
1.7.3	The Canonical NFkB Pathway and Angiogenesis	40
1.7.4	The Non-Canonical NFKB Pathway and Angiogenesis	41
1.8	The development and use of IKK inhibitors	43
1.9	Aims and Objectives	45

Chapter 2 Materials and Methods

2.1	Materials	47
2.1.1	General Reagents	47
2.1.2	Adenoviruses	48
2.1.3	SU Compounds	49
2.1.4	Microscopy	49
2.1.5	Antibodies	49
2.2	Cell Culture	50
2.2.1	Human Umbilical Vein Endothelial Cells (HUVECs)	51
2.2.2	Subculturing HUVECs by Trypsinisation	51
2.2.3	Human Embryonic Kidney (HEK293)	51
2.3	Western Blot Analysis	52
2.3.1	Preparation of Whole Cell Extracts	52
2.3.2	SDS-Polyacrylamide Gel Electrophoresis (SDS-PAGE)	52
2.3.3	Electrophoretic Transfer of Proteins to Nitrocellular Membrane	53
2.3.4	Immunological Detection of Protein	53
2.3.5	Nitrocellulose membrane stripping and reprobing	53
2.4	Preparation of Nuclear Extracts	54
2.4.1	Determination of Protein Concentration in the Nuclear Extracts	54
2.5	Luciferase reporter activity assay	55
2.6	Immunofluorescent Microscopy	55
2.6.1	Transfection of YFP-tagged IKK α or YFP-tagged p65 DNA	55

2.6.2	Direct Immunofluorescent Microscopy	55
2.7	Preparation of recombinant adenovirus	56
2.7.1	Crude Adenovirus Lysate	56
2.7.2	Generation and purification of High-Titre Stocks of adenovirus	56
2.7.3	Titration of adenovirus by End-Point Dilution	57
2.7.4	Infecting HUVECs with adenovirus	60
2.8	Scanning Densitometry	60
2.9	Data Analysis	60

Chapter 3Characterisation the role of IKKα in the regulation of
Canonical and Non-Canonical NFκB pathways

3.1	Introduction	62
3.2	Characterisation of nuclear IKKa in HUVECs	62
3.2.1	Subcellular fractionation to determine the nuclear translocation of IKK α	
	and p65	63
3.3	Comparison of knockdown efficiency for Adv. shRNA IKK α	
	and siRNA IKKα	66
3.3.1	TNF- α induces I κ B- α loss in HUVECs in a time dependent manner	68
3.3.2	The effect of Adv. DN-IKK β in the regulation of TNF- α	
	induced IκB-α degradation in HUVECs	68
3.3.3	Characterisation of the role of IKK α in the regulation of TNF- α	
	induced I κ B- α degradation in HUVECs	68

3.3.5	The effect of Adv. DN-IKK α in the regulation of TNF- α	
	induced p65 phosphorylation in HUVECs	74
3.3.6	Lack of the effect of Adv. DN-IKK α in IL-1 β	
	activated canonical NFkB pathway in HUVECs	76
3.3.7	Characterisation of the effect of IL-1 β on the canonical NF κ B	
	pathway in HUVECs	76
3.3.8	The effect of Adv. DN-IKK α and β on IL-1 β induced NF κB	
	activation in HUVECs	76
3.4	The role of IKK α in the regulation of the non-canonical NF κB	
	pathway	79
3.4.1	Characterisation of the effect of TNF- α on the non-canonical	
	NFκB pathway in HUVECs	79
3.4.2	Characterisation of the effect of Adv. DN-IKK α in the non-	
	canonical NFkB pathway in HUVECs	80
3.4.3	Characterisation of the effect of siRNA IKK α on the non-	
	canonical NFkB pathway in HUVECs	80
3.5	The role of IKK α in the regulation of NF κ B transcriptional	
	activity	88
3.5.1	TNF- α induces NF κ B reporter activity in HUVECs	88
3.5.2	Characterisation of the role of IKK α in the regulation of	
	TNF- α induced NF κ B reporter activity	88
3.5.3	Characterisation of the role of IKK α in the regulation of	
	IL-1β induced NFκB reporter activity	89

3.6	The role of IKK α in the regulation of ICAM-1 and VCAM-1	
	expression in HUVECs	93
3.6.1	TNF- α induced ICAM-1 and VCAM-1 expression in HUVECs	
	in concentration dependent manner	93
3.6.2	Characterisation of the role of IKK β in the regulation of	
	TNF- α induced ICAM-1 and VCAM-1 expression in HUVECs	93
3.6.3	Characterisation of the role of IKK α in the regulation of	
	TNF- α induced ICAM-1 and VCAM-1 expression in HUVECs	94
3.6.4	Characterisation of the effect of IL-1 β on both ICAM-1 and	
	VCAM-1 expression in HUVECs	102
3.6.5	Characterisation of the role of IKK β on IL-1 β induced ICAM-1	
	and VCAM-1 expression in HUVECs	102
3.6.6	Characterisation of the role of IKK α on IL-1 β induced ICAM-1	
	and VCAM-1 expression in HUVECs	103
3.7	Discussion	106

Chapter 4 Characterisation of the inhibitory effect of IKKα selective inhibitors (SU compounds) on Canonical and Non-Canonical NFκB pathways

4.1	Introduction	117
4.2	Determining the dissociation constant (Ki value) of the SU compounds	118
4.3	Characterisation of SU compounds on TNF- α or IL-1 β induced NF κ B	
	Canonical pathway in HUVECs	120
4.3.1	Characterisation of the effect of SU compounds on TNF- α induced	
	I κ B- α degradation in HUVECs	120
4.3.2	Characterisation of the effect of SU compounds on TNF- α induced	
	p65 phosphorylation in HUVECs	126
4.3.3	Characterisation of the effect of SU compounds in the regulation of	
	IL-1 β induced I κ B- α or p65 phosphorylation in HUVECs	129
4.4	Characterisation of SU compounds on TNF- α induced p100 processing	
	in HUVECs	131
4.5	Characterisation of the effect of SU compounds in the regulation of TNF	'-α
	induced NF κ B reporter gene activity in HUVECs	141
4.6	Characterisation of SU1007 and SU1010 on TNF- α induced ICAM-1 and	d
	VCAM-1 expression in HUVECs	143
4.7	Characterisation of SU compounds in combination with Adv.DN-IKK β	on
	TNF-α induced ICAM-1 expression on HUVECs	151

4.8	Characterisation of SU compounds on IL-1 β induced ICAM-1 and VCAM-1		
	in HUVECs	151	
4.9	Discussion	154	

Chapter 5 General Discussion

160

List of Figures

Chapter 1 General Introduction

Figure 1.1	Schematic representation of the MAPK cascades and their		
	nuclear targets	7	
Figure 1.2	The PI3K/PTEN/AKT signalling pathway	10	
Figure 1.3	NFκB subunits	13	
Figure 1.4	Schematic diagram of NFkB pathways	17	
Figure 1.5	Cellular mechanisms regulating atherosclerosis	37	
Figure 1.6	The role of the NF κ B pathways in the regulation of angiogenesis	42	

Chapter 2 Materials and Methods

Figure 2.1	The End-Point Dilution for adenovirus titration	58
Figure 2.2	The calculation of the titre of the adenovirus	59

Chapter 3Characterisation the role of IKKα in the regulation of
Canonical and Non-Canonical NFκB pathways

Figure 3.1	TNF- α induced nuclear translocation of IKK α in HUVECs using	
	sub-cellular fractionation	64
Figure 3.2	TNF- α induced nuclear translocation of IKK α in HUVECs	
	assessed by immunofluorescence	65

Figure 3.3	The effect of Adv. shRNA IKK α and siRNA IKK α on IKK α expression in		
	HUVECs	67	
Figure 3.4	TNF- α -mediated I κ B- α loss in HUVECs	70	
Figure 3.5	The effect of Adv DN-IKK β upon TNF- α -mediated I κ B- α loss in		
	HUVECs	71	
Figure 3.6	The effect of Adv. DN-IKK α upon TNF- α -mediated I κ B- α loss in		
	HUVECs	72	
Figure 3.7	The effect of Adv. DN-IKK α upon TNF- α -mediated I κ B- α degradation		
	in HUVECs	73	
Figure 3.8	The effect of Adv. DN-IKK α or β upon TNF- α -mediated p65		
	phosphorylation in HUVECs	75	
Figure 3.9	Time course and concentration dependency of IL-1 β mediated I κ B- α		
	degradation and p65 phosphorylation	77	
Figure 3.10	The effect of Adv. DN-IKK α upon IL-1 β -mediated I κ B- α degradation		
	and p65 phosphorylation in HUVECs	78	
Figure 3.11	Time Course of TNF- α -mediated p100 processing in HUVECs	81	
Figure 3.12	Concentration dependency of TNF- α -mediated p100 processing in		
	HUVECs	82	
Figure 3.13	Time Course of IL-1 β -mediated p100 processing in HUVECs	83	
Figure 3.14	The effect of Adv. DN-IKK α or β upon TNF- α -mediated p100 processing	g in	
	HUVECs	85	
Figure 3.15	The effect of Adv. DN-IKK β and siRNA IKK α upon TNF- α -mediated		
	p100 processing in HUVECs	87	

× / × /	
XX	I
$\Lambda\Lambda$	L

Figure 3.16	Time Course and concentration dependency of TNF- α or IL-1 β -mediated	
	NFκB reporter activity in HUVECs	91
Figure 3.17	The effect of Adv. DN-IKK α on NF κ B luciferase reporter activity mediate	ed
	by TNF- α and IL-1 β in HUVECs	92
Figure 3.18	Concentration dependence of TNF- α -stimulated induction of ICAM-1 and	1
	VCAM-1 expression in HUVECs	96
Figure 3.19	The effect of Adv. DN-IKK β upon TNF- α -mediated ICAM-1 and VCAM	[-1
	expression in HUVECs	97
Figure 3.20	The effect of Adv. DN-IKK α upon TNF- α -mediated ICAM-1 and VCAM	[-1
	expression in HUVECs	98
Figure 3.21	The effect of Adv. DN-IKK α upon TNF- α -mediated ICAM-1 expression	in
	HUVECs	99
Figure 3.22	The effect of Adv. DN-IKK α upon TNF- α -mediated VCAM-1 expression	n in
	HUVECs	100
Figure 3.23	The effect of Adv. DN-IKK α and siRNA IKK α upon TNF- α -induced	
	ICAM-1 and VCAM-1 expression in HUVECs	101
Figure 3.24	Time course and concentration dependency of IL-1 β mediated ICAM-1 ar	nd
	VCAM-1 expression in HUVECs	104
Figure 3.25	The effect of Adv. DN-IKK β upon IL-1 β -mediated ICAM-1 and VCAM-	1
	expression in HUVECs	105
Figure 3.26	The effect of Adv. DN-IKK α upon IL-1 β –mediated ICAM-1 and VCAM	-1
	expression in HUVECs	106

Chapter 4 Characterisation of the inhibitory effect of IKKα selective inhibitors (SU compounds) on Canonical and Non-Canonical NFκB pathways

Figure 4.1	The effect of SU1007 upon TNF- α -mediated I κ B- α degradation	
	in HUVECs	121
Figure 4.2	The effect of SU1010 upon TNF- α -mediated I κ B- α degradation	
	in HUVECs	122
Figure 4.3	The effect of SU1007 and SU1010 upon TNF- α -mediated	
	I κ B- α degradation induced by 1ng/ml of TNF- α in HUVECs	124
Figure 4.4	The effect of prolonged pre-treatment of SU1007 and SU1010 upon	
	TNF- α -mediated I κ B- α degradation in HUVECs	125
Figure 4.5	The effect of SU1007 upon TNF- α -mediated p65 phosphorylation	
	in HUVECs	127
Figure 4.6	The effect of SU1010 upon TNF- α -mediated p65 phosphorylation	
	in HUVECs	128
Figure 4.7	The effect of SU1007 and SU1010 upon IL-1 β -mediated	
	I κ B- α degradation and p65 phosphorylation in HUVECs	130
Figure 4.8	The effect of SU1007 upon TNF- α -mediated p100 processing	
	in HUVECs	133
Figure 4.9	The effect of SU1010 upon TNF- α -mediated p100 processing	
	in HUVECs	135
Figure 4.10	The effect of SU compounds on Adv. DN-IKK β pre-treated	

100	•
n (W)	nrococcing
1) 1 (// /	DIOCESSIN
P100	processing

Figure 4.11	The effect of SU1028 and SU1053 upon TNF- α -mediated p100 processin	g
	in HUVECs	139
Figure 4.12	The effect of SU937 upon TNF- α -mediated p100 processing	
	in HUVECs	140
Figure 4.13	The effect of SU1007 and SU1010 upon TNF- α -mediated	
	NFκB luciferase activity in HUVECs	142
Figure 4.14:	The effect of SU1007 upon TNF- α -mediated ICAM-1 expression	
	in HUVECs	144
Figure 4.15	The effect of SU1010 upon TNF- α -mediated ICAM-1 expression	
	in HUVECs	145
Figure 4.16	The effect of SU1007 upon TNF- α -mediated VCAM-1 expression	
	in HUVECs	147
Figure 4.17	The effect of SU1010 upon TNF- α -mediated VCAM-1 expression	
	in HUVECs	148
Figure 4.18	The effect of SU1007 and SU1010 upon TNF- α -mediated ICAM-1	
	and VCAM-1 expression in HUVECs	150
Figure 4.19	The effect of SU1007 and SU1010 incorporated with Adv. DN-IKK β	
	upon TNF-α-mediated ICAM-1 expression in HUVECs	152
Figure 4.20	The effect of SU1007 and SU1010 upon IL-1 β -mediated ICAM-1	
	and VCAM-1 expression in HUVECs	153

List of Tables

Chapter 4 Characterisation of the inhibitory effect of IKKα selective inhibitors (SU compounds) on Canonical and Non-Canonical NFκB pathways

Table 4.1The effects of SU1007 or SU1010 on IKK α and β

in vitro kinase activities

119

Chapter 1

General Introduction

1.1 Physiology of the Endothelium

In multicellular organisms, blood vessels are essential for the function of the circulatory system. Oxygen and nutrients are delivered from blood vessels to various tissues, while waste products are removed from tissues to the circulation and then excreted from the body. The circulation system is also responsible for inter-organ communication such as transporting hormones from one organ to another. Vascular endothelium is a thin layer of endothelial cells that lines the tunica intima of the inner layer of the blood vessels, forming an interface between circulating blood in the lumen and the rest of the vessel wall. Endothelial cells line the entire circulatory system, from the heart to the smallest capillary. The endothelial cell surface in an adult human is composed of approximately 1 to 6 x 10¹³ cells, weighs approximately 1kg, and covers a surface area of approximately 1 to 7 m² (Augustin et al., 1994, Fishman, 1982). One of the roles of vascular endothelium is in delivering oxygen and nutrients to various underlying tissues and removing wastes from tissues. There is also a reciprocal relationship between the endothelium and its surrounding microenvironment which is important not only during organ development and patterning, but also in regulating the local immune response under pathological conditions (Baeuerle and Baltimore, 1988, Nikitenko, 2009).

The endothelium plays a key role in the regulation of blood pressure. Its presence reduces turbulent blood flow by controlling blood vessel contraction and relaxation (Stankevicius et al., 2003). Both vasoconstriction and vasodilatation are crucial in the regulation of blood pressure. Vasoconstriction is regulated by vasoactive molecules such as angiotensin II and vasopressin which act on their cognate receptors to induce vascular smooth muscle contraction, thus reducing the size of lumen and hence raising blood pressure. However the propensity towards vasoconstriction must be counterbalanced by vasodilatation, otherwise blood vessels would be damaged under high blood pressure over a prolonged period. Vasodilatation is mediated by the synthesis and release of key endothelial derived relaxing factors such as nitric oxide (NO) (Harrison, 1997, Kubes et al., 1991, Kugiyama et al., 1996), prostacyclin (PGI₂) (Camacho et al., 2008, Jaffe and Weksler, 1979) and epoxyeicosatrienoic acids (Stankevicius et al., 2003) which migrate into the vascular smooth muscle layer and cause relaxation of the vascular smooth muscle. For example, NO diffuses into the smooth muscle cell and activates soluble guanylyl cyclase to increase intracellular cyclic guanosine monophosphate (cGMP) concentration. An increase of cGMP level results in a decrease of

the intracellular calcium concentration in smooth muscle cells leading to relaxation (Stankevicius et al., 2003). In contrast, prostacyclin (PGI₂), released by the endothelial cell through arachidonic acid metabolism, regulated in turn by phospholipase A_2 and cyclooxygenase (COX) activity, interacts with specific receptors on the smooth muscle surface resulting in increases in intracellular cAMP and subsequent relaxation (Coleman et al., 1994). The effects of both NO and PGI₂ are strongly related to each other, PGI₂ facilitates NO release from endothelial cells, while NO potentiates PGI₂ effects in smooth muscle (Stankevicius et al., 2003).

1.2 Dysfunction of the Endothelium

A number of risk factors for cardiovascular disease contribute to endothelial dysfunction such as smoking, hypertension, hypercholesterolaemia, diabetes and obesity. Moreover, physical inactivity can also lead to endothelial dysfunction (Suvorava et al., 2004). This is linked to cellular changes for example, disregulation of NO synthesis is associated with these conditions and also diabetes, shock, infarction, neurodegeneration, arthritis and chronic inflammation (Gross and Wolin, 1995). Endothelial dysfunction also leads to decreased secretion of vasodilator mediators, and increased production of endothelium-derived vasoconstrictors, and/or resistance of vascular smooth muscle to vasodilators (Mombouli and In addition to promoting vasodilation, normal endothelium has anti-Vanhoutte, 1999). atherosclerotic and anti-thrombotic functions. It inhibits platelet aggregation, monocyte adhesion, vascular smooth muscle cell proliferation, and thrombosis, all of which are important factors in the development of atherosclerosis and eventual plaque disruption (Coleman et al., 1994, Flavahan, 1992, Levine et al., 1995, McGorisk and Treasure, 1996, Vane et al., 1990, Vogel, 1997). Endothelial dysfunction and atherosclerosis will be discussed further in section 1.6.5.

The initial stage of atherosclerosis is believed to be associated with the dysfunction or activation of vascular endothelial cells (Dzau et al., 2002, Libby et al., 2010b, Ross, 1999). In the latter stage, endothelial cells are associated with the deposition of lipids such as low density lipoprotein (LDL) and cholesterol, infiltration of inflammatory cells, and proliferation of vascular smooth muscle cells (Dzau et al., 2002, Libby et al., 2010b, Ross, 1999). Unstable atherosclerotic plaques become susceptible to rupture that contributes to the formation of thrombus and this may lead to more severe events such as myocardial infarction (Dzau et al.,

2002, Libby et al., 2010b, Ross, 1999) and stroke (Fadini et al., 2007). Infiltrating macrophages and T lymphocytes produce inflammatory cytokines such as tumour necrosis factor-alpha (TNF- α) and interleukin-1 beta (IL-1 β) which regulate endothelial cell responses. This includes the induction of cellular adhesion molecules such as vascular cell adhesion molecule-1 (VCAM-1) on the endothelial cell surface, which in turn mediates recruitment of circulating leukocytes to the arterial wall (Libby et al., 2010b, Price and Loscalzo, 1999, Ross, 1999). Endothelial cells along with macrophages and vascular smooth muscle cells release matrix metalloproteinases (MMPs) which increase the degradation of the components of the fibrous cap covering the atherosclerotic plaque and contributes to plaque disruption (Libby, 2002, Saren et al., 1996).

The role of the endothelium is not restricted to cardiovascular disease. There is evidence in the cancer field that endothelial cells provide support for tumour growth and metastatic spread. Excessive proliferation and transformation of endothelial cells can lead to pathological angiogenesis or lymphangiogenesis or vascular malfunctions, which are now considered to be the hallmarks of malignant disorders (Nikitenko, 2009).

1.3 Signalling mechanisms in Endothelium

There are several intracellular signalling events that are required for endothelial cells to induce different physiological responses. These responses are initiated by a considerable number of extracellular agents including; growth factors such as vascular endothelial growth factor (VEGF), Fibroblast growth factor (FGF) and Epidermal growth factor (EGF), cytokines including TNF- α and IL-1 β , activators of G-protein coupled receptors (GPCR) such as endothelin-1 (ET-1), angiotensin-II (Ang-II) and vasopressin and stimuli such as hypoxia, oxygen free radicals and cellular distension. Cellular activation by these agents converge on a number of key intracellular signalling pathways which are essential in the regulation of endothelial cell function. In particular, the Mitogen Activated Protein Kinase (MAPK), Phosphoinositide-3 Kinase (PI3K) and Nuclear Factor Kappa B (NF κ B) pathways have been strongly implicated. These pathways are intimately linked to cell inflammation, proliferation, migration and apoptosis and as they are tightly regulated in normal conditions, over-activation of these pathways leads to development of many diseases such as arthritis and cancers. Therefore studying the signalling mechanisms involved in endothelial cell function provides valuable information for targeting and treatment of diseases.

1.3.1 Mitogen-Activated Protein Kinase (MAPK) Signalling Pathways

The MAPKs superfamily is a major system participating in the transduction of signals from the cell membrane to nuclear and other intracellular targets. MAPKs respond to numerous stimuli by phosphorylating a variety of substrates including transcription factors, enzymes, and other kinases. They are a three-tiered kinase cascade comprising a MAP kinase kinase kinase (MAPKKK), a MAP kinase kinase (MAPKK) and an effector MAPK. The most wellknown MAPK signalling pathways included the extracellular regulated kinases 1/2 (ERK1/2), c-Jun N-terminal kinase (JNK) and the p38 MAP kinase pathway (Qi and Elion, 2005). There are numerous isoforms of each kinase in particular, JNK and p38, and multiple MAPKKKs and MAPKKs which allow for specific and overlapping regulation of each MAPK (Qi and Elion, 2005). Figure 1.1 outlines each of the MAPK signalling pathways. The members of the MAPK families are responsible for regulation of a large variety of cellular processes such as cell growth, differentiation, development, cell cycle regulation, death and survival. While ERK1/2 and also Big MAPK-1 (BMK1) play an important role in growth and cytoprotective functions, JNK and p38 proteins are implicated in cell death, inflammatory and stress responses. Because they have predominantly contradictory roles, the relative activation of these kinases is important to the functional status of the cell. Additionally, there is known to be a crosstalk between MAPK cascades whereby the activity of one MAPK can be influenced by another. Thus, these proteins collectively integrate the pro- and antiinflammatory stimuli acting on the cell to produce the appropriate downstream effects (Hoefen and Berk, 2002).

Whilst not examined directly in this thesis, there is evidence to show that MAP kinases play a role in endothelial cell function. Pro-inflammatory cytokines or genotoxic stress such as ionising radiation or cisplatin can induce JNK translocation into the mitochondria from the cytoplasm. JNK then phosphorylates numerous targets such as Bcl-X, Bax and 14-3-3 σ (Cadalbert et al., 2005, Dhanasekaran and Reddy, 2008, Kharbanda et al., 2000, Kim et al., 2006, Yamamoto et al., 1999). These proteins are involved in release of cytochrome-c which is a crucial step for the initiation of apoptosis. Some studies showed inhibition of JNK activity leads to a reduction of apoptosis in cardiomyocytes. (Remondino et al., 2003, Aoki et al., 2002). Hydrogen peroxide (H₂O₂) activates SAPK/Erk kinase (SEK1)/JNK phosphorylation and hence activates caspase-3 and leads to apoptosis. Inhibiting JNK activity by dominant negative JNK (DN-JNK) and SEK1 adenovirus however showed reverse of

apoptosis (Murakami et al., 2005). Although JNKs plays an important role in the regulation of apoptosis, JNKs are also involved in the regulation of pro-survival activity. Inhibition of JNK1/2 using a synthetic peptide (JNKI 1) specifically inhibits JNK1 phosphorylation and leads to an increase in HUVECs apoptosis, although the same inhibitor did not affect cell migration or proliferation (Salameh et al., 2005). Therefore, JNK has an essential role in balancing both the pro- and anti-apoptotic pathways.

Another important MAPK sub-group is ERK. There are six isoforms of ERK but the most studied among these are ERK-1, ERK-2 and ERK-5. ERK signalling can be activated by a wide range of stimuli through activation of a number of different receptors such as epidermal growth factor receptor (EGFR), GPCRs and transforming growth factor β receptor (TGFR) (Schramek, 2002). ERK1 and ERK2 are activated by MAP (Mitogen-Activated Protein) Kinase/ERK (Extracellular Signal-Regulated Kinase) Kinase 1 (MEK1) and MEK2 (Zheng and Guan, 1993) and regulate several transcription factors such as E-twenty six (ETS)-like transcription factor 1 (Elk-1) (Marais et al., 1993), c-myc (Gupta et al., 1993) and upstream binding factor (UBF) (Stefanovsky et al., 2001) which are important in the regulation of cell survival, proliferation and differentiation. In fact, ERKs play an important role in endothelial cell proliferation and migration. Endothelial cells from ERK1 and 2 double knockout mutant embryos showed a 3-fold decrease in proliferation of endothelial cells compared with wild type (Srinivasan et al., 2009). Moreover, a marked diminution of proteins regulating G1 to S phase progression was observed, including cyclin D1, cyclin E, Cyclin-dependent kinase 4 (CDK4) and c-Myc (Srinivasan et al., 2009). In addition, an in vitro experiment showed endothelial cells lacking both ERK1 and 2 had attenuated cell migration and invasion compared with control (Srinivasan et al., 2009). Furthermore, the MEK inhibitor PD184352, inhibited phosphorylation of ERKs and induced HUVECs apoptosis. Over-expression of dominant-negative MEK1 (MEK1A) in HUVECs also inhibited phosphorylation of ERKs and lead to failure of tubule formation (Mavria et al., 2006). Overall, these studies outlined above confirm the role of ERKs in the regulation of the proliferation and migration of endothelial cells.



Figure 1.1 Schematic representation of the MAPK cascades and their nuclear targets. Activation of ERK1/2 signalling pathway results in activation of a series of transcription factors such as E-twenty six like transcription factor (Elk-1), specific protein 1(SP-1) and c-Myc which play an important role in the regulation of cell survival and proliferation. Whilst activation of JNKs pathway results in activation of transcription factors including c-Jun, activating transcription factor (ATF-2) for cell survival, proliferation and also apoptosis (Adapted from (Plotnikov et al., 2011)

1.3.2 The Phosphoinositide-3 Kinase Signaling Pathway

The PI3K/PTEN/AKT pathway is important in the regulation of cellular function, in particular cytoskeletal organisation and cellular survival (Gautreau et al., 1999). Cells exposed to growth factors show increased recruitment and phosphorylation of phosphoinositide-3 kinase (PI3K) by the activated receptor. PI3K phosphorylates phosphatidylinositol 4,5-bisphosphate (PIP₂) to form phosphatidylinositol-3,4,5-trisphosphate (PIP₃) which subsequently activates a serine/threonine-protein kinase (AKT) through the intermediate phosphoinositide-dependent kinase-1 (PDK1). AKT is also called protein kinase B (PKB) which phosphorylates and regulates the activity of a number of downstream targets including kinases, transcription factors and other regulatory molecules (Fayard et al., 2010). The activity of the PI3K pathway is negatively regulated by Protein Tyrosine Phosphatase (PTEN) which is implicated in a number of cancers. PTEN suppresses tumour cell growth by antagonising the action of receptor tyrosine kinases and regulates tumour cell invasion and metastasis in human brain, breast and prostate (Li et al., 1997), pancreatic ductal adenocarcinoma (Ying et al., 2011) and colorectal cancer (Byun et al., 2011). Figure 1.2 illustrates the PI3K/PTEN/AKT signalling pathway. In fact, recent evidence indicates there is cross-talk between PI3K and other signalling pathways. For instance, a number of studies show one of the targets of AKT is the Inhibitory kappa B kinase (IKK) complex which is a critical kinase in regulation of NFkB pathway (see below) (Ozes et al., 1999, Rajavashisth et al., 1995).

It is well recognised the PI3K plays a key role in endothelial cell function, in particular in response to VEGF and other growth factors. Once AKT is phosphorylated, it can down-regulate the expression of apoptotic factors such as caspase-9 (Cardone et al., 1998) and BAD (Datta et al., 1997). Furthermore, it also activates survival factors such as NF κ B (Ozes et al., 1999) and apoptosis regulator bcl-2. Moreover, constitutive up-regulation of B-cell lymphoma-2 (bcl-2) expression prevents endothelial cell apoptosis in a model of capillary network formation (Pollman et al., 1999), therefore AKT is crucial for promoting in endothelial cell survival and angiogenesis. In addition, AKT is associated strongly with several cytoskeletal-mediated processes, such as regulation of actin reorganisation during VEGF-induced migration of endothelial cells and signalling by cytoskeleton-plasma membrane linker protein (Morales-Ruiz et al., 2000). AKT interacts with Ezrin which also functions as plasma membrane-actin cytoskeleton linker and aids cell survival (Gautreau et

al., 1999). In addition, PI3K/AKT signalling pathway regulates endothelial nitric oxide synthase (eNOS) activation. Dimmeler and co-workers showed inhibition of the PI3K/AKT pathway or mutation of the AKT site on eNOS protein (at serine 1177) attenuated the serine phosphorylation and prevents the activation of eNOS (Dimmeler et al., 1999). eNOS produces NO which plays an important role in the local regulation of vascular homeostasis and platelet aggregation (Moncada and Higgs, 1993). A decrease in the bioavailability of NO is a characteristic feature in patients with coronary artery disease (Zeiher, 1996) and aggravates the development of atherosclerotic lesions (Moroi et al., 1998).



Figure 1.2 illustrates the PI3K/PTEN/AKT signaling pathway. The diagram shows the AKT is a key factor in controlling the activity of regulatory molecules. AKT regulates protective mechanism by activating the endothelial nitric oxide synthase (eNOS) and IKK in endothelial cell against vascular injury. In addition, AKT inhibits the pro-apoptotic factors including Bcl-2-assoicated death promoter (BAD) protein, glycogen synthase kinase-3 (GSK3) and caspase-9 to inhibit apoptosis.
1.3.3 The Nuclear Factor Kappa B Signalling Pathway

Nuclear factor kappa B (NF κ B) is a ubiquitous signalling pathway which plays a key role in a number of both physiological and pathophysiological processes. This includes responses to infection, fetal development, cancer, inflammation, atherosclerosis and angiogenesis (Brand et al., 1996, Brand et al., 1997b). NF κ B was first identified by Sen and Baltimore in 1986 (Sen and Baltimore, 1986) as a transcription factor which binds to a specific decameric DNA-sequence (5'-GGGACTTTCC-3') within the intronic enhancer of the immunoglobulin kappa light chain gene in mature B cells (Sen and Baltimore, 1986). NF κ B has been found in many cell types and regulates transcriptional activity through binding to specific binding sites, termed κ B sites, within the promoters of a number of genes, which have the consensus sequence 5'-GGGRNYYYCC-3', where R is purine and Y is pyrimidine (Kunsch et al., 1992, Parry and Mackman, 1994). Figure 1.3 illustrates the structure of different NF κ B members.

1.3.3.1 Structure of NFкB

Early studies showed NF κ B to exist as a heterodimer, purification of NF κ B revealed two proteins involved in DNA binding. The molecular weight of those proteins were approximately 65kDa and 50kDa, latterly named as Rel A and p50 respectively (Ghosh et al., 1990, Kawakami et al., 1988, Zabel et al., 1991). Rel A was also called p65, while p50 named as NF κ B1. Later studies showed that the heterodimer of NF κ B bound to the cognate κ B-site of the immunoglobulin kappa light chain gene (Baeuerle and Baltimore, 1989, Urban et al., 1991). Further study of the structure of p65 showed it to contain a Rel homology domain (RHD) (Nolan et al., 1991, Ruben et al., 1991). Later, another heterodimer of NF κ B was discovered which consisted of Rel B and p50 (Ryseck et al., 1992).

Currently, five subunits of NF κ B have been identified; Rel A, Rel B, c-Rel, p52 and p50. In fact, NF κ B is not always found as a heterodimer, but also exists as homodimers. Indeed, different combinations of NF κ B dimer have been discovered including; p52/c-Rel, p65/c-Rel, Rel A/Rel A, p50/p50, p52/p52 and p50/p52 (Ganchi et al., 1993, Hansen et al., 1994, Kang et al., 1992, Molitor et al., 1990). All NF κ B subunits contain a conserved N-terminal region called RHD within which lies DNA-binding and dimerization domains, DNA binding and I κ B-binding motifs and a nuclear localisation signal (NLS).

NF κ B is further sub-divided in two classes. Class 1 NF κ Bs consist of p100/p52 and p105/p50. These subunits do not have C-terminal transactivation regions, instead they contain a transrepression domain which consists of a number of ankyrin repeats. Studies have shown that homo- or heterodimers of p50 and p52 can repress κ B-dependent transcription in vivo (Brown et al., 1994, Kang et al., 1992, Lernbecher et al., 1993, Plaksin et al., 1993, Schmitz and Baeuerle, 1991). It is believed that the inhibitory effect upon transcriptional activity by p50 and p52 homodimers is due to competition for DNA-binding with other transcriptionally active dimers such as p65/p50 (Plaksin et al., 1993, Schmitz and Baeuerle, 1991).

Rel A, Rel B and c-Rel are classified as Class 2 NF κ B subunits. In contrast, these NF κ B subunits contain a transactivation domain within the C-terminal RHD. The transactivation domain contains serine, acidic and hydrophobic amino acids. Mutation of any of these amino acids reduces the transactivating activity and transcription of NF κ B dependent genes (Dobrzanski et al., 1993). There is an N-terminal leucine zipper-like region in Class 2 NF κ B subunits which allow the Class 2 NF κ B physically attached to each other. Activation of the C-terminal leads to full activation of most NF κ B isoform, however interestingly, both N and C-terminals are required for complete activation of Rel B (Dobrzanski et al., 1993).



Figure 1.3 NF κ B/I κ B subunits. The structures of various subunits of NF κ B and I κ B members. The number of amino acid is shown next to each NF κ B and I κ B members. LZ, leucine zipper; TD, transactivation domain, Rel homology domain, PEST region, polypeptide sequence enriched in proline (P), glutamic acid (E), serine (S) and threonine (T) region, GRR, glycine-rich region, DD, death domain.

1.3.3.4 Genes Regulated by NFкB

NF κ B plays a key role in a regulation of a number of cellular functions including growth, survival, apoptosis, differentiation and development. This in turn is mediated by transcriptional regulation of a number of NF κ B dependent genes following binding of homo or heterodimer to specific DNA binding motifs.

For p65/p50, the consensus sequence 5'-GGGRNYYYCC-3' (where R=A or G; Y=C or T; W=T or A; V=A, C or G) (Chen and Ghosh, 1999, Chen et al., 1998), has been identified. Through this sequence, Rel A regulates expression of genes such as ICAM-1, VCAM-1 and COX-2 (Denk et al., 2001). Rel A (p65) is also responsible for the expression of anti-apoptotic genes, for instance X-linked inhibitor of apoptosis protein (XIAP) and cellular inhibitor of apoptosis protein (c-IAP) (Chu et al., 1997, Stehlik et al., 1998b) and pro-survival factors such as bcl-2 (Catz and Johnson, 2001). These genes are responsible for regulation of cell survival and proliferation.

In contrast to Rel A (p65), Rel B is responsible for lymphoid development, immune cell trafficking and angiogenesis. As the genes regulated by Rel B are clearly different than Rel A, unsurprisingly, the DNA binding sequence of Rel B is also distinct. The DNA binding sequence of Rel B has been identified as 5'-GGGRVWTTYY-3' (Britanova et al., 2008). In 2002, Dejardin and co-workers discovered that Rel B controls the expression of several chemokines such as secondary lymphoid tissue chemokine (SLC), B lymphocyte chemoattractant (BLC), EBI-1-ligand chemokine (ELC) and stromal cell-derived factor-1a (SDF1). These genes are responsible for regulating the lymphoid organogenesis process (Dejardin et al., 2002). Wharry and co-workers found genes such as cytokine B cell activation factor (BAFF) and chemokines including chemokine C-X-C motif ligands (CXCL12, CXCL13), C-C motif ligands (CCL19 and CCL21) are up-regulated by constitutive activation of the non-canonical pathway in pancreatic ductal adenocarcinoma (PDAC). Several of these genes are strongly related to the development of lymphoid malignancies and solid tumours (Wharry et al., 2009).

1.3.3.5 Introduction to the NFkB signalling pathway

As mentioned, five subunits of NF κ B have been identified which mediate cellular activation of the pathway in the form of specific NF κ B dimers with DNA binding and transactivation activity. There are three main signalling pathways which mediate NF κ B activation. These are: 1) the canonical NF κ B pathway also known as the classical pathway, which is activated by cytokines such as TNF- α and IL-1 β , 2) the non-canonical NF κ B pathway prevalent in a restricted number of cell types, which is activated by lymphotoxin and 3) the p105 NF κ B pathway, which is constitutively active. Figure 1.4 illustrates the mechanisms involved in the activation of each pathway which will be discussed in detail in subsequent sections. For each pathway however, there is a common final mode of activation. At rest, NF κ B binds to a corresponding inhibitory molecule called inhibitory kappa B (I κ B) which functions to retain NF κ B within the cytoplasm. Following cellular activation through a diverse number of signals, including reactive oxygen species (Schreck et al., 1992, Schreck et al., 1991), ultraviolet light (Devary et al., 1993) and pro-inflammatory cytokines, NF κ B dissociates from the I κ B to allow translocation into the nucleus. In turn, I κ B is regulated by a number of upstream mechanisms which will be discussed in the following section.

1.3.3.6 Structure of IkB

In 1988, Baeuerle and co-workers first identified an inhibitory kappa B protein which acted as a specific inhibitor of NF κ B transcriptional activity by binding and retaining NF κ B within the cytosol (Baeuerle and Baltimore, 1988). All I κ B molecules are now known to contain between three and seven ankyrin repeats and to date, cDNAs encoding eight structurally related members of the mammalian I κ B family have been cloned: p100, p105 (Mercurio et al., 1993, Rice et al., 1992), I κ B- α (Davis et al., 1991), I κ B- β (Thompson et al., 1995), I κ B- γ (Inoue et al., 1992a), I κ B- ϵ (Li and Nabel, 1997) and Bcl-3 (Ray et al., 1995). Figure1.3 illustrates the structure of all the I κ B members.

I κ B molecules require the ankyrin repeats to bind to the regions of the RHD within Rel/NF κ B dimer (Cervantes et al., 2009, Hatada et al., 1992). This allows the I κ B protein to mask the nuclear localization sequence (NLS), and hence prevent nuclear translocation of Rel/NF κ B protein (Beg et al., 1992, Henkel et al., 1992). Moreover, mutation of any of the five regions

within the ankyrin repeats of $I\kappa B$ - β results in loss of binding to NF κB p65 and leads to NF κB translocation to the nucleus and enhanced DNA binding (Inoue et al., 1992b).

A number of studies have demonstrated that separate I κ B proteins preferentially inhibit distinct Rel/NF κ B protein dimers. For example, I κ B- α and I κ B- β both strongly bind both c-Rel and Rel A (Baeuerle and Baltimore, 1989, Beg et al., 1992, Thompson et al., 1995). In addition, studies show that other I κ B proteins such as the 70kDa form of I κ B- γ (Grumont and Gerondakis, 1994) and I κ B- ϵ (Li and Nabel, 1997) also inhibit Rel A/p50 dimers. In contrast, it has been demonstrated that p100 and p105 associate less selectively with NF κ B isoforms. They can form inhibitory complexes with p50, p52, Rel A and c-Rel (Mercurio et al., 1993, Rice et al., 1992). Unlike any other subunits, however, Rel B can only bind to p100 and p105, which effectively inhibits p52/Rel B heterodimers formation (Dobrzanski et al., 1995). In each of the three NF κ B pathways, the relevant I κ B proteins function in overlapping but distinct ways suggesting a variety of functional outcomes.



Figure 1.4. Schematic diagram of NF\kappaB pathways. There are three NF κ B pathways which are canonical and atypical (non – canonical and p105) pathways. Stimulants bind to various receptors and trigger the signalling pathways, subsequent NF κ B dissociated from I κ B and translocates into the nucleus to bind to specific DNA sequence and express adhesion molecules and endothelial cytokine (adapted from Beinke and Ley 2004).

1.3.3.6.1 Inhibitory kappa B-alpha (ΙκΒ-α)

Two major IkB isoforms were discovered in 1990, which were found to associate with NFkB. These were designated as IkB- α and IkB- β with molecular weights of 37 kDa and 43 kDa respectively (Zabel and Baeuerle, 1990). A number of studies have confirmed that IkB- α but not IkB- β plays an important role in the regulation of the canonical pathway. At rest, IkB- α acts as a specific inhibitor, which binds to the NFkB dimer (p65/p50) to prevent translocation into the nucleus (Jaffray et al., 1995). IkB- α binds to p50 homodimers with the lowest affinity, p65 homodimers with medium affinity, and p50/p65 heterodimers with the highest affinity (Malek et al., 1998). This may be due to more contact sites between IkB- α and p65 than for p50 when bound to the heterodimer (Huxford et al., 1998). Mutation of the NLS site of p65 results in failure of p65/IkB- α binding, this suggests that the p65/ IkB- α binding domain is within the NLS. There are two major regions within the p65 which are important for the binding of IkB- α ; aa151 to 312 and aa450 to 500. Mutation within these regions leads to failure of p65/IkB- α binding, but does not modify subsequent events and DNA binding domain enhanced activation (Ganchi et al., 1992).

Following cellular activation, $I\kappa B-\alpha$ is subsequently phosphorylated and ubiquitinated. Serine residues 32 and 36 in $I\kappa B-\alpha$ have been shown to be crucial sites for phosphorylation in response to cellular stimulation by TNF- α , phorbol 12-myristate 13-acetate (PMA) and ionomycin (Brockman et al., 1995, Brown et al., 1993). Phosphorylation of $I\kappa B-\alpha$ functions to initiate ubiquitination and targets the protein for proteasomal degradation which in turn frees NF κ B to translocate to the nucleus. Ubiquitination of $I\kappa B-\alpha$ is an essential step before $I\kappa B-\alpha$ degradation (Li et al., 1995). In the same year, it was found that lysine residues 21 and 22 serve as the primary sites for signal-induced ubiquitination of $I\kappa B-\alpha$, substitution of both residues results in inhibition. With mutation of these sites, $I\kappa B-\alpha$ can still be phosphorylated by the IKK complex (see below) but fails to undergo ubiquitination and degradation, hence NF κ B cannot dissociate from $I\kappa B-\alpha$ and remains within the cytoplasm (Chen et al., 1995).

1.3.3.6.2 Inhibitory kappa B NFκB2 (p100/p52)

The activation of RelB/p52 is largely distinct from other NF κ B subunits and the role of p100 as an I κ B member is unique. p100/p52 is also called NF κ B2 but was originally named as p98

when characterised as a 98kDa protein. Mercurio and co-workers originally found that the NF κ B related protein p55 was produced from p98 via a processing mechanism (Mercurio et al., 1992, Mercurio et al., 1993). Later, this mechanism was designated as p100/p52 processing to reflect the slight difference in the molecular weights of these two particular proteins. Similar to I κ B- α , p100 also contains ankyrin repeats in the C-terminal region, whilst the C-terminal domain of both p100 and p105 are required to retain Rel B in cytoplasm. At rest, both p100 and p105 are capable of inhibiting agonist-induced NF κ B activation (Mercurio et al., 1993, Rice et al., 1992). The proteolytic processing of p105 and p100 involves proteasome-mediated degradation of the C-terminal half of the protein including the ankyrin repeat domain, resulting in formation of p50 and p52 respectively (Fan and Maniatis, 1991, Siebenlist et al., 1994). Although p100 usually forms a complex with Rel B, in fact, p100 is also capable of interacting with c-Rel. Both p100/c-Rel and p105/c-Rel complexes were found to be processed into active p52/c-Rel complexes (Mercurio et al., 1993). Further details will be discussed in relation to signalling through the non-canonical pathway in Section 1.3.3.9.

1.3.3.7 Identification of Inhibitory Kappa B kinase

As described in Section 1.3.3.6, I κ B phosphorylation is essential for activation of the NF κ B signalling pathway. However, the kinases responsible for this process were not easily identified and a series of candidates were investigated including PKC isoforms, MAP kinases, MEKKs and PKB. In 1996 Chen and co-workers identified a high molecular mass complex (700-900 kDa) which regulated the phosphorylation of I κ B- α at serine 32 and serine 36 (Chen et al., 1996). This complex was identified and later named as I κ B kinase (IKK). The functional role of the IKK complex was then elucidated by DiDonato and co-workers in 1997. They first demonstrated that in HT-29 cell, TNF- α activated IKK complexes and then showed over-expression of IKK α induced I κ B- α phosphorylation and degradation. Moreover, overexpression of IKK increased TNF- α or IL-1 β induced NF κ B luciferase activity, while anti-sense IKK completely abolished TNF- α or IL-1 β induced NF κ B luciferase activity (DiDonato et al., 1997).

The IKK complex consists of IKK α , IKK β and IKK γ . Both IKK α and IKK β are catalytically active kinases. An early study showed IKK α and IKK β to have similar structures with the overall protein sequence of IKK α and IKK β being 52% identical. Both proteins have an NH₂-

terminal kinase domain sharing 64% identity, a C-terminal helix-loop-helix (HLH) domain and a leucine-zipper motif located in the centre of the kinase which is involved in homodimer and heterodimer formation (Woronicz et al., 1997). Also called NF κ B-essential-modifier (NEMO), IKK γ is a non-catalytic regulatory component. An amino-terminal α -helical region of NEMO is responsible for binding to the carboxyl-terminal of either IKK α or IKK β . Blocking the association of NEMO with the IKKs leads to inhibition of cytokine-induced NF κ B activation and hence the genes regulated by NF κ B (May et al., 2000). Studying the role of individual IKK subunits is important to broaden the knowledge of NF κ B signalling pathways and this has been a topic of intense study over the last 15 years.

1.3.3.8 The canonical (classical) NF_KB pathway

Concomitant with the discovery of the IKKs as key kinases involved in the regulation of NF κ B activation was the identification of intermediates which link initial cellular activation to IKK itself. This was the case in particular for cytokine receptors, for example TNFR1 and the IL-1 β receptor as well as the toll-like receptors (TLRs) which were found to mediate the action of both Gram positive and Gram negative bacteria and other infectious agents. These receptors are well recognised to activate the NF κ B pathway in a large number of cell types (Plotnikov et al., 2011).

In response to TNF- α , the IKK complex is recruited to TNFR1 via TNF receptor-associated factor 2 (TRAF2) and then activated. The interaction occurs through the leucine zipper motif of IKK α , IKK β and the RING finger domain of TRAF2 (Devin et al., 2001). In support of this model, Rothe and co-workers showed that over-expression of TRAF2 results in increased p65-DNA binding and also NF κ B-driven luciferase reporter activity through activation of the other TNF- α receptor isoform, TNFR2. Furthermore, HEK 293 cells containing mutated TRAF2, which lacks the NH₂ terminal 86 amino acids that comprise the RING finger domain, inhibited TNF- α stimulated NF κ B luciferase activity (Rothe et al., 1995). Transforming growth factor β -activated kinase 1 (TAK-1) is a member of the MAP3K family which also plays a critical role in activation of the IKK complex through phosphorylation. TNF- α induces a rapid association of TAK1 with IKK α and IKK β and also TRAF2. Moreover, siRNA directed against TAK1 inhibits TNF- α induced I κ B- α degradation and hence the activation of NF- κ B pathway (Takaesu et al., 2003).

In contrast, TRAF6 but not TRAF2 is required for IL-1 β induced NF κ B activation. In response to IL-1 β stimulation, an additional intermediate, TAB2, translocates from the membrane to the cytosol and acts as an adaptor to link TAK-1 with TRAF6. (Takaesu et al., 2000). A recent study showed TAK-1 binding protein 2 (TAB2) facilitates ubiquitination of TRAF6 in IL-1 stimulated cells (Kishida et al., 2005). Kanayama and co-workers found TAB2 and TAB3 bind preferentially to lysine 63-linked polyubiquitin chains of TRAF6 through a highly conserved zinc finger (ZnF) domain which subsequently leads to ubiquitination of the IKK complex, while mutation of the ZnF domain abolishes the ability of TAB2 and TAB3 to bind to polyubiquitin chains and hence fails to activate TAK1 and the IKK complex (Kanayama et al., 2004). Subsequently activated TAK1 phosphorylates the IKK complex and the downstream pathway (Wang et al., 2001).

Interestingly it has been demonstrated that there is a negative feedback mechanism in the canonical NF κ B pathway which involves the deactivation of IKK β . When pro-inflammatory agents such as IL-1 activate the canonical pathway, free NF κ B translocates to the nucleus to regulate the expression of specific genes one of which is I κ B- α . While I κ B- α is resynthesised, it is then exported from the nucleus back to cytoplasm and binds to NF κ B to form an inactive complex again. In parallel, protein phosphatase 2A (PP2A) dephosphorylates IKK β that allows for stabilization of resynthesized I κ B- α in cytoplasm (Barisic et al., 2008).

1.3.3.8.1 The role of IKKβ in the regulation of canonical NFκB pathway

The initial identification and purification of the IKK complex indicated that IKK β played a critical role in the canonical NF κ B pathway. However subsequent studies were required to confirm this model and involved the use of IKK β knockout mice, cellular expression of dominant-negative mutants (DN) of IKK β and more recently siRNA approaches.

Early work revealed difficulties in generating IKK β knockout mice for studies both *in vitro* and *in vivo* as IKK β knockout embryos died at approximately at day E 12.5. Histopathological examination revealed the cause of death to be massive liver apoptosis (Li et al., 1999c, Li et al., 1999d). Interestingly, an earlier study indicated that p65 and p50 knockout mice also died at E12.5 (Horwitz et al., 1997), implicating IKK β in the canonical NF κ B signalling. Despite these difficulties, it was shown that in IKK β deficient mouse embryonic fibroblast (MEFs), there was a marked reduction in TNF- α and IL-1 α induced

NF κ B activity (Li et al., 1999c) and as a consequence, enhanced apoptosis in response to TNF- α (Li et al., 1999d).

Later, additional experiments confirmed that IKK β is crucial in regulating NF κ B in various cell types. Overexpression of wild type IKK β in HEK239 cells led to phosphorylation on serine residues 32 and 36 of I κ B- α and serines 19 and 23 of I κ B- β and activation of NF κ B, whilst conversely, overexpression of a DN-IKK β (K44A) (deficient for ATP binding) inhibited cytokine induced NF κ B activation (Woronicz et al., 1997). TNF- α induced I κ B- α degradation was inhibited in MEFs from IKK β knockout mice which also failed to response with an increase in NF κ B activity following IL-1 α stimulation (Solt et al., 2009). Catley and co-workers showed that DN-IKK β inhibited either TNF- α -or IL-1 β -induced NF κ B-DNA binding and NF κ B linked luciferase activity in human pulmonary epithelial A549 cells (Catley et al., 2005). Moreover, TNF- α induced I κ B- α degradation was inhibited by DN-IKK β kinase adenovirus which resulted in abolition of NF κ B activity in HUVECs (Denk et al., 2001) and vascular smooth muscle cell (VSMC) (MacKenzie et al., 2007).

Importantly, a number of these studies also demonstrated that IKK β can also mediate phosphorylation of NF κ B p65 to initiate transactivation, leading to increased transcriptional activation following DNA binding. DN-IKK β adenovirus inhibited NF κ B p65 phosphorylation at serine 536 in HUVECs (Denk et al., 2001), with a similar result obtained in vascular smooth muscle (MacKenzie et al., 2007) and HeLa cells (Sakurai et al., 1999).

Significantly, many of these studies also linked IKK β to the regulation of a number of cellular responses including, inflammation and cell survival through the activation of p65 NFkB dependent gene transcription. It came as little surprise to discover that the inhibition of IKK β prevented the expression of genes important in these events. Again following DN-IKK β expression, Denk and co-workers found that TNF- α failed to induce ICAM-1, VCAM-1, MCP-1, E-selectin and COX-2 expression (Denk et al., 2001). Similarly, MacKenzie and co-workers also showed that IKK β was crucial in the regulation of ICAM-1 and VCAM-1 expression in vascular smooth cells (MacKenzie et al., 2007). Furthermore, inhibition of IKK β kinase activity also led to inhibition of inflammatory genes expression such as IL-8, COX-2 and ICAM-1 in A549 cell (Catley et al., 2005). This evidence also links IKK β to a number of diseases states such as arthritis and cancers and this will be examined in subsequent sections.

1.3.3.8.2 The role of IKKa in the regulation of the canonical NFkB pathway

Although IKK α and IKK β share 64% identity in the NH₂-terminal kinase domains as mentioned in Section 1.4.2, the functional role of IKK α within the canonical pathway remained obscure. As with IKK β , early studies again showed difficulty in developing IKK α knockout mice. Similar to IKK $\beta^{-/-}$ mice, global knockout of IKK α also resulted in embryonic lethality. However, the phenotype of IKK $\alpha^{-/-}$ embryos was completely different from that of IKK $\beta^{-/-}$ at the same developmental stage. Furthermore IKK $\alpha^{-/-}$ mice were born and lived for 30 minutes. The rudimentary limb and tail, a large omphalocele, and craniofacial features were all deformed in new-born IKK $\alpha^{-/-}$ mice. The most obvious phenotype was the skin which was taut, shiny, and completely devoid of wrinkles (Hu et al., 1999). Other studies suggested that skin abnormality in IKK $\alpha^{-/-}$ mice could be attributable to dysregulation of keratinocyte differentiation and proliferation (Hu et al., 1999, Li et al., 1999b, Takeda et al., 1999). Neither IKK $\beta^{-/-}$ nor IKK $\gamma^{-/-}$ mice showed a similar skin phenotype, which suggested that IKK α had a role independent from the canonical NF κ B pathway.

More recent experiments have further confirmed an essential role for IKK α in skin function. IKK α controls the production of a soluble factor such as keratinocyte differentiation-inducing factor (KDIF) that induces keratinocytes differentiation (Hu et al., 2001). Later, Liu and coworkers found a marked reduction in IKK α expression in poorly differentiated human squamous cell carcinomas (SCC) and identified IKK α mutations in exon 15. It is believed that the activity of IKK α in the epidermis may be responsible for resistance to chemical carcinogen-induced mitogenic and angiogenic activities, repressing tumour progression and metastases, thus lack of IKK α may lead to development of human skin cancers (Liu et al., 2006). In addition, IKK α was found to be essential for mammary gland development during pregnancy. A low level of cyclin D1 expression leads to impaired proliferation of mammary epithelial cells in IKK α ^{-/-} mice (Cao et al., 2001). The abnormality is due to the inability of mammary epithelial cells to respond to receptor activator of NF κ B (RANK) ligand, which activates NF κ B-induced cyclin D1 expression (Cao et al., 2001).

The distinctive phenotype of the IKK α knockout mice/cells relative to the knockout IKK β and their suggested different functional roles has been supported by several types of cellular study. Initial finding confirmed that IKK α is not required for upstream activation of the canonical NF κ B pathway by pro-inflammatory stimuli. TNF- α , IL-1 β and LPS can still

activate the IKK complex to induce IkB- α degradation and p65 nuclear translocation in IKK $\alpha^{-/-}$ MEF, primary keratinocytes and liver tissue (Hu et al., 1999). Similar results were achieved in different cell types, inhibiting IKK α kinase activity did not affect TNF- α induced IkB- α degradation in HUVECs (Denk et al., 2001), vascular smooth muscle cells (MacKenzie et al., 2007) and MEFs (Solt et al., 2009). In addition, Yang and co-workers demonstrated that deletion of IKK α did not affect LPS-induced p65 phosphorylation in MEFs (Yang et al., 2003). Interestingly, contradictory studies showed that both TNF- α and IL-1 β induced NFkB/DNA binding could be reduced in IKK $\alpha^{-/-}$ MEFs (Hu et al., 1999, Solt et al., 2009), while, Catley and co-workers demonstrated that inhibiting IKK α kinase activity through expression of a K44A inactive mutant resulted in a failure to induce NFkB driven luciferase reporter activity in response to TNF- α or IL-1 β human pulmonary A549 cells (Catley et al., 2005). Taken together, these studies suggested the potential of a cell type dependency in regulation of the NFkB pathway by IKK α .

A number of studies have shown that despite IKKa playing no role in regulating p65 nuclear translocation and DNA binding, the expression of several p65 dependent genes could be regulated by IKKa including MCP-1, IL-8, ICAM-1, VCAM-1, E-selectin and cIAP-2. (Denk et al., 2001, Gloire et al., 2007, Huang et al., 2007). These results suggested the potential for IKK α to regulate the canonical NF κ B pathway at a site downstream of the IKK/IκB-α/p65 complex. In 2003, Anest and co-workers discovered TNF-α induced IKKα translocation into the nucleus from the cytoplasm (Anest et al., 2003). They showed that a requirement for IKKa in mediating H3 Serine 10 phosphorylation at different NFkBdependent promoters such as IkB-a and IL-6 (Anest et al., 2003). Another study confirmed a role for IKKa cytokine-induced phosphorylation and subsequent acetylation of specific residues in histone H3 and in addition, revealed that IKKa interacts with CREB-binding protein (CBP) within the nucleus and in conjunction with p65, is recruited to $I\kappa B-\alpha$ and IL-8 promoters (Yamamoto et al., 2003). Later, Huang and co-workers found that the IKKa can also down-regulate the expression of p53 dependent genes. They found IKKa-mediated CBP phosphorylation not only switches the binding preference of CBP from p53 to p65, but also causes redistribution of CBP recruitment from p53 responsive to p65 responsive promoters, thereby enhancing NFkB dependent genes expression such as ICAM-1 and cIAP-2 (Huang et al., 2007).

Other more recent studies have implied that some of the actions of IKK α may be independent of kinase activity. In IKK $\alpha^{-/-}$ MEFs, reconstitution with DN-IKK α allows gain of function for a subset of genes. Relative to wild type IKK α , this subset was small but nevertheless suggest that IKK may function as a scaffold protein (Li et al., 2002). Similar data was obtained by Massa and co-workers, expression of a kinase inactive IKK α mutant (K44M) in IKK $\alpha^{-/-}$ MEF rescued expression of approximate 28% of NF κ B genes (Massa et al., 2005). These observations can also be found in other cell types, Hu and co-workers showed that IKK α regulates the secretion of an autocrine factor that induces keratinocyte differentiation, independent of its kinase activity (Hu et al., 2001). Overall, the studies outlined above confirmed a role for IKK α in the regulation of the canonical pathway but indirectly and at the level of p65-dependent gene transcription.

1.3.3.8.3 The role of IKKy in the regulation of canonical NFkB pathway

Whilst the function of IKK α and β have been studied in great detail, IKK γ has been examined to a much lesser extent. IKK γ was purified and found to have an overall protein structure somewhat different from either IKK α or IKK β . Molecular cloning and sequencing indicated that IKK γ contains several potential coiled-coil motifs (Rothwarf et al., 1998). IKK γ is a 48kDa non-catalytic protein, as described above; it plays a regulatory role and is required for IKK complex activation (Rothwarf et al., 1998, Yamaoka et al., 1998). Early studies showed mutation within the C-terminal results in failure to activate the IKK complex and hence NF κ B activity (Rothwarf et al., 1998). Later, May and co-workers found that IKK γ interacts with a domain within IKK β , designated the NEMO-Binding Domain (NBD), to regulate the formation of the IKK complex (May et al., 2000, May et al., 2002). More recent evidence suggests that the NBD is essential for IKK complex activation and regulation of canonical NF κ B activity. Furthermore, a mutation in the NBD results in failure to activate the canonical NF κ B pathway (Solt et al., 2009).

In contrast, mutating the NBD of IKK γ to prevent interaction with IKK α does not affect p100 processing, thus implicating that the association of IKK α and IKK γ is not involved in regulation of the non-canonical pathway (Solt et al., 2009). Activation of the IKK complex as a component of up-stream signalling required ubiquitination of IKK γ . A CBM complex consisting of the proteins CARMA1, Bcl10 and MALT1 is thought to be important in mediating ubiquitination of IKK γ . Following activation of the TNF- α or IL-1 β receptor,

Bcl10 is conjugated with K63-linked ubiquitin chains and associates with IKK γ (Wu and Ashwell, 2008). This mechanism allows phosphorylation of IKK α/β and subsequent translocation of p65/p50 into the nucleus.

1.3.3.9 The up-stream activation of the Non-Canonical NF_KB pathway

Concomitant with analysis of the classic canonical pathway is the study of IKK α and β in the regulation of other NFkB pathways. The non-canonical pathway is classified as one of these atypical pathways prevalent in a restricted number of cell types and is characterised by p100/p52 processing. Figure 1.4 shows the important kinases and proteins involved in the pathway and some of the cognate agonists involved. The non-canonical signalling pathway is activated by TNFR superfamily members, including B cell-activating factor receptor (BAFFR) (Claudio et al., 2002, Kayagaki et al., 2002), CD40 (Coope et al., 2002), lymphotoxin β receptor (LTβR) (Dejardin et al., 2002), RANK (Novack et al., 2003), TNFR2 (Rauert et al., 2010). An early study showed that Receptor activator of nuclear factor kappa-B ligand (RANKL) can activate the non-canonical NFkB pathway through processing of p100 to p52 through RANK in osteoclast precursors (Novack et al., 2003) to control differentiation (Teitelbaum, 2000). Both TNF receptor-associated factor 2 (TRAF2) and TNF receptorassociated factor 3 (TRAF3) are recruited to the receptor complex following ligand ligation (Bishop and Xie, 2007), then both TNAF2 and TNAF3 undergo degradation which leads to activation of NIK and the further downstream IKKa itself. Similar to TAK-1, NFkB-inducing kinase (NIK) is also a MAP3K member which activates the IKK complex but in particular, IKKα dimers, to subsequently regulate p100 processing (Xiao et al., 2001; Senftleben et al., 2001). Ling and co-workers showed that NIK phosphorylated IKKα at residue serine-176. A mutated form of IKKa containing alanine at residue 176 could not be phosphorylated or activated by NIK and inhibited IL-1 or TNF- α -induced NF κ B activation (Ling et al., 1998).

NIK itself may further regulate p100 processing through ubiquitination (Xiao et al., 2001). p100 contains a region called the NIK-responsive domain (NRD). Substrate-binding subunit $SCF^{\beta TrCP}$ is an ubiquitin ligase complex and the $\beta TrCP$ component of the ligase complex binds to the NRD site of p100 to induce ubiquitination (Fong and Sun, 2002). Later, scientists identified amino acid lysine-856 within the substrate binding subunit of p100 as crucial in the ubiquitination and processing of p100 (Amir et al., 2004). It should also be noted that downstream intermediates can also regulate upstream components of the non-canonical pathway. For example, IKK α phosphorylates NIK and attenuates the stabilisation of NIK, resulting in a decrease in non-canonical NF κ B signalling. It is believed that IKK α destabilises NIK to prevent over activation of non-canonical NF κ B signalling (Razani et al., 2010). The mechanisms underpinning IKK α negative feedback regulation of Rel-B dependent gene expression and the involvement of the other upstream mediators of the non-canonical pathway are not yet fully understood, but it is possible that the phosphorylation of NIK can be considered a biomarker of IKK α activation.

1.3.4 The role of IKKa in the regulation of the Non-Canonical NFkB pathway

A number of studies using deletion mice (Luftig et al., 2004, Muller and Siebenlist, 2003) or other approaches clearly demonstrated a role for IKK α in the non-canonical pathway and at the same time excluded the involvement of IKK β (Claudio et al., 2002, Dejardin et al., 2002, Liang et al., 2006, Senftleben et al., 2001). In addition, no role for NEMO was identified, suggesting a lack of requirement for the classical IKK complex. As mentioned above, activation of this pathway requires both ubiquitination and phosphorylation of p100, and indeed Liang and co-workers found that IKK α phosphorylates p100 within residues serine 866 and serine 870 and this is crucial for p100 processing (Liang et al., 2006). Following expression of a dominant-negative isoform of IKK α , p100 processing is inhibited and p52 formation is reduced (Luftig et al., 2004, Muller and Siebenlist, 2003). Therefore, IKK α is essential for the formation of the Rel B/p52 heterodimer which subsequently translocates into the nucleus to mediate gene transcription.

A number of studies also link IKK α to a specific subset of genes. There are several studies showing that inhibiting IKK α can affect Rel B dependent genes expression. Dejardin and coworkers showed that in MEFs inhibiting IKK α activity leads to reduction in expression of both chemokine and cytokine genes including SLC, BLC, ELC, SDF-1 and BAFF (Dejardin et al., 2002). In addition, Bonizzi and co-workers confirmed that IKK α is required for activation of the non-canonical NF κ B signalling pathway again based on p100 processing. They also found that the promoters of IKK α dependent genes are recognised by Rel B/p52 dimers but not by p65/p50 (Bonizzi et al., 2004). Later, it was shown that IKK α regulates the cell cycle gene S-phase kinase-associated protein (skp2) which is RelB/p52 dependent (Schneider et al., 2006). These results provided strong evidence for a crucial role for IKK α in the regulation of non-canonical NF κ B signalling pathway.

More recent studies have identified other functions of IKK α particularly in relation to cell cycle regulation and apoptosis. IKK α has been shown to regulate cyclin D1 but the findings are contradictory. On the one hand, a number of studies demonstrate that IKK α and Rel B/p52 axis up-regulates the expression of cyclin D1 (Cao et al., 2001, Demicco et al., 2005). On the other hand, other studies have shown that IKK α down-regulates cyclin D1 expression. Song and co-workers also reveal that IKK α plays a critical role in mediating UVB-induced G0/G1 cell cycle arrest by suppressing cyclin D1 expression in MEFs (Song et al., 2010). The mechanism of IKK α induced cyclin D1 degradation has been investigated by Kwak and co-workers. They show that IKK α can regulate the subcellular localization and proteolysis of cyclin D1 by phosphorylation at Threonine-286. (Kwak et al., 2005).

A recent study has not uncovered the potential for Aurora-A to be regulated by IKK α . Aurora-A is a member of mitotic kinases that localise to centrosomes. Aurora-A kinases have been implicated in centrosome separation and spindle assembly (Bischoff and Plowman, 1999, Giet and Prigent, 1999, Nigg, 2001). The activity of Aurora A kinase peaks during the G₂/M phase and coincides with phosphorylation of IKK α (Bischoff and Plowman, 1999). IKK α facilitates the regulation of M phase of the cell cycle by modulating Aurora A phosphorylation and activation (Prajapati et al., 2006). Further studies are required to fully characterise the independence of NF κ B regulation and signaling of IKK α .

1.4 The role of Non-Canonical NFkB pathway in regulation of inflammation

A substantial number of studies implicate a role for the canonical NF κ B pathway and IKK β in the regulation of inflammation and these are examined in the context of diseases such as arthritis, inflammatory bowel disease and atherosclerosis (see Section 1.5). Far fewer studies have examined a role for IKK α . Nevertheless, indirect evidence is now accumulating to link IKK α with the negative regulation of acute inflammation but with a positive role in chronic inflammation through an interaction with the adaptive immune system.

With identification of the TLRs, a large number of studies have conclusively established an important role for the IKK β /I κ B- α /p65 axis (NF κ B canonical signaling pathway) in response

to a vast array of infectious agents. In contrast, a recent study has shown that IKKa activation limits the inflammatory response to bacterial infection in vivo and inhibits activation of NFkB in primary macrophages in vitro (Lawrence et al., 2005). Macrophages from transgenic mice expressing catalytically-inactive IKKa were found to be resistant to pathogen-induced apoptosis, due to elevated expression of anti-apoptotic genes, and showed increased expression of pro-inflammatory cytokines such as TNF- α (Lawrence et al., 2005). In cellular studies, IKK α has been found to negatively regulate the canonical NF κ B pathway and indeed, IKKα knockout macrophages derived from foetal liver stem cells showed an increase level of p65 dependent pro-inflammatory cytokines production and an enhanced ability to stimulate T-cell proliferation (Li et al., 2005). Similarly, another research team also demonstrated increased expression of pro-inflammatory cytokines and chemokines upon stimulation with LPS in IKKa knockout fibroblasts (Xia et al., 1997). Interestingly, the results of IKKa knockout studies can sometimes be different from those utilising DN-IKKα. For instance, the activity of IKK β is elevated in IKK α knockout cells, but this cannot be observed in cells overexpressing DN-IKK α (Lawrence et al., 2005). This may be due to the formation of an IKK β homodimer with increased NF κ B activity which only occurs in absence of the IKK α protein.

In general, studies showing a direct role for IKK α in other aspects of inflammation linked to immune function are lacking, however some information has accumulated via the study of Rel B function, which is downstream of IKK α . Rel B-deficient mice died of multi-organ inflammation (Weih et al., 1995, Xia et al., 1997), due to autoreactivity of endogenous Tcells, subsequently causing abnormal development of the thymus and the breakdown of immunological tolerance. Rel B has also been shown to inhibit I κ B- α stability and therefore limit NF κ B activation (Xia et al., 1999). Moreover, Rel B may negatively regulate canonical NF κ B complexes in the nucleus through protein-protein interactions with Rel A (Jacque et al., 2005). At present, it is unclear how these cellular studies link to *in vivo* affects within their immune system.

Overall it is accepted that the non-canonical NF κ B pathway plays an important role in the regulation of immunity. Deletion of Rel B and NF κ B-2 inhibits the ability of the antigen presenting cell (APC) to activate T cells in defending against foreign bacteria and virus (Gerondakis et al., 1999). Moreover, a Rel B deficiency leads to immune failure and uncontrolled spread of the vaccinia virus in lung and liver (Freyschmidt et al., 2007). The

activators of the TNF superfamily, CD40 ligand, RANKL or osteoclast differentiation factor (ODF), are expressed by the activated T cell. Their interaction with receptors on APC including CD40 and RANK, subsequently induces Rel B activation that promotes survival and augments the ability of the APC to simulate T cell proliferation and activation, conceivably by up-regulating the expression of NF κ B-dependent molecules MHC class 2, CD80, and CD86 (Anderson et al., 1997, Li et al., 1999a, Verhasselt et al., 1999, Wong et al., 1999).

1.5 NFkB and inflammatory diseases

Rheumatoid arthritis is a chronic, systemic inflammatory disease causing joint destruction, and is associated with increased morbidity and mortality (Firestein, 2003, McGorisk and Treasure, 1996, Smolen and Steiner, 2003). Evidence suggests that NF κ B can play a central role in many aspects of the disease. Activated NF κ B has been detected in human synovial tissue during the early stage of joint inflammation (Gilston et al., 1997). Moreover, the p65/p50 complex was found to be located in the endothelium and synovial lining of rheumatoid arthritis patients, particularly in CD14-positive cells, but not in the normal synovium (Handel et al., 1995).

In fact, clinical studies outlined above have been supported by a number of cellular animal studies utilising both pharmacological and molecular studies. Han and co-workers found increased synovial NF κ B-DNA binding precedes the development of joint involvement in murine collagen-induced arthritis (CIA) and gradually increases during the evolution of disease (Han et al., 1998). In addition, synovial NF κ B activation also occurs within a few days after immunization in rat adjuvant arthritis (Tsao et al., 1997). Furthermore, an intra-articular injection of Adv. DN-IKK β reduces nuclear translocation of NF κ B (p65/p50) and clinical synovitis in adjuvant arthritis in rats (Tak et al., 2001). However, inhibiting IKK β also has the potential to worsen the inflammatory condition by reducing the synthesis of anti-inflammatory agents such as cyclopentenone prostaglandins (Rossi et al., 2000) and COX-2 (Chang et al., 2004). Evidence suggests that COX-2 also has anti-inflammatory effect (Gilroy et al., 1999).

Dysregulated cytokine production and signalling mechanisms by intestinal epithelial cells, lymphocytes and macrophages have been implicated in the pathogenesis of inflammatory

bowel disease (IBD). An early study showed there was an increase of NFκB expression in mucosal macrophages from IBD patients due to the over-expression of the cytokines such as TNF- α , IL-1 β and IL-6 (Neurath et al., 1996). Furthermore, NF- κ B-induced TNF- α was in turn able to potentiate the activation of NF- κ B, thereby providing a kind of positive feedback in these cells (Hart et al., 1998, Holtmann and Neurath, 2004). High cellular NF κ B activity was also observed in gastric epithelial cells under the pathogenic condition of helicobacter pylori-associated gastritis (van Den Brink et al., 2000). Similarly, a clinical study also showed high levels of NF κ B DNA-binding activity accompanied by an increased production of IL-1, IL-6 and TNF- α in lamina propria macrophages during Crohn's disease (CD) and ulcerative colitis (UC). Moreover, the expression of NF κ B isoforms including p50, p65 and c-Rel was found to be dramatically high in patients with CD and UC. However, the production of pro-inflammatory cytokines was remarkably reduced using specific antisense phosphorothioate oligonucleotide to down-regulate p65 expression in macrophages during inflammatory bowel disease.(Neurath et al., 1998).

NFκB also plays an important role in several other tissue specific models of inflammation. Activation of NFκB leads to production of pro-inflammatory cytokines and adhesion molecules in nephritic glomeruli. Therefore uncontrolled NFκB regulation may cause the development of glomerulonephritis. A recent study showed that using a potent NFκB inhibitor, pyrrolidine dithiocarbamate (PDTC), glomerular NFκB-DNA binding activity was reduced and subsequently caused reduction of IL-1 β , MCP-1, ICAM-1 and iNOS mRNA expression (Sakurai et al., 1996). PDTC also prevented urinary protein excretion which is a pathophysiological parameter for glomerulonephritis (Sakurai et al., 1996). It was also demonstrated that c-Rel knockout mice have decreased airway hyperreponsiveness and eosinophil infiltration as well as lower levels of serum immunoglobulin (IgE) in an allergeninduced asthma model (Donovan et al., 1999). Moreover, inflammatory airway disease in humans was found to be strongly related to the expression of cytokine and adhesion molecules elevated due to the activation of NFκB. Furthermore high levels of nuclear NFκB were observed especially in airway epithelial cells and led to an increased expression of pro-inflammatory cytokines, chemokines, iNOS and COX-2 (van Berlo et al., 2010).

Overall these observations support the notion that NF κ B plays an important role in many different inflammatory disorders. However, the vast majority of studies implicate only IKK β

in these mechanisms and additional studies are required which focus on the non-canonical $IKK\alpha$ axis.

1.6 Atherosclerosis

Atherosclerosis is a chronic inflammatory condition of the blood vessels. The condition is associated with plaque formation in the sub-endothelial layer, denudation of the endothelial layer and thrombus formation. Subsequent destabilisation of the plaque and rupture leads to blockage of vessels and result in adverse cardiovascular events, heart attack and stroke. This process is initiated by migration of monocytes into the subendothelial space. Monocytes differentiate into macrophages and eventually become lipid-laden foam cells due to take up of modified LDL via lipoprotein molecules circulating in the blood stream. Many studies suggest Apolipoprotein E (ApoE) prevents atherosclerosis by clearing cholesterol-rich lipoproteins from plasma. Under physiological condition, ApoE binds to the cell surface lowdensity lipoprotein (LDLR) or LDLR-related protein (LRP) and initiates receptor-mediated endocytosis to prevent cellular uptake of LDL (Mahley and Rall, 2000). Reducing the expression of ApoE in human patients has been found to be associated with proatherogenic lipoprotein profile and atherosclerotic disease development (Ghiselli et al., 1981, Schaefer et al., 1986). Lipoprotein accumulation, matrix degradation and monocytes migration are the major factors that initiate the atherosclerotic plaque development (Saxena and Goldberg, 1994).

1.6.1 Development of atherosclerosis

The endothelial cell plays a key role in the development of atherosclerosis. These cells respond to a number of pro-inflammatory cytokines including TNF- α and IL-1 β to increase the expression of inflammatory mediators such as IL-6, COX-2, ICAM-1, VCAM-1 and MMPs. Furthermore, there is a concomitant reduction in the release of protective mediators such as PGI2 and NO. Therefore, control of the inflammatory status of endothelial cells, which is achieved by a balance of pro- and anti - inflammatory signals, is crucial to limiting the disease (Hoefen and Berk, 2002). Initiation of atherosclerosis also involves the release of oxygen radicals and lipid mediators which can be susceptible to peroxidation and can also activate endothelial cells (Reddy et al., 2002). LDL which accumulates within the vessel wall is also subject to oxidation which allows endothelial cell activation (Reddy et al., 2002). These events are also related to a number of other diseases of the cardiovascular system

including diabetes. These conditions are always associated with an increase in endothelial dysfunction leading to enhance apoptosis. This will be discussed in more detail in Section 1.6.5.

1.6.2 Adhesion molecules and the development of Atherosclerosis

Migration of monocytes into the sub-endothelial space is essential for the initial stages of the atherosclerotic lesion development as they differentiate into macrophages and take up oxidized LDL. This leads to the formation of foam cells. As outlined previously, migration of monocytes is dependent on the expression of a number of different cell adhesion molecules (CAM) on the endothelial cell surface. There are two main families expressed in endothelial cells including; P and E selectins and the member of immunoglobulin superfamily, ICAM and VCAM. Regulation of CAM expression is crucial and all three types of CAM acting cooperatively in the process of monocyte transmigration into sub-endothelial space (Chia, 1998). Selectins are the first class CAM which regulates the initial weak attachment of monocytes to the endothelium by a process called "rolling", which is a low-affinity binding of selectin to carbohydrate molecules on the monocyte (Dong et al., 1998). E-selectin expression is observed in the atherosclerotic region of blood vessels but not in normal segments (O'Brien et al., 1996), therefore it is more prevalent in atherosclerotic plaque. Additionally, there is an increase in arterial segments showing mononuclear leukocyte accumulation where E-selectin is also expressed. This suggests E-selectin plays a role in monocyte migration in lesion development (O'Brien et al., 1996). ICAM-1 and VCAM-1 are identified as a second class of CAM. These molecules mediate the formation of firm endothelium-monocyte adhesion via interacting with integrins which are expressed on monocytes (Elices et al., 1990, Marlin and Springer, 1987). Moreover, there is evidence which links ICAM-1 deficiency to reduced atherosclerotic lesions in ApoE/ICAM-1 doubleknockout mice (Bourdillon et al., 2000), suggesting that both ICAM-1 and VCAM-1 play crucial roles in monocyte recruitment and initial lesion development. The third class of CAM, the integrins, are expressed on leukocytes. Integrins act as ligands for ICAM-1 and VCAM-1; β2 integrin, LFA-1 and Mac-1 bind to ICAM-1 while VLA-4 is specific for VCAM-1 (Elices et al., 1990, Marlin and Springer, 1987).

In addition to expression of adhesion molecules, a number of other factors are required for the migration of monocytes into the sub-endothelial space. Following firm adhesion, monocytes transmigrate through the endothelial layer into the tunica intima. This process requires chemokines such as monocyte chemoattractant protein-1 (MCP-1) (Harrington, 2000). MCP-1 is classified as the C-C subfamily of chemokines and acts as a potent monocyte chemoattractant when expressed on vascular endothelial cells. MCP-1 was shown to be expressed on atheromatous lesions (Tanaka et al., 2003), and regulates monocyte chemotaxis, which draws firmly attached monocytes into the arterial intima through interaction with its ligand on monocytes, chemokine receptor-2 (CCR-2) (Hemmerich et al., 1999, Salcedo et al., 2000).

After monocytes migrate into sub-endothelial spaces, they differentiate into macrophages in response to monocyte colony stimulating factor (M-CSF) (Krishnaswamy et al., 1999, Sakai et al., 2000). They express receptors that specifically recognize oxidized LDL, including scavenger receptor A (de Villiers and Smart, 1999) and CD36 (Nicholson et al., 2000, Silverstein and Febbraio, 2000) and produce pro-inflammatory cytokines such as TNF- α and IL-1. Inflammation gives rise to lipid peroxidation, and lipid peroxidation in turn results in more inflammation, triggering the first step in the development of an atherosclerotic lesion (Kutuk and Basaga, 2003). In addition to a role for adhesion molecules, evidence implicates an involvement of a family of proteolytic protein known as the matrix metallo-proteinases (MMPs). This comprises at least 12 MMP isoforms and endogenous inhibitors known as tissue inhibitors of metalloproteinases (TIMPs). A recent study shows that inhibition of MMP-12 prevents atherosclerotic plaque development and increases stability in ApoE^{-/-} mice (Johnson et al., 2011). As MMPs play a significant role in the degradation of proteins within the extracellular matrix (ECM) (Ra and Parks, 2007), overexpression of MMPs allow vascular smooth muscle cells to transmigrate into the intima during atherosclerosis. Moreover, there is a study which demonstrates that MMP-2 is responsible for vascular smooth muscle cell proliferation in response to oxidized LDL (Auge et al., 2004) which is strongly associated with atherosclerotic development. Figure 1.5 shows the cellular mechanisms underpinning the development of atherosclerosis.

1.6.3 NFkB and atherosclerosis

Given that inflammation is a key aspect of atherosclerosis development, it is perhaps not surprising that a considerable body of evidence supports a role for NF κ B in this process. Activation of NF κ B is associated with the development of early lesions (Kutuk and Basaga,

2003). In addition, using immunofluorescence and immunohistochemistry, early studies showed activated NF κ B in macrophages, endothelial cells and smooth muscle cells located in the media and intima of thickened fibrous atherosclerotic regions of the vessel wall. By contrast, little or no activation of NF κ B was detected in otherwise healthy parts of the arteries (Brand et al., 1997a, Brand et al., 1996, Brand et al., 1997b). Later, another research team demonstrated activation of NF κ B in the intimal layers of human coronary artery plaques (Wilson et al., 2002).

There is accumulating evidence to suggest that inhibiting NFkB can attenuate atherosclerosis. In apoE/LDLR double knockout mice, lesion development was reduced by an NFkB inhibitor, PDTC (Jawien et al., 2005). A similar study also showed that ablation of IKK γ or expression of a dominant-negative I κ B- α , specifically in endothelial cells, resulted in strong reduction of atherosclerotic plaque formation in ApoE^{-/-} mice fed with a cholesterol rich diet (Gareus et al., 2008). This effect was associated with reduced adhesion molecule induction in endothelial cells and impaired macrophage recruitment to the atherosclerotic plaque (Gareus et al., 2008). This data suggests that inhibiting the NFkB pathway may attenuate atherosclerosis, however, the situation is made more complicated by the fact that inhibiting NFkB in certain cell types may also aggravate atherosclerosis. One knockout study has demonstrated that inhibition of the NF κ B pathway by removal of IKK β in macrophages lead to an increase in lesion size. Moreover, the condition of the lesion became more severe due to increased necrosis and more macrophages in the early lesions (Kanters et al., 2003). Therefore, activating the NFkB pathway may lead to expression of certain genes expected to ameliorate atherosclerosis such as IL-10 which mediates the condition. Therefore inhibiting NFkB may attenuate atherosclerosis if targeted specifically in certain cell types and to specific gene clusters.

An overwhelming number of cellular studies performed *in vitro* correlate with effect observed *in vivo*. In endothelial and vascular smooth cells and macrophages, NF κ B is strongly linked with ICAM-1 and VCAM-1 expression. At a cellular level, these adhesion molecules are well recognized to be expressed on vascular endothelial cells following NF κ B activation (Bunting et al., 2007, Denk et al., 2001, Huang et al., 2007, Iademarco et al., 1992, MacKenzie et al., 2007, van de Stolpe et al., 1994). Other molecules intimate to cellular migration such as MCP-1 are also regulated by canonical NF κ B. Furthermore, NF κ B regulates the expression of many cytokines (TNF- α and IL-1 β) and chemokines involved in

the initiation of atherosclerotic lesions including monocyte recruitment into the vessel wall (Cominacini et al., 2000, Rahman et al., 1998, Zhu et al., 1999).

1.6.4 Oxidized LDL and NF_KB

Atherosclerotic lesions are also regulated by further oxidation of lipids and further secretion of inflammatory mediators by macrophages (Saxena and Goldberg, 1994). For initiation of atherosclerotic lesions development, lipid deposition is necessary and is preceded by initial LDL oxidation. Several studies have demonstrated that oxidized LDL strongly activates NF κ B pathway in monocytes resulting in an enhance release of pro-inflammatory cytokines including TNF- α , IL-1 β , IL-8 and IL-12 (Brand et al., 1997a, Janabi et al., 2000, Mytar et al., 2004). The oxidized LDL has prominent pro-inflammatory properties (Reddy et al., 2002) and might act as an early mediators of chronic inflammatory reactions through activation of NF κ B. This results in induction of the expression of cytokines and cell adhesion molecules which are important factors for the migration of monocytes into the sub-endothelial space (Cushing et al., 1990, Rajavashisth et al., 1995). In addition, Parhami and co-workers also demonstrated that minimally oxidized LDL, activates NF κ B in endothelial cells (Parhami et al., 1993).

Oxidized LDLs are thought to exert biphasic effects on NF κ B activity. Low concentrations of oxidized LDL promote inflammatory effects by up-regulating inflammatory gene expression via NF κ B activation, however, high levels of oxidized LDL mediates immunosuppressive effects by inhibiting NF κ B activation in response to inflammatory agents such as LPS (Robbesyn et al., 2004). Indeed, Wang and co-workers have shown that incubating HUVECs with LPS causes inflammatory gene expression via TLR4/NF κ B activation and induces the expression of TLR4, LOX-1, ICAM-I, and E-selectin, and increases monocyte adhesion to endothelium (Wang et al., 2005). These data, together with other works which showed that increasing NF κ B can prevent oxidised LDL induced apoptosis (Nishio et al., 2000), suggests that NF κ B has the potential to act as a double-edged sword in relation to oxidized LDL function and atherosclerotic development.



Figure 1.5 Cellular mechanism regulating atherosclerosis. The diagram shows the expression of adhesion molecules facilitate the transmigration of monocyte into subendothelial layer. Adopted from (Libby et al., 2010a)

1.6.5 Atherosclerosis development and endothelial dysfunction and apoptosis

A key cellular event in endothelial dysfunction and implicated in atherosclerosis is apoptosis. Apoptosis is controlled programmed cell death which is regulated by a number of key signaling events. Many classic atherosclerotic risk factors such as oxidized LDL (Dimmeler et al., 1997), increased oxidative stress (Lizard et al., 1997) and angiotensin II (Dimmeler et al., 1997) cause endothelial cell apoptosis. Furthermore, the inflammatory environment within the atherosclerotic plaque leads to expression of iNOS which gives rise to a high level of NO and oxygen radicals (Buttery et al., 1996, Luoma et al., 1998, Wilcox et al., 1997). In this situation NO itself could induce apoptotic cell death and destabilise the atherosclerotic plaque (Beckmann et al., 1994, Foresti et al., 1999). Moreover, high concentration of pro-inflammatory cytokines such as TNF- α and IL-1 β found in the atherosclerotic lesion may also contribute to endothelial apoptosis (Frostegard et al., 1999).

Despite these findings, the involvement of NFkB in regulating apoptosis in the context of atherosclerosis is problematic. Whilst inhibiting the expression of inflammatory proteins such as ICAM-1 (van de Stolpe et al., 1994) and VCAM-1(Iademarco et al., 1992) through blocking the canonical pathway may be beneficial, this is detrimental to endothelial cell survival. NFkB regulates pro-survival proteins such as bcl-2 (Catz and Johnson, 2001) and cFLIP (Kreuz et al., 2001) and also anti-apoptotic proteins such as X-linked inhibitor of apoptosis protein (XIAP) (Turner et al., 2007) and porcine inhibitor of apoptotic protein (PIAPs) (Stehlik et al., 1998a). Moreover, VEGF expression is up-regulated by TNF-a induced activation of NFkB pathway which aids endothelial cell survival. However VEGF is also linked to increased adhesion molecule expression which would indicate a role in atherosclerotic development. Furthermore NFkB can itself contribute to endothelial cell apoptosis through the induction of iNOS in macrophages and enhanced formation of peroxynitrite (Cromheeke et al., 1999). Bombeli and co-workers have shown that apoptotic HUVECs become pro-coagulant via increased expression of phosphatidylserine (PS) and loss of anti-coagulant membrane components (Bombeli et al., 1997), suggesting that apoptosis can accelerate the development of atherosclerosis.

These studies are relevant when considering the therapeutic targeting of the canonical NF κ B pathway. Blocking p65 NF κ B translocation or direct pharmacological inhibition of IKK β will often result in endothelial cell apoptosis which will be detrimental to vessel integrity.

Therefore approaches which involve inhibiting the expression of pro-inflammatory molecules such as ICAM-1 and VCAM-1 whilst leaving pro-survival genes unaffected would be desirable. Inhibition of IKK α may therefore be an attractive approach however, it is not supported as yet by studies demonstrating a role for IKK α in the development of atherosclerosis.

1.7 Angiogenesis

1.7.1 Introduction to Angiogenesis

Angiogenesis is the process of vascular growth by sprouting of pre-existing vessels. It is subject to a complex control system with pro-angiogenic and anti-angiogenic factors. In adults, angiogenesis is strictly regulated. The formation of new blood vessels occurs during wound healing, organ regeneration, and in the female reproductive system during ovulation, menstruation, and the formation of the placenta in normal circumstances. Uncontrolled angiogenesis is strongly related to several pathological processes such as cancer development, rheumatoid arthritis, diabetic retinopathy and psoriasis (Hoeben et al., 2004, Munoz-Chapuli et al., 2004). Angiogenesis involves a large number of endothelial cell processes, therefore many extracellular signals are involved. A growth factor central to angiogenesis is VEGF. VEGF virtually regulates almost every aspect of angiogenesis involving endothelial cells including; proliferation (Kanno et al., 2000, Wu et al., 2000), survival (Nor et al., 1999), migration (Favot et al., 2003), extracellular matrix degradation (Funahashi et al., 2011), differentiation and morphogenesis (Stratman et al., 2011). Other growth factors can also play a role, for instance, activation of FGF receptor signalling mechanisms can trigger endothelial cell proliferation (Cabrita and Christofori, 2003, Eliceiri et al., 1999, Hanafusa et al., 2002). Angiopoietin receptors, also called Tie receptors, are also important in regulating cell migration (Carlson et al., 2001). Activation of PI3K pathways through these receptors can also up-regulate the pro-survival genes such as Bcl-2 to enhance endothelial cell integrity (Catz and Johnson, 2001).

1.7.2 NFkB and Angiogenesis

The role of NF κ B in regulation of angiogenesis remains unclear. On the one hand, NF κ B regulates the expression of proteins which may be involved in the angiogenic process such as MMPs 2, 3 and 9 (Agarwal et al., 2005, Mountain et al., 2007, Popov et al., 2006), VEGF

(Grosjean et al., 2006, Ko et al., 2006), IL-6, MCP-1 (Denk et al., 2001) and IL-8 (Agarwal et al., 2005). On the other hand, NF κ B can also enhance the expression of anti-angiogenic factors such as Plasminogen Activator Inhibitor 1 (PAI-1) (Hou et al., 2004), thrombospondin-1 (TSP-1) (Yang et al., 2004) and Vascular Endothelial Growth Inhibitor (VEGI) (Xiao et al., 2005). A number of these genes can be differentially regulated by the canonical and non-canonical NF κ B pathway and this is discussed below. Therefore understanding the role of each of the NF κ B pathways could be beneficial for the treatment of cardiovascular diseases.

1.7.3 The Canonical NF_KB Pathway and Angiogenesis

As mentioned previously, the IKK β /p65 axis regulates expression of a large number of proteins which have been directly or indirectly linked to angiogenesis through effects on endothelial cell proliferation and survival. This includes not only XIAP and c-IAP (Chu et al., 1997, Stehlik et al., 1998a) but other proliferative genes such as Ephrin-A1 (Deregowski et al., 2002), IL-6 (Plumpe et al., 2000) and cyclin D1 (Hinz et al., 1999). Inhibition of p65 also results in a decrease in the expression of angiopoietin-1 (Scott et al., 2005) and 2 (Kim et al., 2000) both of which induced endothelial cell proliferation and enhance angiogenesis.

Endothelial cell migration which is intimate to angiogenesis is also coordinated by p65dependent regulation of both MMPs expression and the corresponding inhibitors of MMPs such as TIMP-1 (Wilczynska et al., 2006). Furthermore, Swiatkowska and co-workers found TNF- α induced the expression of PAI-1, a protein that negatively regulates plasmin formation and as a consequence angiogenesis via activation of the canonical NFkB signalling pathway (Swiatkowska et al., 2005). Activation of p65 by TNF- α can also inhibit tPA expression to further reduce plasmin levels (Ulfhammer et al., 2006). In contrast, there is some evidence to suggest that NFkB can inhibit angiogenesis. VEGI is up-regulated by p65 (Xiao et al., 2005), as is thrombospondin-1 (TPS-1), an anti-angiogenic agent which is responsible for inhibiting endothelial cell proliferation and migration, it has been demonstrated that p65 binds to the promoter sites of these genes to induce transcription activity (Yang et al., 2004). These results suggest that p65 may also negatively regulate extracellular matrix degradation and hence inhibit angiogenesis. In contrast to a large number of cellular studies, there is much less evidence demonstrating a role for the canonical pathway in angiogenesis *in vivo*. A study using Tie2 promoter/enhancer–I κ B α ^{S32A/S36A} transgenic mice which carried mutation in the critical serines at 32 and 36 in I κ B- α resulting in p65 inactivation revealed a striking increase in tumour vascularisation in transgenic mice compared to control (Kisseleva et al., 2006).

1.7.4 The Non-Canonical Pathway and Angiogenesis

Whilst the number of studies examining the role of the non-canonical pathway and in particular IKK α , in the regulation of angiogenesis are limited, there is nevertheless some evidence to support this hypothesis. In endothelial cells, lymphotoxin a known activator of IKKα was found to induce expression of pro-angiogenic factors, for instance, interleukin-8 (IL-8), chemokine (C-X-C) motif ligand 12 (CXCL12) (Liekens et al., 2010) and also increase VEGF synthesis and tubule formation (Liang et al., 2007). Furthermore, DeBusk and co-workers demonstrated that over-expression of IKKa in endothelial cells increased cell migration and also the number of vascular branch points (DeBusk et al., 2008). In addition, IKKa is also responsible for regulation of MMP-2, MMP-9, Urokinase plasminogen activator (uPA) and VEGF (Agarwal et al., 2005). These genes are along with IL-8, also responsible for cell migration and extracellular matrix degradation, so activation of the non-canonical pathway may be important in this process and as a result, angiogenesis. Overall, it would seem that the canonical NFkB pathway plays a role in angiogenesis through regulation of endothelial cell proliferation and survival, whilst the non-canonical pathway is involved predominantly in cellular migration and extracellular matrix degradation. This possibility, whilst requiring more studies suggests that these two processes could be targeted differentially using selective inhibitors. Figure 1.6 highlights the individual roles of both canonical and non-canonical NFkB pathways in the regulation of angiogenesis.



Figure 1.6 The role of the NF κ B pathways in the regulation of angiogenesis. The canonical NF κ B pathway and non-canonical NF κ B pathway play different role in the regulation of angiogenesis process. The canonical NF κ B pathway is responsible for cell proliferation and survival, whilst the non-canonical NF κ B is responsible for cell migration and extracellular matrix degradation.

1.8 The development and use of IKKa inhibitors

As described above, there is considerable evidence which supports a role for NF κ B in processes such as angiogenesis and chronic inflammatory diseases including rheumatoid arthritis and atherosclerosis. Therefore inhibition of NFkB may be an important therapeutic strategy in the treatment of those diseases. Several NFkB inhibitors have been developed for experimental and clinical use including PDTC (Liu et al., 1999), disulfiram (Wang et al., 2003), BAY 117082, MG132 (Dimmeler et al., 1999) and selective p65 NFkB inhibitor dehydroxymethylepoxyquinomicin (DHMEQ). However these agents have a wide range of actions within the pathway. As IKKs play a key role in the regulation of the NFkB pathway inhibiting specifically the IKKs would be beneficial in the treatment of a number of diseases. With the discovery of IKKs, a considerable amount of effort has been put into the development of small molecules IKK inhibitors. Currently the available compounds are directed to and selective for IKKB. The majority of these inhibitors are ATP-competitive molecules or alternatively possessing allosteric actions to limit IKK activities (Gamble et al., 2012). In addition, there are several natural products which can weakly inhibit IKK including curcumin, (-)-epigallocatechin-3-gallate, parthenolide and quercetin (Muller and Siebenlist, 2003), but these are not supplied appropriate lead molecules.

Castro and co-workers discovered the β -carboline compounds such as PS-1145 inhibited the phosphorylation of $I\kappa B \cdot \alpha$, NF κB activation and TNF- α expression by inhibiting IKK (Castro et al., 2003). Other classes of IKKβ inhibitors including N-(6-chloro-9H-β-carbolin-8-yl) N-(6-chloro-7-methoxy-9H-β-carbolin-8-yl)-2-methylnicotinamide (PS-1145) and nicotinamide (ML120B) also inhibited the expression of NFkB (p65/p50) dependent genes such as ICAM-1, IL-6, IL-8 GM-CSF, RANTES, MCP-1, GROa, NAP-2 and ENA-78. Furthermore, in vivo, these inhibitors were found to be as effective as synthetic steroids which inhibited NFkB indirectly for example dexamethasone. There is also therapeutic potential for these inhibitors in the treatment of inflammatory airway diseases, particularly in chronic obstructive pulmonary disease (COPD) or severe asthma (Catley et al., 2005). In relation to cancer, Gasparian and co-workers showed proteasome inhibitors such as epoxomicin and MG132 attenuated NFkB activity much more effectively than the IKK inhibitors BAY 11-7082 and PS-1145 in androgen-sensitive human prostate adenocarcinoma cells (LNCaP) (Gasparian et al., 2009). Overall evidence in this chapter has identified the potential of IKKa as a distinctive regulatory kinase in both canonical and non-canonical $NF\kappa B$ pathways and therefore may represent a novel drug target. Consequently development of selective IKK α inhibitors would be a potential break-through in terms of therapeutic targeting of disease.

1.9 Aims and Objectives

Inhibition of IKK β is a modern therapeutic approach in treatment of cancers and several inflammatory diseases. However, there is a limitation in the use of IKK β inhibitors for the treatment of atherosclerosis. The beneficial effect of an IKK β inhibitor would be to block the expression of inflammatory mediators, however this may also lead to a decrease in antiapoptotic protein expression. In the context of blood vessel integrity, endothelial cells would undergo apoptosis. This chapter has reviewed evidence supporting a role for IKK α in endothelial function. Although overall this role remains uncertain, there may be therapeutic potential if IKK α is predominant in the regulation of the inflammatory mediators rather than the cell death programme. In this thesis, the following aims will be addressed in two main result chapters.

1. Characterisation of the role of IKKα in regulation of NFκB pathways in HUVECs by:

i) Determining the role of IKK α in regulation of canonical NF κ B pathway by overexpression of DN-IKK α .

ii) Assessing the effect of IKK α in the regulation of the adhesion molecule expression following overexpression of DN-IKK α or knockdown of IKK α (siRNA) in HUVECs.

iii) Examining the function of IKK α in regulation of the non-canonical NF κ B pathway by overexpression of DN-IKK α or knockdown of IKK α (siRNA) in HUVECs.

2. Characterisation of the pharmacological effects of putative selective IKKα inhibitors by:

i) Examining the selectivity of IKK α inhibitors within the non-canonical NF κ B pathway.

ii) Assessing the inhibitory effect of p100 processing with the IKKa inhibitors.

iii) Assessing the inhibitory effect of adhesion molecules expression with the IKK α inhibitors.

CHAPTER 2

MATERIALS AND METHODS
2.1 MATERIALS

2.1.1 General Reagents

All materials used were of highest commercial purity available and were supplied by Sigma Aldrich Chemical Company Ltd. (Poole, Dorrset UK) or other reputable companies unless otherwise stated.

Bio-Rad Laboratories (Hertfordshire, UK)

Bio-Rad DCTM Protein Assay Dye Reagent Concentrate Pre-stained SDS-Page molecular weight markers.

Boehringer Mannheim (East Sussex, UK)

Bovine Serum Albumin (BSA , Fraction V)

Fisher Chemical UK Limited (Leicestershire, UK)

Bovine Serum Albumin (BSA, Fraction V)

Insight Biotechnology Limited (Wembley, UK)

Recombinant Human Interleukin-1 beta (IL-1β) Recombinant Human Tumour Necrosis Factor-alpha (TNF-α)

GE Healthcare (Buckinghamshire, UK)

Amersham Hybond ECL Nitrocellulose Membrane

Carl Roth GmbH + CO. KG (Karlsruhe, Germany)

Rotiphorese[®] Gel 30 (37.5:1) acrylamide

Whatmann (Kent, UK) Nitrocellulose Membrane, 3MM blotting paper

GIBCO BRL (Paisley, UK)

Foetal Calf Serum (FCS), L-glutamine, Minimal Essential Medium (x10), Non-essential amino acids, Penicillin/Streptomycin, Sodium Bicarbonate.

Corning B.V (Buckinghamshire, UK)

All tissue culture flasks, plates, dishes and graduated pipettes

Sarstedt AG & Co LTD (Leicester, UK)

Serological pipette 5ml Serological pipette 10ml

Thermo Fisher Scientific Inc (Surrey,UK)

Multidish 6 wells Multidish 12 wells

Roche Diagnostics Ltd (Burgess Hill, UK)

FuGENE® HD Transfection Reagent Dithiothreitol (DTT)

Lonza (Slough, UK)

Endothelial Cell Basal Medium-2 (EBM-2) Endothelial Growth Media (EGMTM-2) SingleQuots

YFP-tag IKKα YFP-tag NFκB p65

2.1.2 Adenoviruses

Clontech Laboratories Inc (Mountain view, CA, USA)

Adeno-X virus purification kit

Vector Biolabs (Philadelphia, USA) NFκB-Luciferase Human Adenovirus type 5 (Adv.NFκB-Luc)

Invitrogen Corp (California, USA) shRNA IKKα shRNA IKKβ The following adenoviruses were kindly provided by Dr D.Goeddel (Tularik Inc., CA, USA)

Adv. DN-IKKα Adv. DN-IKKβ

2.1.3 SU compounds (University of Strathclyde, Glasgow, UK)

SU937

SU1007

SU1010

SU1020

SU1053

2.1.4 Microscopy

Merck-Calbiochem (Nottingham, UK)

Mowiol

Molecular Devices Corp (Downingtown, PA, USA)

MetaMorph Imaging Series software (version 7.0 or 7.6.4)

Nikon Instruments (New York, USA)

Nikon TE-300 Epifluorescence microscope Nikon Eclipse TE2000-E inverted Epifluorescence microscope 40x or 100x oil-immersion Plan Fluor Objective Lens, NA=1.3

VWR International Ltd (Leicestershire, UK)

No. 0, (0.09-0.13 mm thick), circular glass 13 or 22 mm diameter coverslips. 0.8-1.0mm thick glass microscopy slides

2.1.5 Antibodies

Santa Cruz Biotechnology Inc (California, USA)

Rabbit monoclonal anti-IκBα (C-21) Rabbit polyclonal IKKα/β (H-470) Rabbit polyclonal NFκB p65 (C-20) Mouse monoclonal Actin (C-2) Goat polyclonal IKKβ (C-20)

Cell signaling Technology Inc (Hertfordshire, UK) Rabbit polyclonal p-p65 (Ser⁵³⁶) Rabbit polyclonal Histone H₃

Calbiochem (Nottingham, UK)

Mouse monoclonal anti-IKKa (14A231)

R&D Systems Europe Ltd (Oxon, UK)

Biotinylated Anti-human VCAM-1 (CD106) (BAF809) Biotinylated Anti-human ICAM-1 (BAF720)

Trevigen Inc (MD, USA)

TACS[®] Streptavidin-Horseradish Peroxidase Conjugate (Strep-HRP)

Jackson ImmunoResearch Laboratories Inc (PA, USA)

Horseradish peroxidase (HRP)-conjugated goat anti-rabbit IgG (111-035-144) Horseradish peroxidase (HRP)-conjugated sheep anti-mouse IgG Texas Red dye-conjugated AffiniPure donkey anti-goat IgG (705-076-147)

Millipore (U.K.) Limited (Watford, UK) Anti-NFκB p52 - 32534

2.2 CELL CULTURE

All cell culture work was performed in a class II cell culture hood under aseptic conditions. All the cells were grown in 75cm² flasks unless otherwise stated.

2.2.1 Human Umbilical Vein Endothelial Cells (HUVECs)

Cryopreserved Primary HUVECs (\geq 500,000 cells/vial) were purchase from Cascade Biologics. HUVECs were maintained in Endothelial Basal Media (EBM-2) supplemented with EGM-2 single Quots (2% Foetal bovine serum, 0.2ml Hydrocortisone, 2ml hFGFB, 0.5ml VEGF, 0.5ml R3-Insulin like Growth Factor-1, 0.5ml Human Epidermal Growth Factor (hEGF), 0.5ml Gentamicin, Amphotecerin-B (GA-1000), 0.5ml heparin and Ascorbic acid, 0.5ml). Cells were cultured at 37°C in a humidified atmosphere of 5% CO₂ and 95% air. Medium was changed every 2 days thereafter until cells became confluent.

2.2.2 Subculturing HUVECs by Trypsinisation

Cells were subcultured upon reaching approximately 90% confluency. The media was removed and the cells washed twice with a sterile solution of 0.5% (w/v) trypsin, 0.2% (w/v) EDTA in phosphate buffered saline (PBS). The solution was then removed and the flask placed in an incubator at 37°C, 5% of CO₂ and 95% air, for 2-5 min until cells began to round up, indicating that they had begun to detach from the flask. The flask was then gently tapped, to facilitate the cells to completely detach from the surface and then wash in EBM-2. Cells were then diluted 3 times with additional EBM-2 and transferred to fresh flasks (75 cm²) or seeded into plates (12 or 6 wells). Cells were maintained at 37°C in an incubator and the media replaced every 2 days until they reach 90% confluency and ready for stimulation). Cells were used for experimentation between passages 2 to 6 and were stimulated with different reagents in full media.

2.2.3 Human Embryonic Kidney (HEK293)

Low passage human embryonic kidney (HEK) 293 cells were maintained in Minimal Essential (MEM) with Eagle's salts supplemented with 10% (v/v) foetal calf serum (FCS), penicillin (250 units/ml), streptomycin (100 μ g/ml), L-glutamine (27mg/ml), 1x (v/v) non-essential amino acid and 0.375% (v/v) sodium bicarbonate (all GIBCO_®, Invitrogen Ltd). Cells were then incubated at 37°C in a humidified atmosphere with 5% CO2 with media replaced every 2 days. At 90% confluence, HEK 293 cells were passaged using 1 x sodium sodium citrate (SSC) solution (0.00878kg NaCl (1.5mM) and 0.00441kg (0.54mM) sodium acetate dissolved in 100ml of distilled water, then adjusted to pH 7.0). Cells were used for experimentation between passage 30 to 40.

2.3 WESTERN BLOT ANALYSIS

2.3.1 Preparation of Whole Cell Extracts

Cells were exposed to appropriate agonists or vehicle for the relevant period of time and they were then placed on ice to stop the reaction (protein degradation/ dephosphorylation). Cells were immediately washed twice with ice cold PBS before 150μ l of pre-heated laemmli's sample buffer (63mM Tris-HCl (pH6.8), 2mM Na₄P₂O₇, 5mM EDTA, 10% (v/v) glycerol, 2% (w/v) SDS, 50mM DTT, 0.007% (w/v) bromophenol blue) was added. The cells were then scraped and the chromosomal DNA sheared by repeatedly pushing through a 21 gauge needle. The cells were then transferred to Eppendorf tubes and boiled for 5min to denature the proteins in the samples, before keeping at -20°C or -80°C until use.

2.3.2 SDS-Polyacrylamide Gel Electrophoresis (SDS-PAGE)

Gel kit apparatus was first cleaned in 70% ethanol before assembly, then distilled water was added overnight to check the glass plates were flush and not leaking. Resolving gels were prepared containing an appropriate amount (7.5 (w/v) %, 8.5 (w/v) %, 10 (w/v) % 15 (w/v) %) acrylamide (Rotiphorese[®] Gel 30 (37.5:1), 0.375M Tris base (pH8.8), 0.1% (w/v), SDS and 10% (w/v) Ammonium persulfate (APS)). Polymerisation was initiated by the addition of 0.05% (v/v) N, N, N', N' tetramethylethylenediamine (TEMED). The solution was poured between 2 glass plates assembled in a vertical slab configuration according to the manufactures instruction (Bio-Rad) and overlaid with 180µl of 0.1% (w/v) SDS. Following gel polymerization the layer of 0.1% (w/v) SDS was removed and a stacking gel containing 10% (v/v) acrylamide: Rotiphorese[®] Gel 30 (37.5:1), 0.125M Tris base (pH6.8), 0.1% (w/v) SDS and 10% (w/v) Ammonium persulfate (APS) and 0.05% (v/v) TEMED) was poured directly on top of the resolving gel, and a teflon comb was immediately inserted into the stacking gel solution. After polymerization was complete, the comb was removed and the polyacrylamide gels were assembled in a Bio-Rad Mini-PROTEAN 3TM electrophoresis tank, with both reservoirs filled with electrophoresis buffer (25mM Tris, 129mM glycine, 0.1% (w/v) SDS). Aliquots of samples (20-30µg/ml) were then loaded into the wells using a microsyringe. A prestained SDS-PAGE molecular weight marker of known molecular weights was run concurrently in order to identify the polypeptide of interest. Samples were electrophoresed at a constant voltage of 120V, until the bromophoenol dye had reached the bottom of the gel.

2.3.3 Electrophoretic Transfer of Proteins to Nitrocellular Membrane

The proteins separated by SDS-PAGE were transferred to nitrocellulose membranes by electrophoretic blotting following a standard protocol (Towbin et al., 1979). The gel was pressed firmly against a nitrocellulose sheet and assembled in a transfer cassette sandwiched between 2 filter papers and two sponge pads. The cassette was immersed in transferred buffer (25M Tris, 19mM glycine, 20% (v/v) methanol) in a Bio-Rad Mini-PROTEAN 3TM electrophoresis tank and a constant current of 300mA was applied for 1h and 45min, whilst the tank was cooled by inclusion of an ice reservoir. The presence of SDS in the resolving gel confers a negative charge on the protein so the cassette was oriented with the nitrocellulose towards the anode.

2.3.4 Immunological Detection of Protein

Following transfer of the proteins from the gel to the nitrocellulose membrane, the membrane was removed and any remaining protein blocked by incubation in a solution of 2% (w/v) BSA in NaTT buffer (150mM NaCl, 20mM Tris (pH 7.4), 0.2% (v/v) Tween -20) for 90min with gentle agitation on a platform shaker. The blocking buffer was then removed and membranes incubated overnight with antiserum specific to the target protein appropriately diluted in NaTT buffer containing 0.2% (w/v) BSA. On the following day, membranes were washed in NaTT every 15min for 90min with gentle agitation. The membranes were then incubated for a further 90min at room temperature with secondary horseradish peroxidase-conjucated IgG directed against the first immunoglobulin diluted to approximately 1:10000 in NaTT buffer containing 0.2% (w/v) BSA. After 6 washes in NaTT for 90min as described before, immunnoreactive protein bands were detected by incubation in enhanced chemiluminescence (ECL) reagent for 2min with agitation. The membranes were then mounted onto an exposure cassette and covered with cling film, then exposed to X-ray film (B Plus – Full Blue) for the required time under darkroom conditions and developed using X-OMAT machine (KODAK M35-M X-OMAT processor).

2.3.5 Nitrocellulose membrane stripping and reprobing

Nitrocellulose membranes processed by Western Blotting were reprobed for the subsequent detection of other cellulose proteins. This involved stripping the membrane of any previous antibody using a stripping buffer (0.05M Tris-HCl, 2% SDS, and 0.1M of β -

mercaptoethanol). The membrane was incubated in 15ml of stripping buffer for 60min at 70°C in an incubator/shaker (Stuart Science Equipment). The stripping buffer was then discarded in a fume hood sink and the membrane washed 3 times with NaTT buffer at 15min intervals to remove residual stripping buffer. After the final NaTT wash, membranes were then incubated overnight with primary antibody prepared in 0.2% BSA (w/v) in NaTT buffer. At this stage, the blots were ready for the immunological detection protocol (section 2.34).

2.4 Preparation of nuclear extract

Cells were grown on 10cm plates until 90% confluency and then exposed to vehicle or agonist for indicated time period. After washing with 1ml of ice cold PBS, cells were scraped into 0.5ml PBS and transferred to Eppendorf tubes. Cells were then harvested by centrifugation at 13000 rpm for 1min. The pellet was resuspended in 400µl of buffer 1 (10mM Hepes pH 7.9, containing 10mM KCl, 0.1mM EDTA, 0.1mM EGTA, 1mM DTT, 0.5mM PMSF, 10µg/ml aprotinin, 10µg/ml leupeptin and 10µg/ml pepstatin) and incubated on ice for 15min. 25µl of 10% (w/v) NF-40 was added into the samples and vortexed at full speed for 10s. The nuclear fraction was then separated by centrifugation at 13000 rpm for 1min and supernatant were removed and pellet were resuspended in 50µl of buffer 2 (20mM Hepes, (pH 7.9), 25% (v/v) glycerol, 0.4M NaCl, 1mM EGTA, 1mM DTT, 0.5mM PMSF, 10µg/ml leupeptin and 10µg/ml pepstatin) and agitated at 4°C for 15min. Extracts were then sonicated in ice bath for 2 x 30s. Nuclear proteins were then recovered by centrifugation and the soluble nuclear extract collected and stored at -80°C until use.

2.4.1 Determination of Protein Concentration in the Nuclear Extracts

Quantification of protein concentration was determined using the Bio-Rad DC^{TM} Protein Assay Dye Reagent Concentrate Farbstoff-Konzentrat on the Bradford assay method. A standard curve was performed using various concentrations (2-20µg) of BSA. Appropriate dilutions of the standards and samples were made in sterile dH₂O (790-795µl) mixing with 200µl of dye agent and left for 15min at room temperature for the reaction. Samples were transferred into a cuvette and the colour development was quantified at 595nm by Ultrospec®2000 UV/visible spectrophotometer. The protein concentration of each sample was calculated from the standard curve.

2.5 Luciferase reporter activity assay

Adenovirus encoding luciferase gene containing NF κ B binding sites was purchased from (Vector Biolabs Inc). HUVECs were grown to confluency in 12 well plates then after infected with Adv. NF κ B luc for 40h prior stimulation with TNF α at the indicated time points. Cells were exposed to agonists for 8h, followed by washing twice with ice cold PBS and 100 μ l of lysis buffer was added on the cell. The cells were scraped from the plates and transferred to Eppendorf tubes, followed by centrifugation at 13,000 rpm for 2 min. Mixing100 μ l of the supernatant to 100 μ l of luciferase buffer (lysis buffer containing 1mM ATP, 1% (v/v) BSA and 0.2 mM luciferin substrate) in the cuvette. The relative light units were measured using a luminometer (F12 luminometer, Berthold detection system).

2.6 Immunofluorescent Microscopy

2.6.1 Transfection of yellow fluorescent protein (YFP)-tagged IKKα or YFP-tagged p65 DNA

HUVECs were seeded in 12 wells plates, and allowed to reach between 60-70% confluency. 1µg of DNA were then added into 1ml of media per well. FuGENE® HD Transfection Reagent was mixed with 2.5µl per 1µg of DNA. Cells were incubated with 1ml of media which contain transfection reagent/DNA overnight and were replaced with fresh media next day. Cells were left to recover for another 24h followed by stimulation with TNF- α (10ng/ml) for indicated time periods.

2.6.2 Immunofluorescent Microscopy

HUVECs were transfected with YFP-tagged IKK α or p65 in 12 well plates that contained glass coverslips and then stimulated as appropriate, coverslips were washed in ice cold PBS twice prior to fixation with ice cold methanol (500µl) at room temperature in dark for 10 min. After fixation, cells were washed twice with ice cold PBS and cell were incubated with 500µl of DAPI (4',6-Diamidino-2-Phenylindole, Dihydrochloride) (1:2000) for 5 min. After incubation, cells were washed with PBS twice again. Mowiol was added on the slide and the coverslips were placed on top of the slide. Slides were kept away from light and left overnight before microscopy. Cells were visualized using a Nikon TE300-E upright epifluorescence microscope and imaged at x100 and x40 magnification using an oil-

immersion Plan Fluor objective lens. Images were collected using a digital Cool Snap-HQ CCD camera (Roper Scientific, Photometrics, Tucson, AZ). MetaMorph Imaging Series 7.0 (Molecular Devices Corp., Downington, PA, USA) was used for control of image acquisition, processing and modification of all image data. The background average statistical correction editing function in MetaMorph was used to produce background corrected images. This was achieved by determining the average background level of fluorescence from regions of interest drawn adjacent to cells expressing fluorescence.

2.7 Preparation of recombinant adenoviruses

2.7.1 Crude Adenoviral Lysates

Crude lysates of dominant negative inhibitory kappa B kinase alpha (DN-IKK α) and dominant negative inhibitory kappa B kinase beta (DN-IKK β) virus were generated in HEK 293 cells, by infecting a 75cm² flask with 0.75 μ l of original stock virus. Flasks were incubated at 37°C, 5% CO₂, 95% air for 5-7 days until the cytopathic effect had been observed and the cells had started to detach from the surface of the flask. Cells were then removed from the flask and subjected to centrifugation (1500g for 5 min). The supernatant was removed and pellet was washed twice with PBS and centrifuged at 1500g for 5 min again. The collected pellet was then resuspended in 1ml of HE buffer (10mM HEPES pH 7.5, 1mM EDTA), frozen in liquid nitrogen and thawed in a 37°C water bath. This was repeated a further 2 times. After the third cycle, cells were centrifuged at 1500g for 5 min to pellet the debris and the supernatant, which constitute the crude adenoviral lysates, was collected in sterile tube and stored at -80°C until required.

2.7.2 Generation and Purification of High-Titre Stocks of Adenovirus

The high-titre stocks of adenovirus were generated by large scale amplification of the crude adenoviral lysates described in section 2.8.1. 21 T175 flasks of HEK 293 cells were grown to approximately 70-80% confluency.1 flask was used as negative control, while the 20 flasks of medium were then changed to 2% FCS medium (24ml) before receiving 1ml of medium of diluted virus (100µl crude adenoviral lysates added to 25ml of HEK 293 medium). These flasks were incubated at 37°C, 5% CO2 until the cytopathic effect had occurred and cells had detached from the flask. The cells were then collected together with medium from the flasks and pelleted by centrifugation in a 50ml centrifuge tube at 1500g for 5 min. Majority of the

supernatant was removed in a sterile centrifuge tube and stored at 4°C until use. Cells were resuspended in 25 ml of supernatant and frozen in liquid nitrogen and thawed in at 37°C water bath, this was repeated a further 2 times. Subsequently, cells were mixed by vortexing after each thaw. Following the third cycle, cells were pelleted by centrifugation at 1500g for 5min. Supernatant was collected in a sterile tube while the pellet was discarded. A BD Adeno-X virus purification kit from Clontech Laboratories was used for the adenovirus purification. The supernatant was subject to filtering and then incubation with benzonase (25 units/ μ l) (50mM Tris-HCl pH8.0, 20mM NaCl, 2 mM MgCl₂, 50% glycerol) at 37°C for exact 30 min. Meanwhile, 1 x dilution buffer and 1 x washing buffer were prepared by diluting the provided 5 x buffers with sterile H₂O, according to the manufacturing instructions. The benzonase treated filtrate was mixed with an equal volume of 1 x dilution buffer and passed through BD Adeno-X syringe-Filter followed by washing with 1 x washing buffer. Adenovirus was later eluted from the mega filter using elution buffer and the elutate collected in sterile Eppendorf tubes and aliquots then stored at -80°C until titration.

2.7.3 Titration of Adenovirus by End-Point Dilution

Titration of the end point dilution method was used to quantify virus production (Nicklin and Baker, 1999). Serial dilutions of the adenovirus were applied to a 96 well plate of HEK 293 cells that reached approximately 70-80% confluency. The plate was incubated at 37° C, 5% CO₂ overnight after which the medium containing adenovirus was replaced with fresh medium. After that the plate was incubated for 10 days at 37° C, 5% CO₂ until the cytopathic effect of the virus had reached a steady state. Wells containing plaques were counted and the titre of adenovirus stock in terms of plaque-forming unit (pfu) was calculated as in the example shown below.



Figure 2.1 the End-Point Dilution for adenovirus titration. A series of concentrations of the adenovirus were applied on each of the row. The cells were observed in Day 10. The number of cells showed cytopathic effect would be counted in each of the concentration for further calculation.

Calculation of titre value

Using the two lowest concentrations for rest of calculation which present with plaques (although conc 1must be over 50).

<u>Example.</u>

	Concentration	Plaque present wells	Percentage
Concentration 1	10 ⁻⁹	6/10	60
Concentration 2	10 ⁻¹⁰	2/10	20

Percentage of conc1 – 50

Percentage of conc1- percent of conc2



Figure 2.2 The calculation of the titre of the adenovirus. The titre of adenovirus 59 would be used to determine the amount of adenovirus for each well.

2.7.4 Infecting HUVECs with Adenovirus

To establish the appropriate volume of adenovirus to add to cells in order to give an appropriate multiplicity of infection (MOI), cells were grown to approximate 50-60% confluency in 12 well plates. The cell number $(10\mu l)$ was assessed by using a haemocytometer and the cell number in 1ml was determined by the formula:

```
Total amount of cells (10\mu l) \times 10^3 / 4
```

The amount of adenovirus applied to the cells was determined by another formula:

(Pfu) x total number of cell (1ml)/ titre value of the adenovirus

An appropriate MOI of adenovirus (50-300pfu/cell) was added to the cells and incubated for 40h in normal growth medium before stimulation. After this period, cells were exposed to agonist, TNF- α or IL-1 β for the indicated time periods.

2.8 Scanning Densitometry

Western Blots were scanned on an Epson Perfection 1640SU Scanner using Adobe photoshop 5.0.2 software. The captured images were then normalized to a control and quantified using Scion Image (Scion Corp., Maryland, USA).

2.9 Data Analysis

All data shown were expressed as mean \pm s.e.m and were representative of at least three separate experiments. The statistical significance of differences between mean values from control and treated groups were determined by either one-tailed Student's Unpaired t-test or a two tailed one-way analysis of variance (ANOVA) with Dunnett's post test (p<0.05 was considered significant).

Chapter 3

Characterisation the role of IKKα in the regulation of Canonical and Non canonical NFκB pathways

3.1 Introduction

Emerging evidence indicates that IKK α plays a critical role in the regulation of both canonical and non-canonical NF κ B pathway. Studies using knockout models and molecular manipulation supports the hypothesis that IKK α regulates gene transcription by a phosphorylation of a number of nuclear targets including histone H3 (Yamamoto et al., 2003) and CBP (Huang et al., 2007). Furthermore IKK α has been demonstrated to play a role in the regulation of p100 processing (Liang et al., 2006; Luftig et al., 2004 Muller and Siebenlist, 2003) and p65 phosphorylation (Solt et al., 2009). TNF- α induced IKK α phosphorylation has also been demonstrated in many different cell types (Discussed in chapter 1).

Despite however the evidence presented above, there are a number of issues to be considered when examining the role of IKK α . First of all is the difficulty in measuring IKK α activation within the cell. Our previous studies in the laboratory found the specificity of commercial phospho-IKK α antibodies to be problematic, results could not be obtained using Western blotting. Furthermore *in vitro* kinase activity following immunoprecipitation using specific IKK α antibodies is difficult to interpret due to co-immunoprecipitation of both IKK β and IKK γ as part of the complex. However a number of studies have shown TNF α induced nuclear translocation of IKK α in MEF and HeLa cells and this may also be a feature of endothelial cells. Therefore in the first part of the chapter, TNF α stimulated nuclear translocation of IKK α in HUVECs will be examined. Nuclear IKK α could possibly be a good marker for activation of both canonical and non-canonical NF κ B pathways.

Another feature of IKK α which makes defining its role in endothelial cell function difficult is the potential for it to function not only as a kinase but also as a scaffolding protein. Thus techniques designed to inhibit both the kinase activity of IKK and also its expression need to be considered. Therefore in this chapter, we also tested an adenoviral version of dominant negative IKK α and both siRNA and shRNA constructs, the latter again as adenovirus.

3.2 Characterisation of nuclear IKKa in HUVECs

The kinetics of TNF- α induced nuclear translocation of IKK α was examined using two approaches. Firstly nuclear extracts were generated using sub-cellular fractionation (see chapter 2.4) and secondly, cells were transfected with epitope-tagged IKK α .

3.2.1 Subcellular fractionation to determine the nuclear translocation of IKKa and p65

HUVECs were incubated with TNF- α (10ng/ml) to induce nuclear translocation of IKK α . Two methods were employed to isolate nuclei, one a crude method used in routine electrophoretic mobility shift assay (EMSA) studies within the laboratory (Goh et al., 2009, MacKenzie et al., 2007) and the second a modified approach (Solt et al., 2009). Both gave similar results, however the standard method gave more protein to use in immunoblots.

The expression of IKK α in the nuclear fraction of HUVECs at various time points following TNF- α stimulation is shown in figure 3.1. IKK α levels in the nucleus increased by 15 min and reached a maximum at 30 min of approximately 4 folds of basal values. However, translocation was transient, levels returned to basal values by 60 min. TNF- α also stimulated an expected increase in nuclear expression of p65 suggesting that the cells had been effectively stimulated. The recovery of histone H3, a nuclear marker, indicated consistency of loading. The expression of the cytoplasmic protein, β -actin, was used as a negative control to confirm the relative purity of the nuclear samples.

To confirm the results obtained with subcellular fractionation, immunofluorescence staining was performed. Preliminary studies demonstrated that staining of endogenous IKK α with commercially available antibodies gave a strong non-specific background and could not be used routinely. Thus, yellow fluorescence protein (YFP) tagged IKK α DNA was transfected into HUVECs, followed by direct microscopy. In figure 3.2, YFP-tagged IKK α was stained in green, while the nucleus was stained blue by Dapi. In the control, non-stimulated condition, YFP-IKK α was retained within the cytoplasm. In the majority of cells, YFP-IKK α transmigrated into the nucleus following 30 min of stimulation with TNF- α . Similar to results obtained by sub-cellular fractionation, nuclear IKK α gradually returned to a cytosolic location within 2 hours. YFP-tagged p65 was used as positive control, TNF- α induced nuclear translocation of p65 demonstrated that the cells had been effectively stimulated.



Figure 3.1: TNF- α induced nuclear translocation of IKK α in HUVECs using sub cellular fractionation. Cells were stimulated with 10ng/ml of TNF- α for various time periods. Nuclear lysates were prepared, separated by SDS-PAGE, and then assessed for IKK α (84kDa), p65 (65kDa), H3 histone (17kDa) and β -actin (42kDa) expression as outlined in section 2.4. The result is representative of 1 experiment.



 +	+	+	+	TNF-α
				(10ng/ml)
 15	30	60	120	Incubation
				period (min)

Figure 3.2: TNF- α induced nuclear translocation of IKK α in HUVECs assessed by immunofluorescence. Cells were transiently transfected with 1µg/ml of YFP-IKK α or YFP-p65 for 24hr prior to stimulation with TNF- α (10ng/ml) as outlined in section 2.6.1, then followed by direct immunofluorescence as described previously (section 2.6.2). The nuclei were stained in blue colour and the YFP-tagged IKK α was in green colour. In Panels A and C, cells were viewed at x100 magnification, while cells in panel B were viewed at x40 magnification, using a Nikon EP-1 fluorescent microscope. The results are representative of 3 independent experiments.

А

В

С

3.3 Comparison of knockdown efficiency for Adv. shRNA IKKa and siRNA IKKa

Previously in the laboratory, we have used adenoviral versions of both IKK α and β (Goh et al., 2009; MacKenzie et al., 2007) to assess their roles in NF κ B signalling and studies using these constructs are shown later in this chapter. However, because of limitations in their usefulness, we also sought to develop Adv. shRNA approaches to ablate endogenous IKK α as HUVECs are sensitive to the traditional transfection agents and show substantial toxicity. Figure 3.3 shows the efficiency of knockdown of endogenous IKK α using Adv. shRNA IKK α and as an alternative using standard siRNA IKK α . Panel A is a concentration curve for both shRNA IKK α and as a comparator, IKK β . HUVECs were infected with shRNA IKK α or β for 96h, although similar results were obtained following shorter periods of infection. Increasing the concentration of shRNA IKK α up to a maximal concentration (1000pfu) showed no significant difference upon endogenous IKK α expression in comparison to control. Similarly, a maximum concentration (1000pfu) of shRNA IKK β also failed to knockdown endogenous IKK β expression in HUVECs. The presence of LacZ did not affect the expression of either IKK α or IKK β .

In panel B, HUVECs were transfected with siRNA IKK α at increasing concentrations up to a maximum of 200nM. Negative control non-targeting siRNA acted as a scrambled sequence for IKK α and showed a minimal effect on endogenous IKK α expression. Surprisingly, the lowest concentration (100nM) of siRNA IKK α effectively knocked down endogenous IKK α expression, reducing levels by approximately 90%. Furthermore siRNA IKK α even at maximum concentration was without effect on IKK β levels suggesting a selective effect. Total p65 was used as a loading control and demonstrated that either non-targeting or IKK α do not cause a toxic effect in HUVECs. Overall this result confirmed that siRNA IKK α was effective and also induced a minimal cytotoxic effect to the cell. Therefore, siRNA IKK α was used for further studies.



 	500	700	1000					shRNA IKKα (pfu)
 				300	500	700	1000	shRNA IKKβ (pfu)
 +								LacZ (1000pfu)

В

 $^+$

+

+

+



Figure 3.3: The effect of Adv. shRNA IKK α and siRNA IKK α on IKK α expression in HUVECs. A) Cells were infected with increasing concentration of Adv. shRNA IKK α or shRNA IKK β for 96h and B) Cells were transfected with non-targeting scramble sequence or siRNA IKK α up to a concentration of 200nM. Whole cell lysates were prepared, separated by SDS-PAGE, and then assessed for IKK α (84kDa), IKK β (86kDa) and p65 (65kDa) as outlined in section 2.3. The results are representative of 3 independent experiments.

+

+

+

Α

Fugene $(3.7\mu L)$

3.3.1 TNF-a induced IkB-a loss in HUVECs in a time dependent manner

It has been shown previously in the laboratory that TNF- α induced a loss in cellular I κ B- α in HUVECs leading to the activation of the NF κ B pathway (discussed in Section 1.3.3). Therefore the aim of this initial experiment was to confirm the optimal time period for TNF- α induced I κ B- α loss. In figure 3.4, I κ B- α degradation was observed 5min after TNF- α (10ng/ml) stimulation and levels reached a minimum at 15min (% basal expression = 1.37%) and was sustained over 30min (% basal expression = 1.10%). Subsequently, the expression of I κ B- α gradually recovered, returning towards basal values by 90min (% basal expression = 21.5%).

3.3.2 The effect of Adv. DN-IKK β in the regulation of TNF- α induced I κ B- α degradation in HUVECs

A number of studies have shown that IKK β plays an intimal role in the regulation of the canonical NF κ B pathway (Discussed in Section 1.3.3.8.1). Figure 3.5 shows a concentration curve for the effect of Adv. DN-IKK β on TNF- α induced I κ B- α degradation in HUVECs. HUVECs were infected with increasing concentrations of Adv. DN-IKK β 40h prior to stimulation with TNF- α (10ng/ml). As expected, TNF- α induced I κ B- α degradation after 15 min (% basal I κ B- α expression = 6.21%, n = 1). Whilst this response was not altered in the presence of β -galactosidase (LacZ), the extent of I κ B- α degradation was greatly reduced (% basal I κ B- α expression = 73.7%, n = 1) following overexpression of DN-IKK β . Both DN-IKK β and the LacZ did not affect the expression of p65, this also indicated loading of each sample was equal.

3.3.3 Characterisation of the role of IKK α in the regulation of TNF- α induced I κ B- α degradation in HUVECs

The role of IKK α in the regulation of the canonical NF κ B pathway can be cell type dependent. Evidence has shown IKK α plays no role in the I κ B- α loss in HeLa cells (Hu et al., 1999), vascular smooth muscle cells (MacKenzie et al., 2007) and MEF (Solt et al., 2009), whereas IKK α does play a role in the regulation of I κ B- α loss in HT-29 cells (DiDonato et al.,1997) The following experiment was performed to determine whether IKK α shares similar characteristics with IKK β in the regulation of the canonical NF κ B pathway. Figure 3.6 shows a concentration curve for the effect of Adv. DN-IKK α on TNF- α induced I κ B- α loss in

HUVECs. HUVECs were infected with Adv. DN-IKK α 40h prior to stimulation with TNF- α (10ng/ml). As expected, TNF- α significantly induced I κ B- α degradation after 15min (% basal I κ B- α expression = 5.43%). In contrast, and unlike IKK β , increasing the concentration of Adv DN-IKK α did not reverse I κ B- α loss. Similar to previous experiments, the agonist induced response was not altered in the presence of LacZ. The expression of p65 was used as a loading control and indicated that the concentration of protein in each sample was equal.

In order to confirm the results obtained with individual treatments, HUVECs were simultaneously pre-treated with a maximum concentration of either Adv. DN-IKK α or β for 40h prior to stimulation with TNF- α (10ng/ml). Figure 3.7 demonstrated that TNF- α significantly induced I κ B- α degradation after 15min (% basal I κ B- α expression = 12.26 ± 4.57, n = 3, P<0.01). Whilst this response was not altered in the presence of DN-IKK α , the extent of I κ B- α degradation was significantly reduced (% basal I κ B- α expression = 88.9 ± 5.38, n = 3, P<0.01) following overexpression of Adv. DN-IKK β . This confirmed the selective effect of Adv. DN-IKK β within the canonical NF κ B pathway.



Figure 3.4: TNF-\alpha-mediated I\kappaB-\alpha loss in HUVECs. Cells were stimulated with TNF- α (10ng/ml) as indicated. Whole cell lysates were prepared, separated by SDS-PAGE, and then assessed for I κ B- α (37kDa) and p65 (65kDa) as outlined in Section 2.3. The result is representative of 1 experiment.



			+	+	+	+	+	TNF-α (10ng/ml)
		300			100	200	300	DN-IKKβ (pfu)
	+			+				LacZ (300pfu)



IkB- α (37kDa)

 $\leftarrow I \kappa B - \alpha (37 k D a)$ $\leftarrow I K K \alpha (84 k D a)$ $\leftarrow p 65 (65 k D a)$

			+	+	+	+	+	TNF-α (10ng/ml)
		300			100	200	300	DN-IKKa (pfu)
	+			+				LacZ (300pfu)

В





A



Figure 3.7: The effect of Adv. DN-IKKa upon TNF-a-mediated IkB-a degradation in **HUVECs.** Cells were infected with Adv. DN-IKK α , β or LacZ for 40h prior to stimulation with TNF-α (10ng/ml) for 15min. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for A) IkB-a (37kDa), IKKa (84kDa), IKKβ (86kDa) and p65 (65kDa) as outlined in section 2.3. Blots were quantified for B) % basal IkB-a expression and c) IkB-a/p65 ratio by scanning densitometry, each value represents the mean \pm S.E.M. The results are representative of 73 3 independent experiments. *P<0.05, **P<0.01 compared with agonist stimulated control.

3.3.5 The effect of Adv. DN-IKKα in the regulation of TNF-α induced p65 phosphorylation in HUVECs

Similar to the previous experiment, HUVECs were infected with either DN-IKK α or β for 40h prior to stimulation with TNF- α (10ng/ml). In figure 3.8, panel A shows that TNF- α induced a 3 to 5 fold increase in the phosphorylation of p65 after 15min of stimulation. This response was not altered in the presence of 300 pfu of Adv. DN-IKK α , in contrast, increasing concentrations of Adv. DN-IKK β (see panel B) reduced phosphorylation of p65. Maximum concentrations of LacZ did not affect p65 phosphorylation and total p65 was used as a loading control.



 	300			100	200	300	DN-IKKα (pfu)
 			+				LacZ (300pfu)
 		+	+	+	+	+	TNF-α (10ng/ml)

В



 	300			100	200	300	DN-IKKβ (pfu)
 			+				LacZ (300pfu)
 		+	+	+	+	+	TNF-α (10ng/ml)

Figure 3.8: The effect of Adv. DN-IKK α or β upon TNF- α -mediated p65 phosphorylation in HUVECs. Cells were infected with A) Adv. DN-IKK α , B) Adv. DN-IKK β or LacZ for 40 hours prior stimulation with TNF- α (10ng/ml) for 15min. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for p-p65 (65kDa), IKK α (84kDa), IKK β (86kDa) and p65 (65kDa) as outlined in section 2.3

3.3.6 Lack of effect of Adv. DN-IKKα in IL-1β activated canonical NFκB pathway in HUVECs

Similar to TNF- α , IL-1 β can also induce activation of the canonical NF κ B pathway. However the effect of IL-1 β is thought to be mediated by a different upstream mechanism, this would still lead to activation of the IKK complex, subsequent I κ B- α degradation and p65 nuclear translocation (discussed in Section 1.3.3.8). In this section, the role of IKK α in IL-1 β induced NF κ B activation was assessed and the result compared to TNF- α induced responses.

3.3.7 Characterisation of the effect of IL-1 β on the canonical NF κ B pathway in HUVECs

Figure 3.9 shows a time course of I κ B- α degradation (panel A) and phosphorylation of p65 (panel B) induced by of IL-1 β . Following stimulation of HUVECs with 10ng/ml of IL-1 β , I κ B- α levels decrease rapidly reaching a minimum value between 10 and 30 min (% basal I κ B- α expression = 17.3% and 10.6% respectively). As with TNF- α , degradation was transient with I κ B- α returning to basal values by 120 min. This response compared well with the positive control, TNF- α , which gave maximal degradation between 15 and 30min. In addition, IL-1 β induced p65 phosphorylation. This response started as early as 5min and reached maximum at 30min of stimulation. Figure 3.9 shows a concentration response curve for IL-1 β induced I κ B- α degradation and p65 phosphorylation. Both parameters increased in a concentration dependent manner between 1 and 10ng/ml at which the maximal response was obtained. However, the curve was very steep and concentration curves could not be easily quantified. Nevertheless, a concentration of 10ng/ml of IL-1 β was used for subsequent experiments.

3.3.8 The effect of Adv. DN-IKKa and β on IL-1β induced NFkB activation in HUVECs

In figure 3.10, HUVECs were infected with Adv. DN-IKK α or β for 40h prior to stimulation with IL-1 β . Using 10ng/ml of IL-1 β resulted in a marked stimulation of I κ B- α degradation after 30min (% basal I κ B- α expression = 23.7%). Overexpression of Adv. DN-IKK α up to 300pfu or the lacZ control at the same concentration failed to affect the IL-1 β induced loss in I κ B- α . Similar results were obtained for IL-1 β induced p65 phosphorylation, Adv. DN-IKK α was without effect.



 +	+	+	+	+			IL-1β (10ng/ml)
 					+	+	TNF-α (10ng/ml)
 5	10	30	60	120	15	30	Incubation period (min)

В



Figure 3.9: Time course and concentration dependency for IL-1 β mediated IkB- α degradation and p65 phosphorylation. A) Cells were stimulated with IL-1 β (10ng/ml) or TNF- α (10ng/ml) for the indicated time points. B) Cells were stimulated with increasing concentrations of IL-1 β or 10ng/ml of TNF- α for 15 and 30min respectively. Whole cell lysates were prepared, separated by SDS-PAGE, and then assessed for IkB- α (37kDa) degradation, phosphorylation of p65 (65kDa) and total p65 (65kDa) as outlined in section 2.3. The results are representative of 3 independent experiments



 		+	+	+	+	+	IL-I p (IOng/mI)
 	+			100	200	300	DN-IKKa (300pfu)
 +			+				LacZ (pfu)

Figure 3.10: The effect of Adv. DN-IKK α upon IL-1 β mediated I κ B- α degradation and p65 phosphorylation in HUVECs. Cells were infected with Adv. DN-IKK α or LacZ for 40h prior to stimulation with IL-1 β (10ng/ml) for 30 min. Whole cell lysates were prepared, separated by SDS PAGE and assessed for a) I κ B- α (37kDa), p-p65 (65kDa), IKK α (84kDa) and p65 (65kDa) as outlined in section 2.3.

3.4 The role of IKKa in the regulation of the non-canonical NFkB pathway

The processing of p100 is believed to be a critical up-stream activation step in the noncanonical NF κ B pathway in which the formation of p52 allows NF κ B (RelB/p50) to translocate into the nucleus. The literature reviewed in chapter 1 suggests that IKK α is the only IKK member to regulate p100 processing. Therefore the aims of the subsequent experiments are to investigate the effect of TNF- α on p100 processing in HUVECs and determine the role of IKK α in the regulation of this pathway.

3.4.1 Characterisation of the effect of TNF- α on the non-canonical NF κ B pathway in HUVECs

Figure 3.11 panels A and B show an extended time course over 24h, using 2 blots, for TNF- α induced p100 processing. The response was found to be very slow with a lag period of 4 to 6h before an increase in p52 formation was observed. A maximum response was obtained by 8h and remained at these levels for the remainder of the time course. Interestingly, p100 expression also increased with similar kinetics to that observed for p52 formation. Again maximal expression was achieved at 24h of TNF- α stimulation (Fold stim = 3.69)

Figure 3.12 shows a concentration curve for TNF- α induced p100 processing. TNF- α stimulated the formation of p52 over the low ng/ml concentration range with a maximal response of 3.84 fold obtained at 10ng/ml. In contrast, p100 induction seemed to be maximal by 3ng/ml of TNF- α , however this difference was not investigated further. The experiment in figure 3.13 is similar to figure 3.12 panel B, 20ng/ml of IL-1 β was used instead of TNF- α as a stimulant. A very weak response was observed after 8h stimulation with IL-1 β which was sustained for the remainder of the time course.

3.4.2 Characterisation of the effect of Adv. DN-IKKα in the non-canonical NFκB pathway in HUVECs

Initially, HUVECs were pre-treated with Adv. DN-IKK α or β for 40h prior to stimulation with 10ng/ml of TNF- α for a further 14h (Figure 3.14). As expected, TNF- α significantly increased the formation of p52 (Fold stim = 4.26 ± 0.44, n=3). However, whilst the response was not affected by overexpression of Adv. LacZ control, Adv. DN-IKK α significantly reduced p52 formation (Fold stim = 2.02 ± 0.50, n = 3, p<0.01). Furthermore, whilst p52 formation was significantly enhanced in the presence of Adv. DN-IKK β (Fold stim = 5.91 ± 0.89, n = 3 p<0.05), incubation with the non-selective IKK inhibitor, BMS (10µM), significantly reduced the TNF- α response (Fold stim = 0.65 ± 0.074, n = 3, p<0.01). This inhibitory effect corresponded with a significant reduction in the expression of p100. Indeed, when the ratio of inhibition of p100 induction and p52 formation were compared (panel C), it was found that TNF- α stimulation and both Adv. DN-IKK α or β did not affect the p100:p52 ratio.

3.4.3 Characterisation of the effect of siRNA IKKα on the non-canonical NFκB pathway in HUVECs

In order to confirm the results obtained with the Adv. DN-IKK α , the effect of siRNA IKK α was examined in figure 3.15, HUVECs were pre-treated with siRNA IKK α or as a comparator infected with Adv. DN-IKK β prior to stimulation with 10ng/ml of TNF- α for 14h. TNF- α significantly increased the formation of p52 (Fold stim = 4.26 ± 0.44, n=3, p<0.01). Although, the response was not affected by the presence of Adv. LacZ, siRNA IKK α significantly reduced TNF- α induced p100 processing (Fold = 1.62 ± 0.42, n = 3, p<0.01). In contrast, Adv. DN-IKK β enhanced the response of TNF- α induced p100 processing (Fold= 6.38 ± 0.41, n = 3, p<0.05). Again, when the ratios of p52 formation were compared to the increase in p100 expression, it was found that TNF- α stimulation and both Adv. DN-IKK α or β did not affect the p100:p52 ratio.



В

А



Figure 3.11: Time Course of TNF- α -mediated p100 processing in HUVECs. Cells were stimulated with TNF- α (10ng/ml) for A) a maximum 8h time period and B) an extended time period. Whole cell lysates were prepared, separated by SDS-PAGE, and then assessed for p100 (100kDa) and p52 (52kDa) formation and total p65 (65kDa) as outlined in section 2.3. The results are representative of 1 independent experiment.



Figure 3.12: Concentration dependency of TNF- α -mediated p100 processing in HUVECs. Cells were stimulated with increasing concentrations of TNF- α for 14h. Whole cell lysates were prepared, separated by SDS-PAGE, and then assessed for p100 (100kDa) and p52 (52kDa) formation and total p65 (65kDa) as outlined in section 2.3.


 +	+	+	+	+	+	+	+	IL-1β (20ng/ml)
 8	10	12	14	16	18	20	24	Incubation period (hours)

Figure 3.13: Time Course of IL-1 β -mediated p100 processing in HUVECs. Cells were stimulated with IL-1 β (20ng/ml) for the indicated time points. Whole cell lysates were prepared, separated by SDS-PAGE, and then assessed for p100 (100kDa), p52 (52kDa) formation and total p65 (65kDa) as outlined in section 2.3.







А



D



Figure 3.14: The effect of Adv. DN-IKK α or β upon TNF- α -mediated p100 processing in HUVECs. Cells were infected with Adv. DN-IKK α or β for 40h prior to stimulation with TNF- α (10ng/ml) for 14h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for A) p100 (100kDa), p52 (52kDa) and p65 (65kDa) as outlined in section 2.3. Blots were quantified for B) Fold stimulation of p52 and C) p100 are measured by scanning densitometry and D) The ratio of p100:p52, each value represents the mean \pm S.E.M. The results are representative of 3 independent experiments. *P<0.05, **P<0.01 compared with agonist stimulated control.

35



 			+	+	+	+	+	TNF-α (10ng/ml)
 	+					+		siRNA IKKα (100nM)
 		+					+	DN-IKKβ (300pfu)
 +					+			NT (100nM)
 				+				Laz (300pfu)

В

А



86



Figure 3.15: The effect of Adv. DN-IKK β and siRNA IKK α upon TNF- α -mediated p100 processing in HUVECs. Cells were transfected with non-targeting siRNA (100nM), siRNA IKK α (100nM) or infected with DN-IKK β (300pfu) for 40h prior stimulation with TNF- α (10ng/ml) for 14h. Whole cell lysates were prepared, separated by SDS PAGE and assessed for A) p100 (100kDa), p52 (52kDa) and p65 (65kDa) as outlined in section 2.3. Blots were quantified for B) Fold stimulation of p52 and C) p100 are measured by scanning densitometry and D) The ratio of p100:p52, each value represents the mean \pm S.E.M. The results are representative of 3 independent experiments.

3.5 The role of IKKa in the regulation of NFkB transcriptional activity

The previous section demonstrated that IKK α is not involved in the up-stream regulation of the canonical NF κ B pathway; however, there are numerous studies have demonstrated IKK α regulation of p65 dependent genes including ICAM-1, VCAM-1 and IL-8 as discussed in chapter 1. Therefore there is a possibility that IKK α plays a role in the regulation of NF κ B (p65) transcriptional activity. In the following experiments, a luciferase assay was used and cells were infected with an Adv. NF κ B-Luc gene expressing 5 repeat sequences (TGGGGACTTTCCGC) which represents a p65 dependent binding site.

3.5.1 TNF-α and IL-1β induced NFκB reporter activity in HUVECs

Figure 3.16 shows a time course of TNF- α (panel A) or IL-1 β (panel B) induced NF κ B reporter activity over 24h. HUVECs were infected with Adv. NF κ B-Luc, Adv. DN-IKK α and/or β for 40h prior to simulation with 1ng/ml of TNF- α for the indicated time points. In panel A, the onset of the TNF- α induced NF κ B reporter activity was very slow; an increase in activity was only observed after 4h stimulation with TNF- α . A maximum response was obtained by 6h to 8h (Fold stimulation = 31264 and 28989) and levels returned toward basal values after 24h stimulation. A similar pattern was observed for IL-1 β , the onset of the response was found to be very slow and an increase of NF κ B reporter activity was observed after 4h stimulation. A maximum response was obtained by 6h to 8h (Fold stimulation = 26307 and 28310) and again returned to the basal level after 24h stimulation. In addition, luciferase buffer alone did not elicit the NF κ B luciferase activity without the presence of the agonist and Adv. NF κ B-Luc. Adv. LacZ was used as negative control which had no effect on the response.

3.5.2 Characterisation of the role of IKK α in the regulation of TNF- α induced NF κ B reporter activity

In figure 3.17, HUVECs were infected with Adv. NF κ B-Luc, Adv. DN-IKK α and/or β for 40h prior to simulation with 1ng/ml of TNF- α for a further 8h. TNF- α alone dramatically increased the NF κ B reporter activity (28920 fold). Although lower concentrations of Adv. DN-IKK α did not affect the response, the maximum concentration (300pfu) of Adv. DN-IKK α significantly reduced the NF κ B reporter by 45%. As expected, 200pfu of Adv. DN-IKK β was able to reduce approximately 40% of the reporter activity (% maximum

stimulation = 57.17 ± 20.22, n=3, p<0.01) and by approximately 70% in the presence of a maximum concentration (300pfu) of Adv. DN-IKK β (% maximum stimulation = 30.14 ± 6.42, n=3, p<0.01). In order to determine the possibility of a synergistic or additive inhibitory effect, cells were infected with both Adv. DN-IKK α and β . A combination of a fixed concentrations. Adv. DN-IKK β (50pfu) with increasing concentration of Adv. DN-IKK α (100pfu-300pfu) elicited a greater inhibitory effect than single kinase inhibition alone. Furthermore all the double treatments showed significant reduction in the response. However the effects were found to be additive rather than synergistic. For example, Adv. DN-IKK β (50pfu) alone reduced about 18% of the NF κ B reporter activity and 100pfu of Adv. DN-IKK β with 100pfu of Adv. DN-IKK β with 100pfu of Adv. DN-IKK β brought down about 40% of the response. Addition of both values equal to 40%, which is almost equal to the % inhibitory of double IKK inhibition. This was also found in other concentrations of the combinations of Adv. DN-IKK α and β , therefore the double inhibitory effects rather than synergic effect.

3.5.3 Characterisation of the role of IKK α in the regulation of IL-1 β induced NF κ B reporter activity

In figure 3.17 panel B, HUVECs were infected with Adv. NFκB-Luc, Adv. DN-IKKα and/or β for 40h prior to simulation with 1ng/ml of IL-1 β for a further 8h. Similar to the result in panel A, the presence of a low concentration of Adv. DN-IKKa did not affect the response, whilst 200pfu (% maximum stimulation = 54.86 ± 8.25 , n = 3, p<0.05) and a maximum concentration of 300pfu (% maximum stimulation = 57.71 ± 1.66, n = 3, p<0.05) reduced 42% of the NFκB reporter activity. Again, lower concentrations of Adv. DN-IKKβ significantly inhibited the response, as did a maximum concentration (300 pfu) which reduced level by as much as 70% (% maximum stimulation = 31.48 ± 0.81 , n=3, p<0.01). A combination of fixed concentrations of DN-IKKB (50pfu) or (100pfu) with increasing concentration of Adv. DN-IKKa (100pfu-300pfu) elicited greater inhibitory effect than single kinase inhibition and all these treatments showed a significantly reduction in the response. For example, Adv. DN-IKKB (50pfu) alone induced only 3.6% inhibition and 100pfu of IKKα alone elicited about 8.1% inhibition. Combination of 50pfu of Adv. DN-IKKβ with 100pfu of Adv. DN-IKKa brought down about 37% of the response. In addition, 50pfu of Adv. DN-IKKβ alone reduced 3.6% and Adv. DN-IKKα (300pfu) reduced about 42.3 % of the response, whilst the combined treatment of 50pfu of Adv. DN-IKKB with 300pfu of Adv.

DN-IKK α resulted in 69% inhibition of the response, however addition of both values only equal to 46%, which is lower than the inhibitory effect of double inhibition, therefore there may be a synergic inhibitory effect on IL-1 β induced NF κ B reporter activity when blocking both IKK α and β . Moreover, combination of both Adv. DN-IKK α (100pfu)and β (100pfu) was able to induced maximum inhibitory effect, increasing the concentration of Adv. DN-IKK α to 100pfu of Adv. DN-IKK β did not show a further reduction of the response.



Figure 3.16: Time Course and concentration dependency of TNF- α or IL-1 β mediated NF κ B reporter activity in HUVECs. Cells were infected with Adv. NF κ B-Luc 40h prior to the stimulation of A) TNF- α (1ng/ml) and B) IL-1 β (1ng/ml). Cell lysates were then measured for luciferase activity as previously described in section 2.5. Data shown is expressed as NF- κ B luciferase activity (RLU/s)



Figure 3.17: The effect of Adv. DN-IKK α on NF κ B luciferase reporter activity mediated by TNF- α and IL-1 β in HUVECs. Cells were infected with Adv. NF κ B-Luc (50pfu) and additionally DN-IKK α (50-300pfu) and/or β (50-300pfu) as indicated for 40h prior to stimulation with A) TNF- α and B) IL-1 β for 8h. Cell lysates were then measured for luciferase activity as previously described in section 2.5. Data shown is expressed as % agonist stimulation and each value represents the (mean ± s.e.m, n=3). *<0.05, **P<0.01 compared with agonist-stimulated control

А

3.6 The role of IKKa in the regulation of ICAM-1 and VCAM-1 expression in HUVECs

Although the previous study in our laboratory showed IKK α was not involved in the regulation of either ICAM-1 or VCAM-1 expression in human vascular smooth muscle cells (MacKenzie et al., 2007), the luciferase studies in this chapter showed the potential for IKK α to play a partial role in the regulation of the canonical NF κ B pathway. Therefore, theoretically IKK α should be able to regulate p65 dependent genes. Indeed, there are numerous studies which have demonstrated that IKK α plays a role in the regulation of those p65 dependent genes (discussed in chapter 1). The following experiment was conducted to identify the role of IKK α in the regulation of TNF- α and IL-1 β stimulated ICAM-1 and VCAM-1 expression.

3.6.1 TNF-α induced ICAM-1 and VCAM-1 expression in HUVECs in concentration dependent manner

It has been shown previously in the laboratory that TNF- α induced ICAM-1 and VCAM-1 in both HUVECs and human vascular smooth muscle cells (Al-Mutairi et al., 2010; MacKenzie et al., 2007). Therefore the aim of this initial experiment was to confirm the optimal concentration for TNF- α induced ICAM-1 and VCAM-1 expression. In figure 3.18, HUVECs were stimulated with increasing concentrations of TNF- α for 8h. The response to TNF- α was very sensitive, the lowest concentration (0.1ng/ml) was sufficient to strongly stimulate both ICAM-1 and VCAM-1 expression. Whilst both responses reached a maximum following stimulation with 10ng/ml of TNF- α , a standard concentration of 1ng/ml was used for subsequent studies.

3.6.2 Characterisation of the role of IKKβ in the regulation of TNF-α induced ICAM-1 and VCAM-1 expression in HUVECs.

It has been shown that IKK β plays a key role in the regulation of both ICAM-1 and VCAM-1 expression (Denk et al., 2001; Mackenzie et al., 2007; Meiler et al., 2002). Figure 3.19 demonstrates a concentration curve for the effect of Adv. DN-IKK β on TNF- α induced ICAM-1 and VCAM-1 expression in HUVECs. As expected, TNF- α induced both ICAM-1 and VCAM-1 expression after 8h (Fold stim = 17.59 and 60.83). Whilst this response was not altered following infection with Adv. Lac, increasing the concentration of Adv. DN-IKK β gradually inhibited TNF- α induced ICAM-1 expression in HUVECs in a

concentration-dependent manner. A maximum concentration of DN-IKK β almost abolished both ICAM-1 and VCAM-1 expression (% inhibition = 74% and 76.3% respectively).

3.6.3 Characterisation of the role of IKKα in the regulation of TNF-α induced ICAM-1 and VCAM-1 expression in HUVECs.

The following experiment was then performed to determine whether IKK α shares similar characteristics with IKK β in the regulation of ICAM-1 and VCAM-1 expression in HUVECs. Figure 3.20 shows a concentration curve for the effect of Adv. DN-IKK α on TNF- α induced ICAM-1 and VCAM-1 expression. TNF- α induced both ICAM-1 and VCAM-1 expression after 8h, increasing concentration of Adv. DN-IKK α gradually reduced ICAM-1 expression which was approximately 50% at a maximum of 300 pfu. In contrast low concentration of DN-IKK α had little effect on VCAM-1 expression, although a maximum concentration (300pfu) gave a small but significant reduction of the VCAM-1 expression (% inhibition = 27.5%). The presence of the Adv. lacZ slightly reduced TNF- α induced ICAM-1 expression but did not alter VCAM-1 expression.

In order to confirm the results obtained with individual treatments, HUVECs were simultaneously pre-treated with a maximum concentration of Adv. DN-IKK α (300pfu) and Adv. DN-IKK β (200pfu). In figure 3.21, ICAM-1 expression was found to be significantly increased following 8h stimulation with TNF- α (Fold stim = 20.06 ± 1.91, n = 3, p<0.01), whilst Adv. DN-IKK α reduced the response by 25 % (Fold stim = 13.7 ± 2.28, n = 3, p<0.05). As expected, Adv. DN-IKK β caused more than a 75% reduction of the response (Fold stim = 2.87 ± 0.78, n = 3, p<0.01). Combined infection with both Adv. DN-IKK α and β completely abolished the expression of ICAM-1 (Fold stim = 0.78 ± 0.38, n = 3, p<0.01).

A similar result was obtained for TNF- α induced VCAM-1 expression (Figure 3.22). Inhibition of IKK α function by Adv. DN-IKK α only reduced the response by around 20% (Fold stim = 40.20 ± 4.43, n = 3, p<0.05). In contrast, the response was reduced by up to 80% following Adv. DN-IKK β infection (Fold stim = 9.86 ± 4.72, n = 3, p<0.01). The additional presence of Adv. DN-IKK α enhanced the inhibitory effect of Adv. DN-IKK β (Fold stim = 40.20 ± 4.43, n = 3, p<0.01).

In order to confirm the effect of Adv. DN-IKK α on TNF- α induced ICAM-1 and VCAM-1 expression, siRNA IKK α was used in the following experiment. In figure 3.23, HUVECs were pre-treated with either Adv. DN-IKK α , β or siRNA IKK α for 40h prior to stimulation of TNF- α . Alone 300pfu of DN-IKK α strongly inhibited TNF- α induced ICAM-1 expression, over 85% inhibition was observed. A low concentration of Adv. DN-IKK β (75pfu) reduced ICAM-1 expression by approximately 30%. Interestingly, knockdown of endogenous IKK α using siRNA proved more effective and reduced ICAM-1 expression by more than 60%. Similar to the previous experiment, combining both DN-IKK α and β generated a greater inhibitory effect than blocking either of the kinases alone (% inhibition = 88.5%), IKK α knockdown and DN-IKK β also induced a stronger inhibitory effect (% inhibitory = 80.6%). The non-targeting scrambled DNA had a slight inhibitory effect on the TNF- α induced response.



Figure 3.18: Concentration dependency of TNF- α -stimulated induction of ICAM-1 and VCAM-1 expression in HUVECs. Cells were stimulated with increasing concentration of TNF- α for 8h. Whole cell lysates were prepared, separated by SDS PAGE, and then assessed for ICAM-1 (90kDa), VCAM-1 (120kDa) and p65 (65kDa) expression as outlined in section 2.3.



Figure 3.19: The effect of Adv. DN-IKK β upon TNF- α -mediated ICAM-1 and VCAM-1 expression in HUVECs. Cells were infected with Adv. DN-IKK β or LacZ for 40h prior to stimulation with TNF- α (1ng/ml) for 8h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for ICAM-1 (90kDa), VCAM-1 (120kDa), IKK β (86kDa) and p65 (65kDa) as outlined in section 2.3.



Figure 3.20: The effect of Adv. DN-IKK α upon TNF- α -mediated ICAM-1 and VCAM-1 expression in HUVECs. Cells were infected with Adv. DN-IKK α , or LacZ for 40h prior to stimulation with TNF- α (1ng/ml) for 8h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for ICAM-1 (90kDa), VCAM-1 (120kDa), IKK β (86kDa) and p65 (65kDa) as outlined in section 2.3.





Figure 3.21 : The effect of Adv. DN-IKK α upon TNF- α -mediated ICAM-1 expression in HUVECs. Cells were infected with DN-IKK α , β or LacZ for 40h prior to stimulation with TNF- α (1ng/ml) for 8h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for A) ICAM-1 (90kDa), IKK α (84kDa), IKK β (86kDa) and p65 (65kDa) as outlined in section 2.3. Blots were quantified for B) Fold stimulation of ICAM-1 expression by scanning densitometry, each value represents the mean \pm S.E.M. The results are representative of 3 independent experiments. *P<0.05, **P<0.01 compared with agonist stimulated control.



Figure 3.22: The effect of adv. DN-IKK α upon TNF- α -mediated VCAM-1 expression in HUVECs. Cells were infected with DN-IKK α , β or LacZ for 40h prior to stimulation with TNF- α (1ng/ml) for 8h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for A) VCAM-1 (120kDa), IKK α (84kDa), IKK β (86kDa) and p65 (65kDa) as outlined in section 2.3. Blots were quantified for B) Fold stimulation of VCAM-1 expression by scanning densitometry, each value represents the mean \pm S.E.M. The results are representative of 3 independent experiments. *P<0.05, **P<0.01 compared with agonist stimulated control. 100



 		+	+	+	+	+	+	+	TNF-α (1ng/ml)
 	+			+			+		DN-IKKa (300pfu)
 +						+		+	siRNA IKKα (100nM)
 					+		+	+	DN-IKKβ (75pfu)
 	+		+						NT (100nM)

Figure 3.23: The effect of Adv. DN-IKK α and siRNA IKK α upon TNF- α - mediated both ICAM-1 and VCAM-1 expression in HUVECs. Cells were transfected with non-targeting scramble DNA, siRNA IKK α or infected with Adv. DN-IKK α or β for 40h prior to stimulation with TNF- α (1ng/ml) for 8h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for ICAM-1 (90kDa), VCAM-1 (120kDa), IKK α (84kDa), IKK β (86kDa) and p65 (65kDa) as outlined in section 2.3.

3.6.4 Characterisation of the effect of IL-1 β on both ICAM-1 and VCAM-1 expression in HUVECs

Previous experiments conducted using TNF- α were repeated using IL-1 β . Figure 3.24, panels A and B show time courses of ICAM-1 and VCAM-1 expression and concentration dependent responses induced by IL-1 β . Following stimulation of HUVECs with 10ng/ml of IL-1 β , a very slow lag time of 4 to 6h was observed before an increase in ICAM-1 expression was manifest. A maximum response was obtained between 8h and 12h of stimulation (Fold stim 25.18 and 29.58 respectively) but dropped to 13.61 fold by 24h. Similarly, VCAM-1 expression also showed a lag period of 4 to 6h before an increase in VCAM-1 expression was observed. Again maximum expression was achieved by 8h (Fold stim = 20.08), the response dropped to around 10 fold by 12h and returned back toward basal levels by 24h. In panel B, HUVECs were stimulated with increasing concentrations of IL-1 β by 8h. Both ICAM-1 and VCAM-1 expression was not induced by very low concentrations of IL-1 β (0.1-0.5ng/ml), however, significant expression was observed by 1ng/ml of the cytokine and reached a maximum level by 10ng/ml (Fold stim = 56.84 and 72.56 respectively).

3.6.5 Characterisation of the role of IKKβ on IL-1 β induced both ICAM-1 and VCAM-1 expression in HUVECs

Figure 3.25 shows a concentration curve for the effect of Adv. DN-IKK β on TNF- α induced ICAM-1 and VCAM-1 expression in HUVECs. HUVECs were infected with increasing concentrations of Adv. DN-IKK β 40h prior to stimulation. As expected, IL-1 β induced ICAM-1 expression after 8h (Fold stim = 32.66), increasing the concentration of Adv. DN-IKK β gradually inhibited IL-1 β induced ICAM-1 expression in HUVECs. A maximum concentration (200pfu) of Adv. DN-IKK β almost abolished ICAM-1 expression (Fold stim = 4.43 and % maximum inhibition = 86.4%). The effect of DN-IKK β on IL-1 β induced VCAM-1 expression was also shown to be concentration dependent. IL-1 β induced VCAM-1 expression after 8h (Fold stim = 22.79), a maximum concentration of DN-IKK β (200pfu) again nearly abolished the expression of VCAM-1 (Fold stim = 2.79, % maximum inhibition = 87.7%).

3.6.6 Characterisation of the role of IKK α on IL-1 β induced ICAM-1 and VCAM-1 expression in HUVECs

In order to confirm the results obtained with individual treatments, HUVECs were simultaneously pre-treated with a maximum concentration of Adv. DN-IKK α (300pfu) and the lower Adv. DN-IKK β (75pfu). In figure 3.26, ICAM-1 expression was found to be significantly increased following 8h stimulation with IL-1 β (Fold stim = 26.96). Whilst Adv. DN-IKK α reduced by approximately 50% the ICAM-1 response (Fold stim = 13.53), 75pfu of Adv. DN-IKK β reduced the ICAM-1 levels by 40% (Fold stim = 59.77). The double inhibition of both IKK α and β completely abolished the expression of ICAM-1 (Fold stim = 1.07). A similar result was obtained for VCAM-1 expression. A maximum concentration (300pfu) of DN-IKK α caused a 88% reduction in the expression of VCAM-1 whilst 75pfu of DN-IKK β caused a 75% inhibition. Unsurprisingly, infection of both Adv. DN-IKK α and β resulted in virtual inhibition of VCAM-1 expression. (Fold stim = 0.14).



Figure 3.24: Time course and concentration dependency of IL-1 β mediated ICAM-1 and VCAM-1 expression in HUVECs. A) Cells were stimulated with IL-1 β (1ng/ml) for the indicated time points. B) Cells were stimulated with increasing concentrations of IL-1 β for 8h. Whole cell lysates were prepared, separated by SDS-PAGE, and then assessed for ICAM-1 (90kDa), VCAM-1 (120kDa), and p65 (65kDa) as outlined in section 2.3.



 +								Laz (pfu)
 		50	75	100	125	150	200	DN-IKKβ (pfu)
 	+	+	+	+	+	+	+	IL-1 β (1ng/ml)

Figure 3.25: The effect of Adv. DN-IKK β upon IL-1 β -mediated ICAM-1 and VCAM-1 expression in HUVECs. Cells were infected with Adv. DN-IKK β or LacZ (300pfu) for 40h prior to stimulation with IL-1 β (1ng/ml) for 8h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for ICAM-1 (90kDa), VCAM-1 (120kDa), IKK β (86kDa) and p65 (65kDa) as outlined in section 2.3.



 			+	+	+	+	+	IL-1 β (1ng/ml)
 	+				+		+	DN-IKKa (300pfu)
 		+				+	+	DN-IKKβ (75pfu)
 +				+				Laz (pfu)

Figure 3.26: The effect of Adv. DN-IKK α upon IL-1 β –mediated ICAM-1 and VCAM-1 expression in HUVECs. Cells were infected with Adv. DN-IKK α , β or LacZ (300pfu) for 40h prior to stimulation with IL-1 β (1ng/ml) for 8h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for ICAM-1 (90kDa), VCAM-1 (120kDa), IKK α (84kDa), IKK β (86kDa) and p65 (65kDa) as outlined in section 2.3.

3.7 Discussion

The aim of the chapter was to characterise the role of IKK α in both the canonical and noncanonical NF κ B pathways in HUVECs. Whilst the significance of IKK β in this cell type has been explored to some extent (Bu et al., 2005, De Martin et al., 2000), the role of IKK α has been largely neglected. Two approaches were considered, firstly the activation of the kinase itself and pathways known to be activated downstream and secondly, molecular inhibition of IKK α activity by using Adv. DN-IKK α and siRNA. These strategies have revealed an important role for IKK α in HeLa, MEFs and have implicated roles in cancer and cardiovascular disease.

The first section of this chapter sought to identify the ability of IKK α to undergo nuclear translocation in response to cellular stimulation. Figure 3.1 demonstrated that TNF- α induced IKK α nuclear translocation by 30 min, but the IKK α level returned to basal within 2 hours. Very few studies have examined IKK α nuclear translocation, however in comparison, Anest and co-workers demonstrated sustained TNF- α induced IKK α nuclear translocation in MEFs which lasted for as long as 120 min (Anest et al., 2003) and Yamamoto and co-workers showed that IKK α interacts with CBP following 30min of TNF- α stimulation in MEFs (Yamamoto et al., 2003). The duration of IKK α retention inside the nucleus may vary according to different cell types. IKK α is translocated to the nucleus due to expression of NLS within the kinase domain at the N-terminus (Gloire et al., 2007, Sil et al., 2004). However other components such as NF κ B and recently IKK β have also been shown to shuffle between the nucleus and the cytosol and this possibility could apply to IKK α (Accardi et al., 2011). Secondly, the duration of stimulation could be increased to determine if the action of IKK α is slow in onset or biphasic.

It was recognised that the studies using crude nuclear extraction were limited and attempts were made to utilise a more refined procedure using high speed centrifugation. However due to the lack of specific equipment, the improved protocol could not be applied routinely in the laboratory. Therefore immunofluorescence (direct staining) was used to confirm the result obtained using crude nuclear extracts. Due to non-specificity of the commercial IKK α antibody, cells were transfected with YFP-tagged IKK α DNA, and visualised using the microscopy. Again, figure 3.2A showed TNF- α induced IKK α transient nuclear translocation in the majority of the cell population by 30min. However, the transfection efficiency was very poor with only around 20% of the whole population transfected with YFP-tagged IKKα DNA.

Due to the limitation of both techniques, the investigation of IKK α activation did not continue, however it would have been useful to determine if Adv. DN-IKK α affected nuclear translocation of IKK α . In fact, up to 700pfu of Adv. DN-IKK α did not affect nuclear translocation of IKK α (data not shown), this implies that the Adv. DN-IKK α (K44M) mutation inhibited ATP binding but not affect the affect the nuclear localisation sequence.

The second part of this chapter used molecular tools to study the role of IKK α in the regulation of NF κ B pathways. Adv. DN-IKK α , siRNA IKK α and shRNA IKK α were used in the study. Firstly, DN-IKK α and also DN-IKK β have been used routinely within the laboratory to define the role of IKK α and β in PAR-2 mediated NF κ B signalling in keratinocytes (Goh et al., 2008) and in TNF- α stimulated human vascular smooth muscle cells, (MacKenzie et al., 2007). The Adv. DN-IKK α used in this study was mutated from serine to alanine within the ATP binding site (K44M) which inhibits the kinase activity. Several other studies have utilised this approach in a number of different cell types such as neuroblastoma (Armstrong et al., 2006), mouse melanoma cells (Rangaswami et al., 2006), rat aortic smooth muscle cells (Torrie et al., 2001) and colorectal cancer cells (Fernandez-Majada et al., 2007).

The advantage of using Adv. DN-IKK α is that in HUVECs the cytotoxic effect of a normal transfection method can be avoided. However, substantial overexpression of a dominant negative protein can lead to problems in specificity of inhibition. In other words, DN-IKK α may have off-target effects. Furthermore, there is evidence to suggest that IKK α can act as a scaffold protein in the absence of kinase activity (Li et al., 2002). In MEFs deficient in IKK α , reinsertion of DN-IKK α resulted in gain of function for the expression of a number of IKK α dependent genes. Therefore knockdown IKK α expression is necessary to confirm the effect of DN-IKK α in HUVECs.

Both siRNA and shRNA have been used widely to silence relevant target genes involved in a number of signalling pathways, however, siRNA and shRNA have distinct mechanisms to mediate target RNA interference pathway. Standard siRNA requires transfection reagents such as lipofectamine or polyethylenimine to access the cytoplasm. siRNA is loaded on to

RNA-interfering silencing complex (RISC) complex then undergoes the RNA interference process via targeting mRNA cleavage and degradation. Moreover, siRNA also binds Dicer and interacts with the double-stranded Tat RNA-binding protein (TRBP) or PACT (PKR activating protein) to mediate RNA interference and miRNA processing. Unlike siRNA, Adv. shRNA however does not require a transfection reagent to get inside the cell. The viral vector transports the shRNA into the nucleus for transcription. shRNA is transcribed by either RNA polymerase II and generates the primary transcript containing a hairpin like stem-loop structure. Then it is processed in the nucleus by a complex containing the RNase III enzyme Drosha and the double-stranded RNA-binding domain protein DGCR8. The processed primary transcripts is the pre-shRNA molecule. It is then transported to the cytoplasm by exportin 5. In the cytoplasm, the pre-shRNA is loaded onto the Dicer/TRBP/PACT complex where they are further processed to mature shRNA. Mature shRNA, then associates with argonaute protein containing RISC providing RNA interference function either through mRNA cleavage and degradation (Rao et al., 2009).

Adv. shRNA IKK α was used initially as the first option to knockdown IKK α because again adenoviral infection avoids stress or damage to the endothelial cell and the infection efficiency is routinely greater than 90%. Ironically, figure 3.3 showed that the maximum concentration (1000pfu) of Adv. shRNA IKK α failed to knockdown IKK α , whilst in contrast, the lowest concentration of siRNA IKK α (100nM) was sufficient to knockdown IKK α without causing a cytotoxic effect nor affecting endogenous IKK β expression. It is generally accepted, as demonstrated by Rao and co-workers, that shRNA constructs are more suitable in utilising the endogenous processing machinery than siRNA constructs. Optimized shRNA constructs allow for a higher potency and sustainable effects resulting in less off-target effects (Rao et al., 2009). However, due to time constraints siRNA was used in subsequent experiments.

Having established what molecular tools would be suitable to inhibit IKK α activation their effects in HUVECs were examined. Firstly, the regulation of the canonical NF κ B pathway was assessed. Agonists such as TNF- α and IL-1 β are well known activators of the canonical NF κ B pathway are therefore were used as stimulants. These cytokines are well recognised to activate the IKK complex leading to cellular I κ B- α loss and p65 phosphorylation, subsequent p65 translocation into the nucleus and binding to a specific DNA sequence for transcription (discussed in chapter 1). Thus I κ B- α loss and p65 phosphorylation were utilised as the key

markers of activation of the canonical NF κ B pathway. As expected, Adv. DN-IKK β reversed TNF- α induced I κ B- α loss in HUVECs, a result consistent with a previous study in the laboratory which had demonstrated that IKK β plays a key role in the regulation of NF κ B pathway in HUVECs (Al-Mutairi et al., 2010). Actually, many studies have shown IKK β to play a crucial role in the regulation of the canonical NF κ B pathway. Tanaka and co-workers showed that TNF- α and IL-1 β failed to induced I κ B- α degradation in IKK β deficient MEFs (Tanaka et al., 1999). Another study demonstrated that expression of DN-IKK β (K44M) inhibited LPS induced NF κ B activity in human monocytes and THP-1 monocytic cells (O'Connell et al., 1998). Furthermore, overexpression of kinase inactive IKK β (K44A) in HeLa cells inhibited TNF- α induced nuclear translocation of RelA (Zandi et al., 1997). A similar response was observed in fibroblast-like synoviocytes, DN-IKK β mutant prevented TNF-alpha-mediated nuclear translocation of NF κ B (Aupperle et al., 1999).

In contrast, inhibition of IKK α by DN-IKK α did not affect the TNF- α induced I κ B- α degradation in HUVECs (figure 3.6 and figure 3.7). Likewise, IKK α played no role in I κ B- α loss in vascular smooth muscle cells (MacKenzie et al., 2007) and also human keratinocytes (Goh et al., 2008). In fact, a similar lack of inhibition was also observed in MEF from IKK α knockout mice (Solt et al., 2009). However, in contrast, Adli and co-workers demonstrated that TNF- α induced I κ B- α loss was diminished in IKK α deficient MEFs and also in siRNA IKK α pre-treated HeLa cells (Adli et al., 2010). Furthermore, inhibition of IKK α blocked the I κ B- α loss in HT-29 cells (DiDonato et al.,1997). A further study has revealed that IKK α can participate in the canonical pathway only in the absence of cellular IKK β . These and other studies would suggest a cell type-specific participation of IKK α in degradation of I κ B- α and subsequent nuclear translocation of p65.

Similarly, phosphorylation of p65 at serine 536 is also strongly associated with activation of the canonical NF κ B pathway (discussed in Section 1.3.3.8). There are numerous studies which show blocking IKK β activity resulted in inhibition of p65 phosphorylation at the serine 536 site (Goh et al., 2008; Yang et al., 2003). Furthermore Hall and co-workers demonstrated the overexpression of dominant-negative IKK α had no effect on p65 phosphorylation or TNF-alpha production, revealing that IKK β , not IKK α , plays a central role in regulation of p65 phosphorylation at serine 536 and TNF- α production in the heart (Hall et al., 2005). As expected, figure 3.8 shows that TNF- α induced p65 phosphorylation at serine 536 was not

altered in the presence of DN-IKK α , whilst infection with Adv. DN-IKK β resulted in a concentration dependent inhibition of the response.

Although TNF- α and IL-1 β can both activate the canonical NF κ B pathway through different up-stream signalling mechanism (discussed in chapter 1), Adv. DN-IKK α was also unable to affect IL-1 β induced I κ B- α loss and p65 phosphorylation. This confirms that IKK α is not involved in regulating the canonical NF κ B pathway in a cytokine dependent manner. However, Solt and co-worker found that blocking IKK α resulted in inhibition of IL-1 α induced p65 phosphorylation in MEFs. This may be due to IKK α playing a selective role in response to IL-1 α stimulation (Solt et al., 2009). Although TNF- α , IL-1 β and LPS are classified as the main agonists for activation of the canonical NF κ B pathway in osteoclast-like cells (Jimi et al., 1996). IL-1 α possesses higher affinity for IL-1 receptor type one (IL-1RI), whereas IL-1 β displays higher affinity for IL-1 receptor type two (IL-1RII) (Ricote et al., 2004). Therefore IL-1 α activated IL-1RI may result in phosphorylation and activation of IKK α which subsequently phosphorylates p65 in MEFs (Solt et al., 2009).

Although IKKa plays no role in the regulation of upstream intermediates of the canonical NF κ B pathway, a number of studies have demonstrated that IKK α can influence NF κ B transcriptional activity (Agarwal et al., 2005; Adli et al., 2010; Solt et al., 2009). However, the role of IKKa in this process in HUVECs remains uncertain. To answer this question, a NFkB luciferase assay was performed using an Adv. NFkB-Luc construct containing the consensus sequence 5'-GGGRNYYYCC-3', which has been identified as a specific p65/p50 binding site (Chen and Ghosh 1999, Chen et al 1998). As expected, inhibition of IKKB dramatically reduced TNF- α and IL-1 β induced p65 NF κ B reporter activity, a high concentration of Adv. DN-IKKB almost completely inhibited the response. Similar results have been obtained in other cell types, Yamamoto and co-workers showed that TNF-a induced NFkB luciferase activity was significantly diminished in IKKB knockout MEF in comparison with wild type (Yamamoto et al., 2003). Also, DN-IKKβ (K44A) significantly inhibited Helicobacter pylori (H.pylori) induced NFkB luciferase activity in a human gastric cancer cell line (AGS) (Hirata et al., 2006). Studies also that showed that removal of endogenous IKK β reduced TNF- α induced NF κ B luciferase activity in HeLa and breast cancer cells (Adli et al., 2010). These studies overwhelmingly support a role for IKKB

activity in mediating both NF κ B translocation and serine 536 phosphorylation of p65 as a requisite for transcriptional activation.

In contrast, high concentrations of Adv. DN-IKK α gave only a partial inhibition of TNF- α or IL-1 β induced NF κ B luciferase activity. Similar results were obtained from other studies, inhibition of IKKa kinase activity resulted in reduction in NFkB luciferase activity in RKO colorectal cancer cells. A maximum concentration of DN-IKKa caused an approximate 85% inhibition of the response (Agarwal et al., 2005). Likewise, siRNA IKKa caused a marked reduction (around 50%) in TNF-a induced NFkB luciferase activity in HeLa and breast cancer cell (Adli et al., 2010). Moreover a knockout study showed that TNF- α induced NF κ B luciferase activity was significantly decreased (around 86%) in IKKa deficient MEF in comparison to wild type and over 90% of IL-1a induced NFkB luciferase activity was reduced in IKKa knockout MEFs (Yamamoto et al., 2003). Taking these studies together. this may suggest IKKa seems to play a predominant role in the regulation of NFkB transcription activity in MEFs rather than in other cells. Inhibition of IKKa in HUVECs, HeLa and breast cancer cells resulted in around 40 to 50 % reduction of NFkB luciferase activity suggesting IKK α may not be as important in regulation of the canonical NF κ B pathway in these cells types. If time had allowed, it would have been useful to use siRNA to confirm these observations.

If IKKα can regulate NFκB reporter activity mediated by the canonical pathway, it is possible that IKKα can subsequently regulate p65 dependent genes. In this chapter, the adhesion molecules ICAM-1 and VCAM-1 were selected as p65 dependent gene. Cytokines such as TNFα and IL-1β have been demonstrated to up-regulate both adhesion molecules in endothelial cells, a process known to be independent upon the NFκB pathway (Xia et al., 1998). Previous studies from our laboratory support a predominant role for IKKβ in the regulation of both ICAM-1 and VCAM-1 expression in either vascular smooth muscle cells (MacKenzie et al., 2007) or in HUVECs (Al-Mutairi et al., 2010). Indeed, infection with Adv. DN-IKKβ ablated ICAM-1 and VCAM-1 expression in response to TNFα or IL-1β (Figure 3.19). Moreover, Denk and co-workers also showed inhibiting IKKβ by retrovirus resulted in a decrease in the expression of both adhesion molecules on the HUVECs cell surface assessed by flow cytometry analysis (Denk et al., 2001). Furthermore, Meiler and coworkers also showed Adv. DN-IKKβ supressed TNF-α induced ICAM-1 and VCAM-1 expression in HUVECs (Meiler et al., 2002). A recent study utilised 7-Hydroxyfrullanolide (7HF), a sesquiterpene lactone which was to found inhibit nuclear translocation of NF κ B by directly inhibiting the phosphorylation of IKK β . This compound leads to suppression of ICAM-1, VCAM-1 and E-selectin expression in LPS-stimulated endothelial cells (Fonseca et al., 2011). These and other approaches confirm that IKK β plays a crucial role in the regulation of the expression of adhesion molecules in HUVECs.

As the previous results showed IKK α to partially regulate NF κ B (p65) transcriptional activity, it is possible that IKKa may also be able to regulate ICAM-1 and VCAM-1 expression. Indeed, infection with a maximum concentration of Adv. DN-IKKa (300pfu) resulted in a significant reduction in TNF- α induced ICAM-1 and VCAM-1 expression. By comparison with Adv. DN-IKKB, the inhibitory effect of Adv. DN-IKKa was weaker and only caused a partial inhibition (figure 3.21 and 3.22). Interestingly, even using up to 1000pfu of DN-IKKα did not affect the TNF-α induced ICAM-1 and VCAM-1 expression in human vascular smooth muscle cells (MacKenzie et al., 2007) nor altered the enterovirus 71 (EV71) induced VCAM-1 expression in rat vascular smooth muscle (Tung et al., 2007), suggesting that the role of IKKα in the regulation of both ICAM-1 and VCAM-1 is again cell type dependent. Studies in this chapter however also confirmed a role of IKKa in the regulation of adhesion molecule expression in endothelial cell and the majority of recent studies support this finding. For example, inhibiting IKKa kinase activity using retroviral infection with DN-IKKα resulted in complete inhibition of TNF-α induced VCAM-1 and IL-8 and partially reduced ICAM-1, MCP-1 and E-selectin expression on the HUVECs cell surface (Denk et al., 2001). In addition, Gloire and co-workers showed mutation of the NLS resulted in inhibition of the TNF- α induced ICAM-1 expression (Gloire et al., 2007). Also, mutation of IKK $\alpha^{AA/AA}$ (S176A, S180A) leads to inhibition of TNF- α induced ICAM-1 expression in mouse stromal cells (Bonizzi et al., 2004). Furthermore, Huang and co-workers demonstrated IKKa plays a role in the regulation of TNF-α induced ICAM-1 expression by directly phosphorylating CBP in the nucleus of HeLa cells (Huang et al., 2007). Similarly, DN-IKKα also inhibited TNF-α inducibility of the VCAM-1 promoter (Tu et al., 2001).

In chapter 1, evidence was discussed relating to a role for IKK α in the regulation of the noncanonical NF κ B pathway. As described in this chapter, the ligands LT α 1 β 2, BAFF and RANKL are the common agonists used to activate the pathway, however recently, it has been shown that TNF- α is also associated with the activation of non-canonical NF κ B pathway (Madge et al., 2008). Therefore TNF- α was tested as an activator of the non-canonical pathway in HUVECs. In these cells, TNF- α stimulated a time dependent increase in both p100 and p52 levels. The increase in p52 levels increased concomitantly with p100 expression in response to TNF- α , suggesting that p100 may dictate the levels of p52 irrespective of whether bonafide processing is on-going. Nevertheless, processing was not observed following IL-1 treatment suggesting a difference in the mechanism of activation by the two cytokines. Infection with Adv. DN-IKK α inhibited p100 processing and significantly reducing p52 levels without affecting p100 expression, this suggests that TNF- α is mediating p100 processing in a IKK α dependent manner. Likewise, knockdown of IKK α using siRNA also caused inhibition of p100 processing. Dejardin and co-workers showed p100 processing was inhibited in both DN-IKK α transfected MEFs and IKK α deficient MEFs (Dejardin et al., 2002). Furthermore, a recent study has confirmed that inhibiting IKK α leads to inhibition of p100 processing in HUVECs (Madge et al., 2008).

Surprisingly, infection of HUVECs with Adv. DN-IKKß increased cellular expression of p52. A recent paper has demonstrated in both HUVECs and MEFs that blocking IKKB resulted in the enhancement of the p52 levels and the expression of CXCL12, a protein dependent on the non-canonical pathway (Madge and May, 2010). This implies that the canonical NF_KB pathway may negatively regulate the non-canonical pathway in HUVECs. Interestingly, blocking both IKK α and β using the non-selective IKK inhibitor BMS-345541 also led to inhibition of p100 processing but also protein expression. This again supports the notion that an increase in p52 levels may also be dependent on the relative expression of p100. To investigate this further the ratio of p52 formation vs p100 expression was examined. If TNF- α is only able to increase the expression of p100 rather than driving processing, the p100/p52 ratio would not be altered. Furthermore, following either DN-IKKa and siRNA IKKα treatment, the ratio would remain the same. However, the results were very difficult to interpret. As Adv. DN-IKKa clearly only reduced the p52 formation without affecting the p100 level, the level of p100/p52 ratio should increase, however the result did not show Adv. DN-IKKa and siRNA IKKa significantly increased the ratio of p100/p52. Both Adv. and siRNA IKKa showed an increase of the ratio but was not significantly different when compared with the agonist control. Taken together it is impossible to determine truly if p52 formation is dependency on p100 processing. Moreover TNF- α largely increased the expression of p100, and a small reduction of p52 level may not be sufficient to give a significantly increase of p100/p52 ratio.

To summarise the results presented in this chapter, $IKK\alpha$ was found not to be involved in the regulation of the canonical NFkB pathway directly, but partially regulated the p65 transcriptional activity and the expression of p65 dependent genes such as ICAM-1 and VCAM-1. These effects are consistent with earlier studies which demonstrated that following stimulation IKKa could translocate to the nucleus. A number of studies have shown that TNF- α induces IKK α nuclear translocation (Anest et al., 2003) and subsequent regulation of specific p65 dependent genes, such as IL-8, through the direct interaction with the promoter of the gene (Yamamoto et al., 2003). Moreover, Gloire and co-workers showed that whilst blocking IKKα activity did not affect the p65 translocation into the nucleus, p65 dependent gene expression however was inhibited (Gloire et al., 2007). Furthermore deletion of IKKa or mutation within the activation loop or NLS all lead to complete inhibition of TNF-α induced p65 binding to the ICAM-1 and MCP-1 promoters in MEFs (Gloire et al., 2007). In addition, Huang and co-workers found that IKKa mediated phosphorylation of the nuclear protein CBP, is crucial for p65 binding to genes promoter such as ICAM-1 and cIAP-2 in HeLa cells (Huang et al., 2007). These experiments support the findings presented in this chapter which demonstrated that IKKa plays a key role in the regulation of the non-canonical pathway in HUVECs. This implies that IKKa may regulate a number of non-canonical dependent genes such as CXCL12. Therefore inhibiting IKKα could be a potential target for novel therapeutic strategies aimed at blocking the altered endothelial cell phenotype at sites of chronic inflammation.

Chapter 4

Characterisation of the inhibitory effect of IKKα selective inhibitors (SU compounds) on Canonical and Non-Canonical NFκB pathways

4.1 Introduction

Experiments in the previous chapter established the potential for IKK α to play a role in the regulation of specific NF κ B controlled events relevant in a condition such as atherosclerosis where endothelial cell dysfunction is a feature. However, there are very few studies which have examined the potential to utilise pharmacological inhibitors in a context of atherosclerosis. A single study has shown that the non-specific NF κ B inhibitor PDTC attenuates atherosclerosis by selectively inhibiting transcription factor NF κ B (Jawien et al., 2005). Early studies demonstrated that PDTC inhibits NF κ B translocation into the nucleus (Ziegler-Heitbrock et al., 1993) and later Li and co-workers found that PDTC did not affect IKK phosphorylation of I κ B- α but interfered with the ubiquitylation process and so prevented IKK α degradation (Li et al., 2006). Apart from proteasome inhibitors, there are also specific transcription factor NF κ B inhibitors that inhibit NF κ B translocation into the nucleus without affecting I κ B- α degradation such as aromatic diamine compound NF- κ B Activation Inhibitor II, JSH-23 (sc-222061) (Shin et al., 2004).

Whilst the compounds outlined above may not be ideal for inhibiting NFkB activity and hence inflammation, nevertheless, inhibition of IKKB remains an attractive approach for diseases such as arthritis and cancer, so numerous IKKß selective and non-selective IKK inhibitors have been developed. The IKKB inhibitors widely studied include; PHA-408, PF-184 (Sommers et al., 2009), TPCA-1 (Birrell et al., 2005), PS-1145 (Castro et al., 2003) and ML120B (Schopf et al., 2006) and also non-selective IKK inhibitors such as BMS-345541 (Burke et al., 2003, MacMaster et al., 2003) and 4-Phenyl-7-azaindoles (Liddle et al., 2009). By comparison with IKK β , no commercial IKK α specific inhibitors have been developed as yet but remain an attractive approach for cardiovascular inflammatory diseases. In collaboration with the medicinal chemistry group at the University of Strathclyde, a number of novel compounds were generated which in vitro studies indicated to be IKKa selective inhibitors. Therefore the aim of this chapter was to characterise the selectivity and the potency of these novel compounds (SU) as inhibitors of IKKa function in HUVECs. The inhibition of NFkB mediated pro-inflammatory molecules such as ICAM-1 and VCAM-1 could be a useful therapeutic approach and several studies have shown that pharmacological inhibition of IKKB can prevent induction of these molecules. However, evidence clearly demonstrates that blocking p65 NFkB translocation or direct inhibition of IKKB will necessarily result in endothelial cell apoptosis which will be detrimental to vessel integrity. Therefore the ideal treatment would be to block both ICAM-1 and VCAM-1 whilst not affecting pro-survival gene expression. This could be achieved by selectively blocking IKKα.

4.2 Determining the dissociation constant (Ki value) of the SU compounds

The effects of two prototypic IKK α inhibitor compounds, SU1007 and SU1010 on IKK α/β kinase activities were examined using *in vitro* kinase assay. The kinase assay for this project was performed by Mrs. Louise C. Young using a kinase assay kit (Cell Signaling Technology, Inc., Danvers, MA, USA) with minor modifications. Both IKK α and IKK β were purified to homogeneity and then the effect of SU compounds on either IKK α or IKK β activity assessed The kinase reaction utilised a biotinylated peptide substrate containing the residues serine 32 of IkB- α . The extent of phosphorylation was assessed by dissociation-enhanced lanthanide fluorescent immunoassay (DELFIA) and the time-resolved fluorescence (TRF) for three individual experiments to determine the IC₅₀ curves. In table 5.1, the Ki value (dissociation constant) of SU1007 for IKK α was 0.182 μ M \pm 0.076 μ M, while the Ki value of IKK β was about 100 fold higher than IKK α (Ki value= 22 μ M \pm 13 μ M) Similarly, the Ki value for IKK β is about 83 fold higher than IKK α .
А

SU1007

Ki value (μM)	Prism (n=3)	n=1	n=2	n=3	Ki (±SEM)
ΙΚΚα	0.093	0.058	0.17	0.32	0.182 (0.076)
ΙΚΚβ	10	30	6	30	22 (13)

SU1010

Ki value (μM)	Prism (n=3)	n=1	n=2	n=3	Ki (±SEM)
ΙΚΚα	0.31	0.26	0.16	0.66	0.36 (0.15)
ΙΚΚβ	30	30	30	>30	30 (?)

Table 4.1: The effects of SU1007 or SU1010 on IKK α and β kinases activities. IKK α and IKK β were purified and the selectivity of SU compounds were assessed. The Ki value of SU1007 for IKK α is 0.093-0.32, whilst the Ki value for IKK β is 10-30. The Ki value for IKK α is almost 100 folds higher than IKK β . Similarly, the Ki value of SU1010 for IKK α is 0.31-0.66, whilst the Ki value for IKK β is around 30. Again the Ki value for IKK α is almost 100 folds higher than IKK β .

4.3 Characterisation of SU compounds on TNF-α or IL-1β induced activation of the NFκB canonical pathway in HUVECs

The previous chapter (figures 3.6 & 3.8) demonstrated that IKK α was not involved in the upstream regulation of canonical NF κ B pathway. Therefore the SU compounds were tested for effects within the canonical pathway to determine if these inhibitors were selective in their actions within HUVECs.

4.3.1 Characterisation of the effect of SU compounds on TNF- α induced I κ B- α degradation in HUVECs

Initially the effect of both putative selective IKK α inhibitors on two markers of the canonical pathway signalling, cellular I κ B- α loss or p65 phosphorylation was examined. Figure 4.1 shows that pre-treatment of HUVECs with SU1007 alone did not alter the basal level of I κ B- α within the cells. As demonstrated previously, TNF- α induced a significant loss in cellular I κ B- α , however, increasing the concentration of SU1007 did not result in a significant reversal. The equal cellular expression of p65 confirmed that the compound had no detrimental effect on cellular integrity. A similar result was obtained following pre-treatment with SU1010, the presence of compound did not affect either the basal level of I κ B- α nor TNF- α induced I κ B- α loss (figure 4.2).



А



Figure 4.1: The effect of SU1007 upon TNF- α -mediated I κ B- α degradation in HUVECs. Cells were pre-treated with SU1007 for 1h prior to the stimulation with TNF- α (10ng/ml) for 15min. Whole cell lysates were prepared, separated by SDS PAGE and assessed for A) I κ B- α (37kDa) and p65 (65kDa) as outlined in section 2.3. Blots were quantified for b) % expression by scanning densitometry, each value represents the mean \pm S.E.M. The results are representative of 3 independent experiments.





Figure 4.2: The effect of SU1010 upon TNF- α -mediated I κ B- α degradation in HUVECs. Cells were pre-treated with SU1010 for 1h prior to the stimulation with TNF- α (10ng/ml) for 15min. Whole cell lysates were prepared, separated by SDS PAGE and assessed for A) I κ B- α (37kDa) and p65 (65kDa) as outlined in section 2.3. Blots were quantified for b) % expression by scanning densitometry, each value represents the mean \pm S.E.M. The results are representative of 3 independent experiments. In order to ensure that using 10ng/ml of TNF- α did not result in supramaximal stimulation, thereby masking any inhibition, the concentration was reduced 10 fold to 1ng/ml for cellular stimulation. Once again, figure 4.3, panel A demonstrates that this concentration whilst being sufficient to induce IkB- α loss, this was not altered by increasing concentrations of SU1007. A similar outcome was observed with SU1010 where low concentrations did not affect the response induced by TNF- α , whilst a maximum concentration (30µM) of SU1010 slightly reversed cellular IkB- α loss.

In order to determine if the initial incubation period of 1h was insufficient to demonstrate inhibition, HUVECs were pre-treated with SU compounds for 8h, prior to stimulation with TNF- α . Figure 4.4 shows that both SU 1007 and SU1010 did not alter TNF- α stimulated I κ B- α loss even at a maximum concentration of 30 μ M. Moreover, neither IKK α nor p65 expression were affected by both compounds suggesting no detrimental effect on the cell.



 3	10	30		0.3	1	3	10	30	SU 1007 (µM)
 			+	+	+	+	+	+	TNF-α (1ng/ml)

А



Figure 4.3: The effect of SU1007 and SU1010 upon TNF- α -mediated I κ B- α degradation induced by 1ng/ml of TNF- α in HUVECs. Cells were pretreated with A) SU1007 or B) SU1010 for 1h prior to the stimulation with TNF- α (1ng/ml) for 15 min. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for I κ B- α (37kDa) and p65 (65kDa) as outlined in section 2.3. The result is representative of 1 experiment. А

В





 3	10	30		0.3	1	3	10	30	SU 1010 (µM)
 			+	+	+	+	+	+	TNF-α (10ng/ml)

Figure 4.4: The effect of prolonged pre-treatment of SU1007 and SU1010 upon TNF- α -mediated I κ B- α degradation in HUVECs. Cells were pre-treated with A) SU1007 or B) SU1010 for 8h prior to the stimulation with TNF- α (10ng/ml) for 15min. Whole cell lysates were prepared, separated by SDS PAGE and assessed for I κ B- α (37kDa), IKK α (84kDa) and p65 (65kDa) as outlined in section 2.3. The result is representative of 1 experiment.

4.3.2 Characterisation of the effect of SU compounds on TNF- α induced p65 phosphorylation in HUVECs

Another marker of activation of canonical NF κ B pathway is the phosphorylation of p65. As shown in figure 3.9, TNF- α stimulated p65 phosphorylation was not altered by Adv. DN-IKK α suggesting that SU1007 and SU1010 should not affect this parameter if selective for IKK α . Indeed figure 4.5 demonstrates that whilst TNF- α significantly enhanced p65 phosphorylation, increasing concentrations of SU1007 did not affect the response. Likewise, the response was not altered in the presence of SU1010 up to maximum concentration of 30µM (figure 4.6). In addition, neither SU compounds elicited any response in the absence of TNF- α .

p-p65 (65kDa) p65 (65kDa) 3 10 30 0.3 3 10 30 $SU\;1007\,(\mu M)$ 1 ---+ + + + $^+$ ---+ TNF-α (10ng/ml) ------



А



Figure 4.5: The effect of SU1007 upon TNF- α -mediated p65 phosphorylation in HUVECs. Cells were pre-treated with SU1007 for 1h prior to the stimulation with TNF- α (10ng/ml) for 15min. Whole cell lysates were prepared, separated by SDS PAGE and assessed for A) p-p65 (65kDa) and p65 (65kDa) as outlined in section 2.3. Blots were quantified for B) % expression by scanning densitometry, each value represents the mean \pm S.E.M. The results are representative of 3 independent experiments.



Figure 4.6: The effect of SU1010 upon TNF-α-mediated p65 phosphorylation in HUVECs. Cells were pre-treated with SU1010 for 1h prior to the stimulation with TNF-α (10ng/ml) for 15min. Whole cell lysates were prepared, separated by SDS PAGE and assessed for A) p-p65 (65kDa) and p65 (65kDa) as outlined in section 2.3. Blots were quantified for B) % expression by scanning densitometry, each value represents the mean ± S.E.M. The results are representative of 3 independent experiments.

p-p65 (65kDa)

p65 (65kDa)

4.3.3 Characterisation of the effect of SU compounds on IL-1 β induced I κ B- α or p65 phosphorylation in HUVECs

In order to exclude that the lack of effect of SU1007 and SU1010 were due to the specific use of TNF- α , the cytokine IL-1 β was examined as a comparator. In figure 4.7, HUVECs were pre-treated with SU1007 or SU1010 for 1h prior to stimulation with IL-1 β . Stimulation with 10ng/ml of IL-1 β resulted in a marked increase in I κ B- α degradation after 30min. Again, a maximum concentration of either SU compound (30 μ M) did not affect IL-1 β induced I κ B- α loss and p65 phosphorylation.



 3	10	30		1	3	10	30	SU 1007 (µM)
 			+	+	+	+	+	IL-1β (10ng/ml)

А



Figure 4.7: The effect of SU1007 and SU1010 upon IL-1 β -mediated I κ B- α degradation and p65 phosphorylation in HUVECs. Cells were pre-treated with A) SU1007 or B) SU1010 for 1h prior to the stimulation with TNF- α (10ng/ml) for 15min. Whole cell lysates were prepared, separated by SDS PAGE and assessed for I κ B- α (37kDa), p-p65 (65kDa) and p65 (65kDa) as outlined in section 2.3. The result is representative of 1 experiment.

130

4.4 Characterisation of SU compounds on TNF-α induced p100 processing in HUVECs

Figure 4.8 shows the effect of SU1007 on TNF- α induced p100 processing in HUVECs. Alone, SU1007 did not induce any increase in p100 processing. As shown in figures 3.11 and 3.12, TNF- α significantly induced p100 processing resulting in an approximate 2 fold increase in p52 formation (Fold stim = 2.08 ± 0.32, P<0.01). With increasing concentrations of SU1007, p52 formation gradually decreased with significance achieved at 10 and 30µM (SU1007 10µM: Fold increase = 1.04 ± 0.15, P<0.01 and SU1007 30µM: fold increase = 0.51 ± 0.05, P<0.01). This inhibition of p52 was accompanied by a significant reduction in the expression of p100 (TNF- α : Fold stim = 3.32 ± 0.01, SU1007 10µM: Fold stim= 2.24 ± 0.09, p<0.05 and SU1007 30µM: Fold stim = 0.95 ± 0.58, p<0.01). Examination the ratio of p100/p52 gave an inconsistent pattern with no significant change. Interestingly, pre-treatment with the non-selective IKK inhibitor BMS-345541 (30µM) also demonstrated a significant reduction of both p52 formation (Fold stim = 0.28 ± 0.002, P<0.01) and p100 (Fold stim = 0.39 ± 0.06, P<0.01).

Similar to the effect of SU1007, pre-treatment with SU1010 reduced TNF- α stimulated p52 formation, gradually decreasing the response over the low μ M range (figure 4.9). Significant inhibition was again obtained at 10 and 30 μ M. TNF- α alone significantly induced p100 processing resulting in an approximate 2 fold increase in p52 formation (Fold stim= 2.01 ± 0.11). With increasing concentrations of SU1010, p52 formation gradually and significantly decreased with significance achieved at 10 and 30 μ M (SU1010 10 μ M: Fold stim= 0.80 ± 0.34, P<0.05 and 30 μ M: Fold stim= 0.30 ± 0.25, P<0.01). Similarly, SU1010 also reduced p100 expression (TNF- α : Fold stim= 1.85 ± 0.21, SU1010 30 μ M: Fold stim= 0.30 ± 0.25, P<0.01). As with the previous experiment, pre-treatment with 10 μ M of the non-selective IKK inhibitor BMS-345541 also resulted in a significant reduction in both p52 formation (Fold stim= 0.86 ± 0.04, P<0.05) and p100 (Fold stim= 0.67 ± 0.09, p<0.01), relative to TNF- α .



	3	10	30		1	3	10	30		SU 1007(µM)
									+	BMS (10µM)
				+	+	+	+	+	+	TNF-α (10ng/ml)

A



132



Figure 4.8: The effect of SU1007 upon TNF- α -mediated p100 processing in HUVECs. Cells were pre-treated with SU1007 for 40h prior to stimulation with TNF- α (10ng/ml) for 14h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for for A) p100 (100kDa), p52 (52kDa) and p65 (65kDa) as outlined in section 2.3. Blots were quantified for B) Fold stimulation of p52 and C) p100 are measured by scanning densitometry and D) The ratio of p100:p52, each value represents the mean \pm S.E.M. The results are representative of 3 independent experiments. *P<0.05, **P<0.01 compared with agonist stimulated control.

A

-		-	-	-	-	-	-	-	-	-	p100 (100kDa)
F	-		-	-	_	-		-	tiniti Manager	-	p52 (52kDa)
-	-	-	-	-	-	-	-	-	-	-	p65 (65kDa)

 3	10	30		1	3	10	30		SU 1010(µM)
 								+	BMS (10µM)
 			+	+	+	+	+	+	TNF-α (10ng/ml)

В





Figure 4.9: The effect of SU1010 upon TNF- α -mediated p100 processing in HUVECs. Cells were pre-treated with SU1010 for 40h prior to stimulation with TNF- α (10ng/ml) for 14h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for A) p100 (100kDa), p52 (52kDa) and p65 (65kDa) as outlined in section 2.3. Blots were quantified for B) Fold stimulation of p52 and C) p100 are measured by scanning densitometry and D) The ratio of p100:p52, each value represents the mean \pm S.E.M. The results are representative of 3 independent experiments. *P<0.05, **P<0.01 compared with agonist stimulated control.

As demonstrated in the previous chapter, Adv. DN-IKK β enhanced rather than decreased TNF- α induced p100 processing (Figure 3.14 and 3.15). Having established that in figure 4.10 panel A, TNF- α induced p52 formation (3.45 fold) in the presence of Adv. DN-IKK β caused increased p52 formation (4.87 fold), whilst increasing the concentration of SU1007 with fixed concentration of Adv. DN-IKK β resulted in decreased in p52 formation (SU1007 10 μ M: Fold stim = 3.75, 30 μ M: Fold stim = 0.26). As shown in panel B, TNF- α induced p52 formation (3.5 fold). Similarly, increasing the concentration of SU1010 10 μ M: Fold stim = 0.90, 30 μ M: Fold stim = 0.32).

А

В



Figure 4.10: The effect of SU compounds on Adv. DN-IKK β pre-treated on p100 processing. Cells were pre-treated with Adv. DN-IKK β (300pfu) for 40h and/or additionally SU1007 (1-30 μ M) or SU1010 (1-30 μ M) for 1h prior to the stimulation of TNF- α for 14h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for p100 (100kDa), p52 (52kDa) and p65 (65kDa) as outlined in section 2.3. The result is representative of 1 independent experiment.

In preliminary experiments, two other IKK selective inhibitors, SU1028 and SU1053, were examined for effects on p100 processing. As shown in figure 4.11, TNF- α induced p100 processing and formation of p52. Concentrations of 3 and 10µM of SU1028 caused an approximate 30% and 36% inhibition of the TNF- α control response, respectively. A dramatic reduction (83.2%) in the response was elicited by a maximum concentration (30µM) of the compound. A similar response was observed for SU1053 with 3 and 10µM of SU1053 causing an approximate 16% and 36% inhibition of the response respectively, whilst the maximum concentration of SU1053 (30µM) caused inhibition of approximately 70%. For both compounds, p100 expression was similarly affected.

To further confirm that p100 processing was IKK α dependent and that SU compounds had an effect on IKK α , the effect of a selective IKK β inhibitor (SU937) on p100 processing was investigated. In figure 4.12, TNF- α was shown to induce p100 processing and the effect was not altered in the presence of DMSO. However, increasing concentrations of SU937 showed no effect on TNF- α induced p100 processing. As a comparison, expression of DN-IKK α reduced p52 formation whilst DN-IKK β slightly enhanced the processing.



5	10	50		1	5	10	50		50 10 2 0 (µ111)
 -								+	BMS (10μM)
 			+	+	+	+	+	+	TNF-α (10ng/ml)



Figure 4.11: The effect of SU1028 and SU1053 upon TNF- α -mediated p100 processing in HUVECs. Cells were pre-treated with either SU1028 or SU1053 for 40h prior to stimulation with TNF- α (10ng/ml) for 14h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for A) p100 (100kDa), p52 (52kDa) and p65 (65kDa) as outlined in section 2.3. The result is representative of 1 independent experiment.

A



 $^+$

 $^+$

--

--

Figure 4.12: The effect of SU937 upon TNF- α -mediated p100 processing in HUVECs. Cells were pre-treated with SU937 for 40h prior to stimulation with TNF- α (10ng/ml) for 14h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for A) p100 (100kDa), p52 (52kDa) and p65 (65kDa) as outlined in section 2.3. The result is representative of 1 independent experiment.

DN-IKKa(300pfu)

DN-IKKB (300pfu)

4.5 Characterisation of the effect of SU compounds in the regulation of TNF-α induced NFκB reporter gene activity in HUVECs

The previous chapter (figure 3.17) showed that Adv.DN-IKK α partially inhibited either TNF- α or IL-1 β stimulated p65 transcription activity in HUVECs. Therefore the SU compounds might be able to also inhibit the transcriptional activity and that was examined as shown in figure 4.13. As predicted, increasing concentrations of SU1007 gradually decreased NF κ B reporter activity. Although the effect of SU1007 on NF κ B transcription activity shows a concentration dependent manner, 3 and 5 μ M of SU1007 gave a similar inhibitory effect whilst 10 μ M caused further inhibition and a maximum concentration (30 μ M) abolished the response. The IC₅₀ value for SU1007 was 8.04 μ M and the IC₅₀ value for SU1010 is 2.31 μ M.



Figure 4.13: The effect of SU1007 and SU1010 upon TNF- α -mediated NF κ B luciferase activity in HUVECs. Cells were pre-treated with SU1007 or SU1010 for 40h prior to stimulation with TNF- α (1ng/ml) for 8h. Cell lysates were then measured for luciferase activity as previously described in section 2.6. Data shown is expressed as % agonist stimulation. The result is representative of 1 experiment.

4.6 Characterisation of SU1007 and SU1010 on TNF-α induced ICAM-1 and VCAM-1 expression in HUVECs

The previous chapter demonstrated that IKK α plays a role in the regulation of p65-dependent ICAM-1 and VCAM-1 expression. Blocking IKK α by Adv. DN-IKK α or siRNA IKK α leads to a partial reduction of both TNF- α or IL-1 β induced ICAM-1 and VCAM-1 in HUVECs (figure 3.21, 3.22 and 3.23). The SU compounds were tested to determine if they could cause a similar inhibitory effect. Figure 4.14 demonstrates the effect of SU1007 on TNF- α induced ICAM-1 expression in HUVECs. TNF- α significantly induced ICAM-1 expression (Fold increase= 58.3 ± 4.72, P<0.01). The response was not altered by the lowest concentration of SU1007 (1 μ M), but thereafter the inhibitory effect was concentration dependent. As little as 3 μ M of SU1007 reduced the expression of ICAM-1 by about 40% (Fold increase= 34.5 ± 13.5, P<0.05), while 10 μ M reduced the response by approximately 82% (Fold increase= 10.4 ± 5.5, P<0.01). A maximum concentration (30 μ M) of SU1007 virtually abolished ICAM-1 expression. Similar to the combined adenoviral mediated inhibition of both IKK α and β (figure 3.21, 3.22 and 3.23), the non-selective IKK inhibitor BMS-345541 also abolished the ICAM-1 expression (Fold increase= 3.04 ± 0.18, P<0.01).

Likewise, SU1010 also caused a concentration-dependent inhibition of TNF- α stimulated ICAM-1 expression in HUVECs (Figure 4.15). Again, the presence of the lowest concentration (1µM) of the compound did not affect the response, however, 3µM gave around a 39% reduction (Fold increase= 31.9 ± 0.52, P<0.05) and over 80% at 10µM (Fold increase= 6.96 ± 0.34, P<0.01). A maximum concentration (30µM) of SU1010 virtually abolished ICAM-1 expression.



 3	10	30		1	3	10	30		SU 1007 (µM)
 								+	BMS (10µM)
 			+	+	+	+	+	+	TNF-α (1ng/ml)



Figure 4.14: The effect of SU1007 upon TNF- α -mediated ICAM-1 expression in HUVECs. Cells were pre-treated with SU1007 for 40h prior to stimulation with TNF- α (1ng/ml) for 8h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for A) ICAM-1 (90kDa) and p65 (65kDa) as outlined in section 2.3. Blots were quantified for B) Fold stimulation of ICAM-1 expression by scanning densitometry, each value represents the mean \pm S.E.M. The results are representative of 3 independent experiments. *P<0.05, **P<0.01 compared with agonist control.





 3	10	30		1	3	10	30		SU 1010(µM)
 								+	BMS (10µM)
 			+	+	+	+	+	+	TNF-α (1ng/ml)



Figure 4.15: The effect of SU1010 upon TNF- α -mediated ICAM-1 expression in HUVECs. Cells were pre-treated with SU1010 for 40h prior to stimulation with TNF- α (1ng/ml) for 8h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for A) ICAM-1 (90kDa) and p65 (65kDa) as outlined in section 2.3. Blots were quantified for B) Fold stimulation of ICAM-1 expression and C) ICAM-1/p65 ratio by scanning densitometry, each value represents the mean \pm S.E.M. The results are representative of 3 independent experiments. *P<0.05, **P<0.01 compared with agonist control.

Similar to ICAM-1, both SU compounds showed inhibitory effects on TNF- α induced VCAM-1 expression. As shown in figure 4.16, TNF- α significantly stimulated VCAM-1 expression over 8h (Fold increase= 28.6 ± 4.60, P<0.01). As little as 3µM of SU1007 reduced the expression of VCAM-1 by approximately 40% (Fold increase = 16.5 ± 3.80, P<0.05), while 10µM of SU1007 elicited up to 90% of the inhibitory effect (Fold increase = 2.89 ± 0.47, P<0.01) and 30µM of SU1007 completely abolished the expression of VCAM-1 (Fold increase = 1.31 ± 0.43, P<0.01). As expected, inhibition of both IKK α and IKK β by BMS-345541 resulted in virtual abolition of the response (Fold increase = 1.52 ± 0.14, P<0.01).

The effect of SU1010 on VCAM-1 expression is illustrated in figure 4.17. TNF- α (1ng/ml) induced VCAM-1 expression significantly (Fold increase = 16.0 ± 4.1, P<0.01). Pretreatment with SU1010 caused a concentration dependent reduction in VCAM-1 expression which was significant with a concentration as low as 3µM (Fold increase= 9.42 ± 0.57, P<0.05), and gave as much as 90% inhibition with 10µM (Fold increase= 1.42 ± 0.54, P<0.01) of SU1010. A maximum concentration of SU1010 (30µM) completely abolished the expression of VCAM-1 (Fold increase= 0.87 ± 0.08, P<0.01). As expected, inhibition of both IKK α and IKK β by BMS-345541 (30µM) resulted in almost abolition of the response (Fold increase= 1.52 ± 0.14, P<0.01).



 3	10	30		1	3	10	30		SU 1007 (µM)
 								+	BMS (10μM)
 			+	+	+	+	+	+	TNF-α (1ng/ml)



Figure 4.16: The effect of SU1007 upon TNF- α -mediated VCAM-1 expression in HUVECs. Cells were pre-treated with SU1007 for 40h prior to stimulation with TNF- α (1ng/ml) for 8h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for A) VCAM-1 (110kDa) and p65 (65kDa) as outlined in section 2.3. Blots were quantified for B) Fold stimulation of VCAM-1 expression by scanning densitometry, each value represents the mean \pm S.E.M. The results are representative of 3 independent experiments. *P<0.05, **P<0.01 compared with agonist control.

A



Figure 4.17: The effect of SU1010 upon TNF- α -mediated VCAM-1 expression in HUVECs. Cells were pre-treated with SU1010 for 40h prior to stimulation with TNF- α (1ng/ml) for 8h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for A) VCAM-1 (110kDa) and p65 (65kDa) as outlined in section 2.3. Blots were quantified for B) Fold stimulation of VCAM-1 expression and C) VCAM-1/p65 ratio by scanning densitometry, each value represents the mean ± S.E.M. The 148 results are representative of 3 independent experiments. *P<0.05, **P<0.01 compared with agonist control.

In preliminary experiments, the effect of SU1028 and SU1053 on adhesion molecule expression was examined in HUVECs (figure 4.18). TNF- α induced a significant increase in ICAM-1 over a 8h period with a fold stim approximate 23 to 26. Both SU1028 and SU1053 also caused a significant inhibition as demonstrated in figure 4.18. However whilst 3µM of SU1028 caused an approximate 50% reduction (Fold stim = 14.1) of the response and 10µM was sufficient to abolish ICAM-1 expression. SU1053 showed a weaker inhibitory effect on the response with 3µM causing only 26% reduction in ICAM-1 expression (Fold stim = 17.4), whilst 10µM elicited a 70% reduction of the response (Fold stim = 7.29). Like other SU compounds, a concentration of 30µM effectively abolished the expression of ICAM-1.

Likewise, both SU1028 and SU1053 also inhibited the TNF- α induced VCAM-1 expression. VCAM-1 expression markedly increased following TNF- α stimulation (Fold increase = 20-33). SU1028 (3 μ M) caused 37% drop in the response (Fold increase = 21.1) and 10 μ M almost abolished the expression. Again, SU1053 also showed a weaker inhibitory effect on VCAM-1 expression. SU1053 (3 μ M) caused around 15% reduction (Fold increase = 17.3) and10 μ M elicited only 50% drop of the response (Fold increase = 10.23), whilst 30 μ M again completely inhibited the VCAM-1 expression.

ICAM-1 (90kDa) VCAM-1 (110kDa) p65 (65kDa) 3 10 30 1 3 10 30 SU 1028 (µM) ------------------------------+ BMS (10µM) --------------------+ + + $^+$ $^+$ + TNF-α (1ng/ml)

В



Figure 4.18: The effect of SU1007 and SU1010 upon TNF- α -mediated ICAM-1 and VCAM-1 expression in HUVECs. Cells were pre-treated with A) SU1028 or B) SU1053 for 40h prior to stimulation with TNF- α (1ng/ml) for 8h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for ICAM-1 (90kDa), VCAM-1 (110kDa) and p65 (65kDa) as outlined in section 2.3. The result is representative of 1 experiment.

4.7 Characterisation of SU compounds in combination with Adv. DN-IKKβ on TNF-α induced ICAM-1 expression on HUVECs

The previous chapter demonstrated that a combination of Adv. DN-IKK α and β induced a greater inhibitory effect on either ICAM-1 and VCAM-1 expression than inhibition of IKK α or β alone (figure 3.21, 3.22). As the SU compounds have been shown to inhibit both ICAM-1 and VCAM-1 expression which may be mediated by an IKKa dependent effect, they may also be able to enhance the inhibitory effect of Adv. DN-IKKB. As figure 4.19 shows pretreatment with SU1007 inhibited TNF- α induced ICAM-1 expression in the expected concentration dependent manner. Whilst 1µM of SU1007 was ineffective, as little as 3µM SU1007 caused around 50% reduction in the response. Adv. DN-IKKβ (75pfu) alone induced around 40% reduction of the response, whilst a combination of 1µM SU1007 with Adv. DN-IKKβ however did not elicit any stronger inhibitory effect than Adv. DN-IKKβ alone. In panel B, SU1010 was examined. Adv. DN-IKKβ (75pfu) alone induced an approximate 60% reduction in TNF-a induced ICAM-1, whilst again 1µM of SU1010 did not enhance the inhibitory effect of Adv. DN-IKKβ. These results together suggest that dual treatments with SU compounds with Adv. DN-IKKß did not cause synergistic inhibitory effect, rather than a combination of 3μM of the SU compounds with Adv. DN-IKKβ seemed to cause an additive inhibitory effect.

4.8 Characterisation of SU compounds on IL-1β induced ICAM-1 and VCAM-1 in HUVECs

As demonstrated in the previous chapter, expression of Adv. DN-IKK α caused a partial inhibition of IL-1 β stimulated ICAM-1 and VCAM-1 expression. Therefore it is possible that a similar inhibitory effect might be observed following pre-treatment with SU compounds. As expected, figure 4.20 panel A shows that SU1007 inhibited both IL-1 β induced ICAM-1 and VCAM-1 expression in concentration dependent manner. As with TNF- α , reduction was observed over the low micromolar range (1 to 30 μ M) with significant and considerable inhibition of both adhesion molecules at 10 and 30 μ M. A similar result was obtained following pre-treatment with SU1010 (Panel B), although overall this compound gave slightly stronger inhibition of either ICAM-1 or VCAM-1.



+

+

+

+

+

TNF- α (1ng/ml)

TNF- α (1ng/ml)

В

+

+

+

+

+

+ + +

+

+

		anger.	•		-				← ICAM-1 (90kDa)
			-lai		No.		100	-	← p65 (65k)	Da)
 	3	10	30		1	3	10	30	SU 1010(µM)	
 				+	+	+	+	+	DN-IKKβ (75pfu)	
		3 	3 10 	3 10 30 	3 10 30 +	3 10 30 1 + +	3 10 30 1 3 + + + +	3 10 30 1 3 10 + + + + +	3 10 30 1 3 10 30 + + + + + +	 ICAM-1 (p65 (65k) p65 (65k) p65 (75pfu)

Figure 4.19: The compound effects of SU1007 and SU1010 with Adv. DN-IKK β overexpression upon TNF- α -mediated ICAM-1 expression in HUVECs. Cells were pre-treated with A) SU1007 or B) SU1010 and Adv. DN-IKK β for 40h prior to stimulation with TNF- α (1ng/ml) for 8h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for ICAM-1 (90kDa) and p65 (65kDa) as outlined in section 2.3. The result is representative of 1 experiment.

+ +

152

ICAM-1 (90kDa) VCAM-1 (110kDa) p65 (65kDa) 3 10 30 1 3 10 30 SU 1007 (µM) ---------+ BMS (10µM) ----------------------------+ $^+$ $^+$ + + IL-1 β (1ng/ml) ------------ $^+$ ICAM-1 (90kDa)



 $^+$

IL-1 β (1ng/ml)

Figure 4.20: The effect of SU1007 and SU1010 upon IL-1β-mediated ICAM-1 and VCAM-1 expression in HUVECs. Cells were pre-treated with A) SU1007 or B) SU1010 1h prior to stimulation with IL-1 β (1ng/ml) for 8h. Whole cell lysates were prepared, separated by SDS-PAGE and assessed for ICAM-1 (90kDa) and VCAM-1 (110kDa) and p65 (65kDa) as outlined in section 2.3. The result is representative of 1 experiment.

 $^+$

А

В

4.9 Discussion

As discussed in chapter 1, NF κ B is strongly associated with the development of inflammatory diseases including atherosclerosis, rheumatoid arthritis and also cancers. In addition, de Martin and co-workers found that NF κ B regulates a series of anti-apoptotic proteins in various cell types including endothelial cells (de Martin et al., 1999, Erl et al., 1999, Stehlik et al., 1998a). Inhibition of NF κ B has become a tempting strategy for the therapeutic treatment of these and other diseases. Indeed, the proteosome inhibitor bortezomib has been utilised for the chemotherapeutic treatment of leukaemia and myeloma (Picot et al., 2011, Zheng et al., 2011). A most well-known strategy to inhibit NF κ B activity is the use of a proteasome inhibitor to inhibit I κ B- α degradation such as MG-132 (Shirley et al., 2005, Snyder et al., 2002) and PS-341 (Gatto et al., 2003, Sunwoo et al., 2001). In addition, the novel NF κ B inhibitor, dehydroxymethylepoxyquinomicin (DHMEQ), is thought to act by inhibiting nuclear translocation of NF κ B p65 or DNA binding of NF κ B downstream of phosphorylation and degradation of I κ B (Ariga et al., 2002, Tatetsu et al., 2005) as a results promotes the induction of apoptosis in chronic lymphocytic leukemia (Horie et al., 2006)

Due to the success of these approaches and the identification of the IKK complex, the development of IKK inhibitors has become the next logical therapeutic target and more specific than those outlined above. The strategy of kinase inhibition has become more commonplace and it is facilitated by new technologies which is allowed rapid advances in compound screening. A considerable number of IKK β inhibitors have been developed by this process including PS-1145 (Castro et al., 2003) and ML120B (Catley et al., 2006). By comparison with IKK β , there are only few compounds present which have a selective inhibitory effect on IKK α such as noraristeromycin which was shown to strongly inhibit the kinase activity of IKK α but not IKK β in HEK293 (Asamitsu et al., 2008) and a peptidyl – glucosamine derivative which inhibited IKK α nuclear translocation (Scotto d'Abusco et al., 2010). Today, there are no commercial IKK α inhibitors in the market.

A series of novel IKK α inhibitors have been developed by the Medicinal Chemistry group in the University of Strathclyde exemplified by prototypic compounds SU1007 and SU1010. These compounds contain an azole ring which specifically binds to the ATP binding domain of the kinase. This is a common mode of action for many kinase inhibitors (i.e.small
molecule inhibitors). Initially, an *in vitro* kinase assay successfully demonstrated both SU1007 and SU1010 have higher affinity for purified IKKα protein rather for IKKβ.

The initial objective was to characterise whether the compounds remain active in a cellular environment. As IKKa plays no role in the canonical NFkB cascade (shown in previous chapter), selective IKKa inhibitors should not influence in TNF-a induced IkB-a loss and p65 phosphorylation. As expected, neither SU1007 nor SU1010 affected these responses, strongly supporting the notion that the compounds do not inhibit IKKβ. However clearly there are a number of issues that need to be addressed in relation to cellular access and drug metabolism which were assessed indirectly. To ensure the compound had sufficient time to access the kinases inside the cell, the pre-treatment period of both compounds was prolonged to 8h. Again, neither SU1007 nor SU1010 had any effect on TNF-a induced IkB-a loss and p65 phosphorylation. In addition, to prevent a strong agonist response masking any inhibitory effect of the compounds, the TNF-α concentration was reduced 10 fold to 1ng/ml for cellular stimulation and again, the response was not altered in the presence of SU1007 and SU1010. Whilst taken together, these data suggest the potential of IKK α selectivity, other approaches could have been considered including in vitro kinase assay following immunoprecipitation. However, this procedure is unable to separate the components of the IKK complex and any activity would have been representative of both kinases. In addition, NFkB/DNA binding may also have been used to confirm selectivity. Studies looking at other putative IKKa inhibitors have not demonstrated pathway selectivity (Asamitsu et al., 2008; Scotto d'Abusco et al., 2010) and this may be crucial in eventual clinical use.

As mentioned previously, IKK α plays a role in the regulation of p65 NF κ B transcription activity in many cell types, the data in chapter 3 showed that Adv. DN-IKK α partially inhibited either TNF- α or IL-1 β induced p65 NF κ B transcription activity in HUVECs. As expected, both SU1007 and SU1010 caused a reduction of TNF- α induced p65 NF κ B transcription activity with high concentrations (10-30 μ M) resulting in complete inhibition of the response. These finding are in contrast with the effect of DN-IKK α which had only a partial effect and may suggest that the compounds have an additional action on IKK β . This is exemplified by SU1007 which gave a biphasic effect. This may reflect the 100 fold selectivity of the compound for IKK α over β . The previous chapter demonstrated the potential of synergy between DN-IKK α and β in causing inhibition of NF κ B transcriptional activity. Therefore a lower concentration of the SU compounds which acts to inhibit IKK α selectively might produce a similar additive effect. In addition, preliminary data (not shown) did reveal that whilst SU1007 and SU1010 did not alter TNF- α induced p65 nuclear translocation, both compounds partially inhibited p65/DNA binding as determined by EMSA (SU1007 3µM: % inhibition = 81%; SU1007 10µM: % inhibition = 64%, SU1010 10µM: % inhibition = 6%; BMS 10µM = 64%). This data supports the possibility of IKK β dependent actions of the compounds.

As discussed in chapter 1, the non-canonical NF κ B pathway is IKK α dependent and the previous chapter also showed that Adv. DN-IKKa or siRNA IKKa inhibited p100 processing, whilst Adv. DN-IKKB did the opposite. In this chapter, all the selective IKKa inhibitors SU1007, SU1010 and also in preliminary experiments, SU1028 and SU1053 demonstrated concentration-dependent inhibitory effects on TNF- α induced p100 processing. In contrast, the selective IKK β inhibitor SU937 did not alter the response. These data suggest that the compounds have some selectivity for IKKa. Recent studies identifying two selective IKKa inhibitors did not test whether processing was inhibited, however geldanamycin (GA) also showed inhibitory effect on BAFF induced p100 processing (Qing et al., 2007). However, there are caveats to these conclusions. As well as reducing p52 formation, the compounds had an inhibitory effect on p100 expression, a feature not observed for GA, therefore high concentrations of SU compounds may have small off-target effects which may mask the effects on p52 formation. Nevertheless, the inhibitory effect of SU1007 or SU1010 overwhelmed the enhancement of p100 processing by Adv. DN-IKKβ indicating the SU compounds are effective in the inhibition of the non-canonical NFkB pathway. Interestingly, inhibition of IKKB by Adv. DN-IKKB enhanced p100 processing, but SU937 did not give a similar response. In conclusion, although SU compounds may block IKKa, the inhibitory mechanisms involved for Adv. DN-IKKa and SU compounds may not be completely the same.

The SU compounds were also tested on downstream markers ICAM-1 and VCAM-1 which were shown in chapter 3 to be regulated by both IKK α and β . Both compounds at moderate concentrations caused complete inhibition of ICAM-1 and VCAM-1 expression in HUVECs in response to either TNF- α or IL-1 β . However these markers were more sensitive to inhibition by the compounds suggesting the potential of IKK α selectivity. It is also possible that inhibition at high concentrations may reflect inhibition of IKK β which could not be detected by assessing both I κ B- α loss and p65 phosphorylation. Similar to Adv. DN-IKK α , the combination of Adv. DN-IKK β with SU1007 or SU1010 also generated an additive inhibitory effect on TNF- α induced ICAM-1 expression. In addition, other modified SU compounds including SU1028 and SU1053 also caused inhibition on TNF- α induced ICAM-1 and VCAM-1 expression. Interestingly, SU1053 showed a weaker inhibitory effect than other SU compounds, thus implying SU1053 may be less selective for IKK β .

The results using SU1007 and SU1010 demonstrated these inhibitory concentrations compared well when examining the effects of other IKK inhibitors on inflammatory protein production. There are many studies showing that NFkB or IKK inhibitors suppress agonist induced ICAM-1 and VCAM-1 expression. For instance, a study showed that cyclopentenone prostaglandins inhibited IKK and NFkB activity and hence the ICAM-1 and VCAM-1 expression in endothelial cells (Zernecke et al., 2003). In addition, the non-selective IKK inhibitor BMS-345541 also inhibited TNF-α induced ICAM-1 and VCAM-1 expression in endothelial cells (MacMaster et al., 2003) which confirms the results in this chapter. Similar results can be observed in other cell types, an IKK inhibitor,2-amino-6-[2-(cyclopropylmethoxy)-6-hydroxyphenyl]-4-piperidin-4-yl-nicotinonitrile (ACHP) also showed inhibitory effects on gene expression of these two adhesion molecules in T-cell leukaemia cells (Sanda et al., 2006). Interestingly, an early study also demonstrated that nonselective IKK inhibitors (BAY11-7082 and BAY11-7085) abrogated TNF-α induced ICAM-1 and VCAM-1 expression in HUVECs (Pierce et al., 1997). The BAY-compounds (5µM) reduced approximately 50% of the adhesion molecule expression. The study also showed that BAY11-7082 and BAY11-7085 reduced swelling in a dose-dependent manner in both the rat carrageenan paw edema assay and in a rat adjuvant arthritis model (Pierce et al., 1997). However, BAY11-7082 may not be the ideal treatment for inflammatory diseases as studies have shown that BAY11-7082 induces apoptosis (Garcia et al., 2005, Mori et al., 2002). Since there is no evidence suggesting that inhibition of IKKa results in apoptosis, it is possible that the SU compounds could have desirable therapeutic effects without causing cell death. This aspect was not examined in the thesis due to time constraints.

Despite the findings in this chapter, it remains difficult to conclusively describe the SU compounds as bonafide selective IKK α inhibitors and if time had allowed a number of other approaches would have been undertaken. This includes analysis of the effect of the compounds on the translocation of IKK α to the nucleus shown to be disrupted by the peptidyl –glucosamine derivative (Scotto d'Abusco et al., 2010). In addition, CHIP could have been

employed to determine if the SU compounds could inhibit IKK α binding to IKK α promoters to reduce gene transcription. Furthermore, other downstream markers of IKK α activity include cellular events associated with angiogenesis (discussed in chapter 1). In preliminary experiments, SU1007 and SU1010 were found to be able to inhibit the migration of HUVECs in a scratch wound healing assay model. Whilst other inhibitory actions of the SU compounds could be involved, there are increasing number of studies which suggest that IKK α plays a role in the regulation of angiogenetic or metastatic gene expression such as IL-8 and MMPs and VEGF (Liekens et al. 2010; Agarwal et al., 2005). Therefore SU compounds may inhibit these angiogenic mediators and hence cell migration through inhibition of IKK α . The SU compounds may be useful tools to reveal the role of IKK α in angiogenesis and may be potentially useful anti-cancer agents.

In conclusion, the SU compounds are stable at 37°C and they were active inside the HUVECs up to 14h. Analysis of a number of intermediates, suggest they do not conclusively have an inhibitory action on the non-canonical NF κ B pathway mediated via IKK α . The SU compounds are nevertheless effective inhibitors of inflammatory mediator expression, therefore they may have potential to reduce inflammations in variety of diseases. In the future, SU compounds could be used in mice models of arthritis to determine if the compounds can reduce swelling of the joint and immunological markers such TH1 cell activation and release of cytokines. A future plan may also focus on cellular invasion to investigate whether MMPs are inhibited by SU compounds. Whether or not these compounds could be used in the context of cardiovascular disease remains uncertain.

CHAPTER 5

GENERAL DISCUSSION

5.1 General Discussion

Based on statistical data from the World Health Organization, an estimated 17.3 million people died from cardiovascular disease (CVD) in 2008 including; heart disease, stroke and atherosclerosis, representing 30% of global deaths. Unfortunately, cardiovascular disease is still the number one killer in the world today. Vascular conditions including multifocal atherosclerosis are key risk factors in the development of heart disease (Barbarash et al., 2011). Atherosclerosis is an inflammatory disorder in which a plaque is formed within the artery wall due to the accumulation of low-density lipoprotein (LDL) within macrophages, the subsequent formation of foam cells and migration of smooth muscle cells into the subendothelial layer (discussed in chapter 1). Foam cells themselves can further release proinflammatory cytokines to activate the NFkB pathway to increase the expression of adhesion molecule expression to facilitate the migration of macrophages. Oxidised LDL can also release chemokines (i.e. MCP-1) and growth factors such as PDGF resulting in proliferation of smooth muscle cells and transmigration into the sub-endothelial layer at the later stage (Libby et al., 2006). Thus, inhibition of the formation of oxidised LDL has become an important therapeutic target for the treatment of atherosclerosis disease. Statins are effective inhibitors of LDL accumulation and work by blocking the HMG CoA reductase enzyme which in turn limits the synthesis of cholesterol by 35-40% (Capurso, 2001, Libby, 2001). Statins are one of the most successful medicines in the treatment of both primary and secondary cardiac episodes and well tolerated. Nevertheless, statins can be pleiotropic. Although statins reduce smooth muscle proliferation which can slow down the progression of atherosclerosis (Bellosta et al., 2004), such an action also affects the synthesis of interstitial collagen which is responsible for the integrity of the plaque's fibrous cap and may leads to rupture and fatal thrombosis (Davies et al., 1993, van der Wal et al., 1994). Therefore despite the obvious benefit of statins and similar drugs, there is still a need for new and improved therapies.

Since atherosclerosis is essentially an inflammatory disease, one pathway which may be a good target for the development of such drugs is Nuclear Factor Kappa B. Studies have shown that non-selective NF κ B inhibitors such as PDTC can effectively reduce plaque sizes in the aortic lumen of mice (Jawien et al., 2005). Indeed, inhibition of the canonical NF κ B pathway is recognised to have a number of outcomes which would make new drug design attractive. This includes a decrease in pro-inflammatory cytokine production including TNF-

α, IL-1, IL-6 and IL-8, a reduction of inflammatory mediator expression, for example COX-2, adhesion molecules ICAM-1 and VCAM-1 or chemoattractant (i.e. MCP-1) and finally reversal of smooth muscle hyper-proliferation. However, this type of inhibition would be problematic for the endothelium resulting in a severe reduction in the expression of antiapoptotic proteins expression and an increase in endothelial cell apoptosis. This could subsequently promote instability of the plaque and superficial erosion of the endothelium which would result in coronary thrombus formation (Libby et al., 2006). Thus IKK β or classical NF κ B (p65) inhibitors may not be ideal for the treatment of atherosclerosis.

Prior to the beginning of the project, a number of studies showed IKK α to be involved in the regulation of inflammatory mediator production including IL-8, MCP-1 and the expression of adhesion molecules such as ICAM-1 and VCAM-1 (Denk et al., 2001; Huang et al., 2007; Gloire et al., 2007; Agarwal et al., 2005). Therefore inhibition of IKK α may be a better strategy for the treatment of atherosclerosis if endothelial cells could be targeted effectively. However, there has been very few studies which have examined the function of IKK α in endothelial cells. Studies completed early on in this project demonstrated that Adv DN-IKK α did not show any inhibitory effect on TNF- α /IL-1 β induced I κ B- α loss nor p65 phosphorylation, suggesting a lack of effect upon the canonical pathway and the potential to have specific roles in endothelial cell function that could be targeted. Therefore the main objectives in this thesis were to firstly determine if IKK α could be activated in endothelial cells. Secondly, to investigate the role of IKK α in the regulation of the non-canonical pathway and thirdly to determine whether this pathway could be inhibited pharmacologically.

As discussed in chapter 1, IKK α contains a nuclear localisation sequence (NLS) within kinase domain, which allows IKK α to shuffle between cytoplasm and nucleus. Early studies demonstrated that TNF- α stimulated the nuclear translocation of IKK α allowing phosphorylation of H₃ histone at ser 10 and phosphorylation of CREB binding protein (CBP) (Anest et al., 2003; Yamamoto et al., 2003). Phosphorylation at serine 10 subsequent allows acetylation of histone H3 at lysine 14 by the IKK-associated histone acetyltransferase CREB binding protein (CBP), a crucial step in modulating chromatin accessibility at NF κ Bresponsive promoters (Gloire et al., 2007, Huang et al., 2007, Yang et al., 2008). In addition, phosphorylation of CBP by IKK α allows a switch in the binding of CBP from p53 to p65, therefore IKK α can regulate the transcription of several p65 dependent genes (Huang et al., 2007). Whilst in this study, the nuclear translocation of IKK α was demonstrated, it was only possible using fluorescent IKK α and translocation was weak and transient. To date no studies have convincingly demonstrated IKK α nuclear translocation in endothelial cell, therefore it is unclear if this is a feature of its activation. Time did not allow further investigating of the nuclear role of IKK α in HUVECs, but several approaches could have been considered. For example immunoprecipitation (IP) experiments could be employed to demonstrate whether IKK α is associated with CBP. Furthermore, the technique of chromatin immunoprecipitation (CHIP) could have been performed to reveal whether IKK α binds to the promoters of known p65 dependent genes such as ICAM-1 and VCAM-1 as demonstrated in other cell types. Given that inhibition of IKK α reduced both p65 NF κ B reporter activity and expression of ICAM-1 and VCAM-1 in HUVECs, then the assumption would be that IKK α has a nuclear component to its range of functions in endothelial cells.

In addition to any proposed nuclear function for IKK α , substantial evidence has accumulated supporting a role for this kinase in the non-canonical pathway, which is presumed to occur within the cytosol. Two distinct pathways leading to the selective activation of RelA:p50 and RelB:p52 dimers, dependent on IKK β or IKK α have been identified (Ghosh and Karin, 2002). Several studies have shown that only the IKK α homodimer plays a role in the regulation of the non-canonical NF κ B pathway. Activation of receptors such as B cell-activating factor receptor (BAFFR), lymphotoxin β or receptor activator of NF κ B (RANK), subsequently trigger p100 processing, which converts p100 to p52 and allows NF κ B (Rel B/p52) to translocate into the nucleus. Indeed in this present study, it has been demonstrated that p100 processing in HUVECs is largely dependent on IKK α , results which have been confirmed by recent studies (Madge and May, 2010, Xiao et al., 2004). One of these studies suggest a negative role for IKK β in p100 processing and the results in chapter 3 support this hypothesis. This suggests the potential for multiple points of regulation by either kinase in the context of p100 processing.

Further to this, p100 processing via IKK α can be regulated further up-stream by other intermediates. NIK is a MAP3 kinase that phosphorylates IKK α leading to p100 processing and p52 formation (Ling et al., 1998). Moreover, NIK is also responsible for ubiquitination of p100 to facilitate the processing itself (Xiao et al., 2001). Inhibition of NIK results in a reduction in NF κ B reporter activity (Torrie et al., 2001). The function of NIK in endothelial cells has not been examined to date, however, in preliminary experiments, over-expression of Adv. wild type NIK resulted in an increase in p100 processing and the formation of p52 in

HUVEC, in the absence of any change in p100 protein expression (data not shown). This suggests that whilst NIK may regulate processing via IKK α , TNF- α utilising additional mechanisms. Further experiments could utilise Adv. DN-NIK to determine if p100 processing in response to TNF- α can be inhibited. Furthermore, the SU compounds which were shown to have effects on both p100 processing and protein expression could be more easily examined for inhibition of NIK induced p100 processing.

Other approaches to examining the role of IKK α in the regulation of p100 processing could involve analysis of p100 phosphorylation. Phosphorylation occurs at both serine 866 and 870 which is crucial for p100 processing (Liang et al., 2006). Given that the SU compounds had effects on both p100 processing and p100 protein expression, it would have been useful to determine whether phosphorylation was also inhibited. However, any attempts to assess p100 phosphorylation using Western blotting was unsuccessful due to interference by a number of non-specific bands making identity of the protein impossible. Other studies have been able to demonstrate p100 phosphorylation in response to different agonists such as BAFF (Claudio et al., 2002, Kayagaki et al., 2002), TNF- α (Madge et al., 2008), RANKL (Maruyama et al., 2010) and LT α 1 β 2 (Madge et al., 2008, Mordmuller et al., 2003). In recent preliminary experiments, over-expression wild type NIK has been shown to increase p100 phosphorylation, suggesting that IKK α activated by NIK can perform this function in endothelial cells (data not shown).

Subsequent to p100 processing, Rel B/p52 translocates into the nucleus to regulate the expression of a subset of NF κ B dependent genes. In some recent experiments, not included this thesis, nuclear extracts were used in Western blotting and probed for Rel B, however once again the Rel B antibody gave multiple non-specific bands. However, this and other approaches including immunofluorence, EMSA and luciferase assay would be useful for future studies to determine the impact of IKK α on the regulation of the non-canonical pathway in HUVECs. Bren and co-workers have developed selective Rel B luciferase gene constructs (1.1kb Rel B-luc and 0.25kb Rel B-luc) (Bren et al., 2001) and experiments using these constructs could determine whether DN-IKK α or SU compounds could inhibit Rel B reporter activity. CXCL12 has been identified as a Rel B/IKK α dependent gene, therefore SU compounds or DN-IKK α could inhibit not only CXCL12 gene expression but also binding of p52 to the CXCL12 promoter, assessed by CHIP. This approach could be applied to a series of IKK α /Rel B dependent genes including Blc, Sdf-1, Elc, Slc, Rxra, Irf3 and Baff (Bonizzi

et al., 2004). These genes are found mostly in lymphoid nodes which are responsible for regulation of the immune system, whether those genes are expressed in endothelial cell remains unknown.

As discussed in chapter 1, the non-canonical NFkB pathway is closely linked to angiogenesis. Angiogenesis is a process involved in blood vessel development and branching, however it is also associated with diseases such as cancers and rheumatoid arthritis. In addition, deletion of the IKKa gene in endothelial cell results in failure of microtubule formation (DeBusk et al., 2008). Moreover, IKK α also plays an important role in the regulation of angiogenic or metastatic gene expression including IL-8, VEGF and MMP-9 which are involved in colorectal cancer cell (Agarwal et al., 2005). Interestingly, in preliminary results, the SU compounds slowed down scratch induced wound healing presumably through the inhibition of endothelial cells migration (data not shown), which may indicate that IKKa can play a key role in this process. As cell migration is closely related to angiogenesis, future studies could examine whether molecular or pharmacological inhibition of IKKa could regulate microtubule formation assessed using an angiogenesis assay. In addition, it is unclear whether IKKα controls the expression of matrix metalloproteinases necessary for angiogenesis. Running a matrigel invasion assay would indicate whether IKKa can influence the degradation of the intracellular matrix. Furthermore, gene array could be employed to identify angiogenesis related genes regulated by IKKa.

As mentioned in chapter 1, IKK α and IKK β share over 50% sequence homology within NH₂terminal kinase domains. The problem in the development of IKK α inhibitors is that selective small molecules can also bind to the IKK β kinase domain which limit their selectivity of the compounds. In addition, whilst the crystal structure of IKK β has been revealed and widely used to generate IKK β selective drugs, the equivalent structure of IKK α has not been resolved and details of the structural topography remain unknown. Although the SU compounds showed good inhibitory selectivity for IKK α over IKK β *in vitro* and some selective inhibition of IKK function in HUVEC, higher concentrations of these compounds have potential to inhibit IKK β , so they could not be considered for clinical use at this time. More selective and potent IKK α are required which need to be assessed in a large series of both *in vitro* and *in vivo* assays. Nevertheless, these compounds remain some of the most promising IKK α selective drugs available. An alternative target within the non-canonical NF κ B pathway is NIK. Indeed, development of a NIK inhibitor is a new approach in drug development. A series of NIK inhibitors have been recently developed and tested by Mortier and co-workers (Mortier et al., 2010), however, the IC₅₀ values of the lead inhibitors, compounds 5 and 6, were found to be high at 51µM and 90µM respectively. These values indicate a very low potency and the potential of binding to other MAP3Ks such as MEKK1 or RAF, off-target effects on other pathways and issues such as cell cytotoxicity.

Another pharmacological approach may involve the disruption of protein/protein interactions. This strategy relies upon identifying defined domains of proteins that are crucial for the regulation and/or deactivation of target proteins. The site of interaction between IKK α/β and NEMO, is a hydrophobic pocket and contains a central conserved hexapeptide sequence, LDWSWL which is also termed the NBD (mentioned in section 1.3.3.7). This domain was initially predicted by means of hydropathy plots (May et al., 2000) but was more recently detailed in an IKK α/β peptide truncated NEMO co-crystal structure (Rushe et al., 2008). Development of cell-permeable peptides has become a promising approach for inhibition of NF κ B. Pre-treatment with NBD peptide caused inhibition of the canonical NF κ B pathway and leading to decrease in the production of IL-6, IL-12 and TNF- α and reduction of the LPS-induced maturation of dendritic cells and T cell proliferation (Tas et al., 2005). Similarly, using NBD peptide in human melanoma cultures resulted in inhibition in the NF-kB/DNA binding and also induced apoptosis via the activation of caspase 3 (Ianaro et al., 2009). This peptide may in future, provide the template for the development of small molecule inhibitors. A similar approach has yet to be tried for IKK α .

Assessment of the IKK α inhibitors implied the potential of two modes of action. Firstly inhibition of IKK α mediated processing of p100 and secondly the inhibition of IKK α mediated transcriptional activation. For the second of these actions, it is unclear if the SU compounds inhibited IKK α within the cytosol or prevented activity within the nucleus. No studies to date have confirmed that IKK α activation is required for nuclear translocation and it is interesting to note that the SU compounds are ATP binding site inhibitors. IKK α contains an NLS within the kinase domain, it is possible that some of the SU compounds may block translocation. If time had allowed studies could have been conducted to determine if the SU compounds were able to block translocation of YFP-tagged IKK α to the nucleus.

Nevertheless, the studies in this thesis demonstrate that by inhibiting IKK α , the SU compounds block the expression of inflammatory mediators including ICAM-1 and VCAM-1. More recent evidence has shown IKK α can also regulate other inflammatory genes such as IL-6, MCP-1 (Yang et al., 2008), IL-8, ICAM-1, CXCL3 (Hirata et al., 2006) and if time allowed these markers could have been examined following pre-treatment with SU compounds. Ideally, microarray studies in HUVECs could be used to undercover all IKK α dependent inflammatory genes sensitive to SU compounds including those outlined above, but also others for example, COX-2. Other arrays assessing angiogenic related genes could also be employed to gain a better overall picture of the function of IKK α in endothelial cell.

Pharmacological or molecular inhibition of the ICAM and VCAM-1 expression has been shown to attenuate the development of atherosclerosis (Fruebis et al., 1999, Pasceri et al., 2000, Patel and Kent, 1998, Zhang and Frei, 2002). IKK inhibition also reduces the release of other inflammatory mediators, thus preventing hyperproliferation of smooth muscle cells and transmigration of macrophages into the subendothelial layer. The results in this thesis suggest the potential of these effects being mediated at least partially by IKK α . Therefore drugs based on the current IKK α inhibitors designed for this project could be used to enhance the effect of statins in combination therapy without a deleterious affect on endothelial cell survival. Thus the SU compounds may have therapeutic potential in the treatment of atherosclerosis.

Nevertheless a causal link between IKK α and cardiovascular disease has not been established. Clinical data is lacking which demonstrates high levels of IKK α in atherosclerotic plaques nor additionally, a correlation between IKK α and serum markers of disease progression. Selenium is a dietary component that detoxifies lipid peroxides and has been shown to inhibit atherosclerosis more effectively than vitamin E alone (Schwenke and Behr, 1998). Zhang and co-workers have also demonstrated that selenium has inhibitory effects on ICAM-1, VCAM-1 and E-selectin expression without interfering with NF κ B (p65) nuclear translocation (Zhang et al., 2002). It would be interesting to determine if selenium had any inhibitory effect on IKK α activation or transcriptional activity. A similar question could be asked of other dietary components which are protective and other pharmacological agents such as the statins. It is clear that many major studies are required to elucidate a role of IKK α in the genesis of cardiovascular disease. In conclusion, it has been a great challenge to study the function of IKK α in HUVECs and other cell types. Recent studies suggest that the role of IKK α is a complex one and the assumption that IKK α and β may have distinctive and non-interactive roles has been continuously questioned. For example, number of recent studies show inhibition of IKK^β to result in enhanced p100 processing in HUVECs and osteoclasts (Madge and May, 2010, Ruocco et al., 2005). In addition, IKKa may also have anti-inflammatory effects in other cell types. Lawrence and co-workers have demonstrated that IKKa can inhibit p65 dependent gene expression including MIP-1 β , MIP-2, TNF- α and ICAM-1 in macrophages (Lawrence et al., 2005). This effect has been shown mediated by a negative feedback effect on the canonical NFkB pathway. For instance, IKKa phosphorylates the protein inhibitor of activated STAT1 (PIAS-1), which inhibits p65 activity (Liu et al., 2007). Furthermore, a recent study showed that IKKa directly phosphorylates the regulatory molecule TAX1BP1, which allows the complex containing TAX1BP1, Ltch and RNF11 to interact with A20 to form the A20 ubiquitin-editing complex which subsequently inhibits p65 activity (Shembade et al., 2011). These and other types of study could be performed in HUVECs to allow better understanding of the role of this important kinase in endothelial cell function in both health and disease.

CHAPTER 6

REFERENCES

References

- ACCARDI, R., SCALISE, M., GHEIT, T., HUSSAIN, I., YUE, J., CARREIRA, C., COLLINO, A., INDIVERI, C., GISSMANN, L., SYLLA, B. S. & TOMMASINO, M. 2011. IkappaB kinase beta promotes cell survival by antagonizing p53 functions through DeltaNp73alpha phosphorylation and stabilization. *Mol Cell Biol*, 31, 2210-26.
- ADLI, M., MERKHOFER, E., COGSWELL, P. & BALDWIN, A. S. 2010. IKKalpha and IKKbeta each function to regulate NF-kappaB activation in the TNF-induced/canonical pathway. *PLoS One*, 5, e9428.
- AGARWAL, A., DAS, K., LERNER, N., SATHE, S., CICEK, M., CASEY, G. & SIZEMORE, N. 2005. The AKT/I kappa B kinase pathway promotes angiogenic/metastatic gene expression in colorectal cancer by activating nuclear factor-kappa B and beta-catenin. *Oncogene*, 24, 1021-31.
- AL-MUTAIRI, M., AL-HARTHI, S., CADALBERT, L. & PLEVIN, R. 2010. Overexpression of mitogen-activated protein kinase phosphatase-2 enhances adhesion molecule expression and protects against apoptosis in human endothelial cells. *Br J Pharmacol*, 161, 782-98.
- AMIR, R. E., HAECKER, H., KARIN, M. & CIECHANOVER, A. 2004. Mechanism of processing of the NF-kappa B2 p100 precursor: identification of the specific polyubiquitin chain-anchoring lysine residue and analysis of the role of NEDD8modification on the SCF(beta-TrCP) ubiquitin ligase. *Oncogene*, 23, 2540-7.
- ANDERSON, D. M., MARASKOVSKY, E., BILLINGSLEY, W. L., DOUGALL, W. C., TOMETSKO, M. E., ROUX, E. R., TEEPE, M. C., DUBOSE, R. F., COSMAN, D. & GALIBERT, L. 1997. A homologue of the TNF receptor and its ligand enhance Tcell growth and dendritic-cell function. *Nature*, 390, 175-9.
- ANEST, V., HANSON, J. L., COGSWELL, P. C., STEINBRECHER, K. A., STRAHL, B.D. & BALDWIN, A. S. 2003. A nucleosomal function for IkappaB kinase-alpha in NF-kappaB-dependent gene expression. *Nature*, 423, 659-63.
- AOKI, H., KANG, P. M., HAMPE, J., YOSHIMURA, K., NOMA, T., MATSUZAKI, M. & IZUMO, S. 2002. Direct activation of mitochondrial apoptosis machinery by c-Jun Nterminal kinase in adult cardiac myocytes. *J Biol Chem*, 277, 10244-50.

- ARIGA, A., NAMEKAWA, J., MATSUMOTO, N., INOUE, J. & UMEZAWA, K. 2002. Inhibition of tumor necrosis factor-alpha -induced nuclear translocation and activation of NF-kappa B by dehydroxymethylepoxyquinomicin. *J Biol Chem*, 277, 24625-30.
- ARMSTRONG, M. B., BIAN, X., LIU, Y., SUBRAMANIAN, C., RATANAPROEKSA, A.
 B., SHAO, F., YU, V. C., KWOK, R. P., OPIPARI, A. W. & CASTLE, V. P. 2006.
 Signaling from p53 to NF-kappa B determines the chemotherapy responsiveness of neuroblastoma. *Neoplasia*, 8, 964-74.
- ASAMITSU, K., YAMAGUCHI, T., NAKATA, K., HIBI, Y., VICTORIANO, A. F., IMAI, K., ONOZAKI, K., KITADE, Y. & OKAMOTO, T. 2008. Inhibition of human immunodeficiency virus type 1 replication by blocking IkappaB kinase with noraristeromycin. *J Biochem*, 144, 581-9.
- AUGE, N., MAUPAS-SCHWALM, F., ELBAZ, M., THIERS, J. C., WAYSBORT, A., ITOHARA, S., KRELL, H. W., SALVAYRE, R. & NEGRE-SALVAYRE, A. 2004. Role for matrix metalloproteinase-2 in oxidized low-density lipoprotein-induced activation of the sphingomyelin/ceramide pathway and smooth muscle cell proliferation. *Circulation*, 110, 571-8.
- AUGUSTIN, H. G., KOZIAN, D. H. & JOHNSON, R. C. 1994. Differentiation of endothelial cells: analysis of the constitutive and activated endothelial cell phenotypes. *Bioessays*, 16, 901-6.
- AUPPERLE, K. R., BENNETT, B. L., BOYLE, D. L., TAK, P. P., MANNING, A. M. & FIRESTEIN, G. S. 1999. NF-kappa B regulation by I kappa B kinase in primary fibroblast-like synoviocytes. *J Immunol*, 163, 427-33.
- BAEUERLE, P. A. & BALTIMORE, D. 1988. I kappa B: a specific inhibitor of the NF-kappa B transcription factor. *Science*, 242, 540-6.
- BAEUERLE, P. A. & BALTIMORE, D. 1989. A 65-kappaD subunit of active NF-kappaB is required for inhibition of NF-kappaB by I kappaB. *Genes Dev*, 3, 1689-98.
- BARBARASH, O. L., ZYKOV, M. V., KASHTALAP, V. V. & BARBARASH, L. S. 2011. [Prevalence and clinical significance of multifocal atherosclerosis in patients with ischemic heart disease]. *Kardiologiia*, 51, 66-71.
- BARISIC, S., STROZYK, E., PETERS, N., WALCZAK, H. & KULMS, D. 2008. Identification of PP2A as a crucial regulator of the NF-kappaB feedback loop: its inhibition by UVB turns NF-kappaB into a pro-apoptotic factor. *Cell Death Differ*, 15, 1681-90.

- BECKMANN, J. S., YE, Y. Z., ANDERSON, P. G., CHEN, J., ACCAVITTI, M. A., TARPEY, M. M. & WHITE, C. R. 1994. Extensive nitration of protein tyrosines in human atherosclerosis detected by immunohistochemistry. *Biol Chem Hoppe Seyler*, 375, 81-8.
- BEG, A. A., RUBEN, S. M., SCHEINMAN, R. I., HASKILL, S., ROSEN, C. A. & BALDWIN, A. S., JR. 1992. I kappa B interacts with the nuclear localization sequences of the subunits of NF-kappa B: a mechanism for cytoplasmic retention. *Genes Dev*, 6, 1899-913.
- BELLOSTA, S., ARNABOLDI, L., GEROSA, L., CANAVESI, M., PARENTE, R., BAETTA, R., PAOLETTI, R. & CORSINI, A. 2004. Statins effect on smooth muscle cell proliferation. *Semin Vasc Med*, 4, 347-56.
- BIRRELL, M. A., HARDAKER, E., WONG, S., MCCLUSKIE, K., CATLEY, M., DE ALBA, J., NEWTON, R., HAJ-YAHIA, S., PUN, K. T., WATTS, C. J., SHAW, R. J., SAVAGE, T. J. & BELVISI, M. G. 2005. Ikappa-B kinase-2 inhibitor blocks inflammation in human airway smooth muscle and a rat model of asthma. *Am J Respir Crit Care Med*, 172, 962-71.
- BISCHOFF, J. R. & PLOWMAN, G. D. 1999. The Aurora/Ipl1p kinase family: regulators of chromosome segregation and cytokinesis. *Trends Cell Biol*, 9, 454-9.
- BISHOP, G. A. & XIE, P. 2007. Multiple roles of TRAF3 signaling in lymphocyte function. *Immunol Res*, 39, 22-32.
- BOMBELI, T., KARSAN, A., TAIT, J. F. & HARLAN, J. M. 1997. Apoptotic vascular endothelial cells become procoagulant. *Blood*, 89, 2429-42.
- BONIZZI, G., BEBIEN, M., OTERO, D. C., JOHNSON-VROOM, K. E., CAO, Y., VU, D., JEGGA, A. G., ARONOW, B. J., GHOSH, G., RICKERT, R. C. & KARIN, M. 2004. Activation of IKKalpha target genes depends on recognition of specific kappaB binding sites by RelB:p52 dimers. *EMBO J*, 23, 4202-10.
- BOURDILLON, M. C., POSTON, R. N., COVACHO, C., CHIGNIER, E., BRICCA, G. & MCGREGOR, J. L. 2000. ICAM-1 deficiency reduces atherosclerotic lesions in double-knockout mice (ApoE(-/-)/ICAM-1(-/-)) fed a fat or a chow diet. *Arterioscler Thromb Vasc Biol*, 20, 2630-5.
- BRAND, K., EISELE, T., KREUSEL, U., PAGE, M., PAGE, S., HAAS, M., GERLING, A., KALTSCHMIDT, C., NEUMANN, F. J., MACKMAN, N., BAEURELE, P. A., WALLI, A. K. & NEUMEIER, D. 1997a. Dysregulation of monocytic nuclear factor-

kappa B by oxidized low-density lipoprotein. Arterioscler Thromb Vasc Biol, 17, 1901-9.

- BRAND, K., PAGE, S., ROGLER, G., BARTSCH, A., BRANDL, R., KNUECHEL, R., PAGE, M., KALTSCHMIDT, C., BAEUERLE, P. A. & NEUMEIER, D. 1996. Activated transcription factor nuclear factor-kappa B is present in the atherosclerotic lesion. J Clin Invest, 97, 1715-22.
- BRAND, K., PAGE, S., WALLI, A. K., NEUMEIER, D. & BAEUERLE, P. A. 1997b. Role of nuclear factor-kappa B in atherogenesis. *Exp Physiol*, 82, 297-304.
- BREN, G. D., SOLAN, N. J., MIYOSHI, H., PENNINGTON, K. N., POBST, L. J. & PAYA,
 C. V. 2001. Transcription of the RelB gene is regulated by NF-kappaB. *Oncogene*, 20, 7722-33.
- BRITANOVA, L. V., MAKEEV, V. J. & KUPRASH, D. V. 2008. In vitro selection of optimal RelB/p52 DNA-binding motifs. *Biochem Biophys Res Commun*, 365, 583-8.
- BROCKMAN, J. A., SCHERER, D. C., MCKINSEY, T. A., HALL, S. M., QI, X., LEE, W.Y. & BALLARD, D. W. 1995. Coupling of a signal response domain in I kappa B alpha to multiple pathways for NF-kappa B activation. *Mol Cell Biol*, 15, 2809-18.
- BROWN, A. M., LINHOFF, M. W., STEIN, B., WRIGHT, K. L., BALDWIN, A. S., JR., BASTA, P. V. & TING, J. P. 1994. Function of NF-kappa B/Rel binding sites in the major histocompatibility complex class II invariant chain promoter is dependent on cell-specific binding of different NF-kappa B/Rel subunits. *Mol Cell Biol*, 14, 2926-35.
- BROWN, K., PARK, S., KANNO, T., FRANZOSO, G. & SIEBENLIST, U. 1993. Mutual regulation of the transcriptional activator NF-kappa B and its inhibitor, I kappa B-alpha. *Proc Natl Acad Sci U S A*, 90, 2532-6.
- BU, D. X., ERL, W., DE MARTIN, R., HANSSON, G. K. & YAN, Z. Q. 2005. IKKbetadependent NF-kappaB pathway controls vascular inflammation and intimal hyperplasia. *FASEB J*, 19, 1293-5.
- BUNTING, K., RAO, S., HARDY, K., WOLTRING, D., DENYER, G. S., WANG, J., GERONDAKIS, S. & SHANNON, M. F. 2007. Genome-wide analysis of gene expression in T cells to identify targets of the NF-kappa B transcription factor c-Rel. J Immunol, 178, 7097-109.
- BURKE, J. R., PATTOLI, M. A., GREGOR, K. R., BRASSIL, P. J., MACMASTER, J. F., MCINTYRE, K. W., YANG, X., IOTZOVA, V. S., CLARKE, W., STRNAD, J., QIU, Y. & ZUSI, F. C. 2003. BMS-345541 is a highly selective inhibitor of I kappa B

kinase that binds at an allosteric site of the enzyme and blocks NF-kappa B-dependent transcription in mice. *J Biol Chem*, 278, 1450-6.

- BUTTERY, L. D., SPRINGALL, D. R., CHESTER, A. H., EVANS, T. J., STANDFIELD, E. N., PARUMS, D. V., YACOUB, M. H. & POLAK, J. M. 1996. Inducible nitric oxide synthase is present within human atherosclerotic lesions and promotes the formation and activity of peroxynitrite. *Lab Invest*, 75, 77-85.
- BYUN, D. S., AHMED, N., NASSER, S., SHIN, J., AL-OBAIDI, S., GOEL, S., CORNER,
 G. A., WILSON, A. J., FLANAGAN, D. J., WILLIAMS, D. S., AUGENLICHT, L.
 H., VINCAN, E. & MARIADASON, J. M. 2011. Intestinal epithelial-specific PTEN inactivation results in tumor formation. *Am J Physiol Gastrointest Liver Physiol*.
- CABRITA, M. A. & CHRISTOFORI, G. 2003. Sprouty proteins: antagonists of endothelial cell signaling and more. *Thromb Haemost*, 90, 586-90.
- CADALBERT, L., SLOSS, C. M., CAMERON, P. & PLEVIN, R. 2005. Conditional expression of MAP kinase phosphatase-2 protects against genotoxic stress-induced apoptosis by binding and selective dephosphorylation of nuclear activated c-jun N-terminal kinase. *Cell Signal*, 17, 1254-64.
- CAMACHO, M., RODRIGUEZ, C., SALAZAR, J., MARTINEZ-GONZALEZ, J., RIBALTA, J., ESCUDERO, J. R., MASANA, L. & VILA, L. 2008. Retinoic acid induces PGI synthase expression in human endothelial cells. *J Lipid Res*, 49, 1707-14.
- CAO, Y., BONIZZI, G., SEAGROVES, T. N., GRETEN, F. R., JOHNSON, R., SCHMIDT, E. V. & KARIN, M. 2001. IKKalpha provides an essential link between RANK signaling and cyclin D1 expression during mammary gland development. *Cell*, 107, 763-75.
- CAPURSO, A. 2001. [The promise of statins]. Ital Heart J Suppl, 2, 224-9.
- CARDONE, M. H., ROY, N., STENNICKE, H. R., SALVESEN, G. S., FRANKE, T. F., STANBRIDGE, E., FRISCH, S. & REED, J. C. 1998. Regulation of cell death protease caspase-9 by phosphorylation. *Science*, 282, 1318-21.
- CARLSON, T. R., FENG, Y., MAISONPIERRE, P. C., MRKSICH, M. & MORLA, A. O. 2001. Direct cell adhesion to the angiopoietins mediated by integrins. *J Biol Chem*, 276, 26516-25.
- CASTRO, A. C., DANG, L. C., SOUCY, F., GRENIER, L., MAZDIYASNI, H., HOTTELET, M., PARENT, L., PIEN, C., PALOMBELLA, V. & ADAMS, J. 2003. Novel IKK inhibitors: beta-carbolines. *Bioorg Med Chem Lett*, 13, 2419-22.

- CATLEY, M. C., CHIVERS, J. E., HOLDEN, N. S., BARNES, P. J. & NEWTON, R. 2005. Validation of IKK beta as therapeutic target in airway inflammatory disease by adenoviral-mediated delivery of dominant-negative IKK beta to pulmonary epithelial cells. *Br J Pharmacol*, 145, 114-22.
- CATZ, S. D. & JOHNSON, J. L. 2001. Transcriptional regulation of bcl-2 by nuclear factor kappa B and its significance in prostate cancer. *Oncogene*, 20, 7342-51.
- CERVANTES, C. F., MARKWICK, P. R., SUE, S. C., MCCAMMON, J. A., DYSON, H. J.
 & KOMIVES, E. A. 2009. Functional dynamics of the folded ankyrin repeats of I kappa B alpha revealed by nuclear magnetic resonance. *Biochemistry*, 48, 8023-31.
- CHANG, Y. J., WU, M. S., LIN, J. T., SHEU, B. S., MUTA, T., INOUE, H. & CHEN, C. C. 2004. Induction of cyclooxygenase-2 overexpression in human gastric epithelial cells by Helicobacter pylori involves TLR2/TLR9 and c-Src-dependent nuclear factorkappaB activation. *Mol Pharmacol*, 66, 1465-77.
- CHEN, F. E. & GHOSH, G. 1999. Regulation of DNA binding by Rel/NF-kappaB transcription factors: structural views. *Oncogene*, 18, 6845-52.
- CHEN, F. E., HUANG, D. B., CHEN, Y. Q. & GHOSH, G. 1998. Crystal structure of p50/p65 heterodimer of transcription factor NF-kappaB bound to DNA. *Nature*, 391, 410-3.
- CHEN, Z., HAGLER, J., PALOMBELLA, V. J., MELANDRI, F., SCHERER, D., BALLARD, D. & MANIATIS, T. 1995. Signal-induced site-specific phosphorylation targets I kappa B alpha to the ubiquitin-proteasome pathway. *Genes Dev*, 9, 1586-97.
- CHEN, Z. J., PARENT, L. & MANIATIS, T. 1996. Site-specific phosphorylation of IkappaBalpha by a novel ubiquitination-dependent protein kinase activity. *Cell*, 84, 853-62.
- CHIA, M. C. 1998. The role of adhesion molecules in atherosclerosis. *Crit Rev Clin Lab Sci*, 35, 573-602.
- CHU, Z. L., MCKINSEY, T. A., LIU, L., GENTRY, J. J., MALIM, M. H. & BALLARD, D.
 W. 1997. Suppression of tumor necrosis factor-induced cell death by inhibitor of apoptosis c-IAP2 is under NF-kappaB control. *Proc Natl Acad Sci U S A*, 94, 10057-62.
- CLAUDIO, E., BROWN, K., PARK, S., WANG, H. & SIEBENLIST, U. 2002. BAFFinduced NEMO-independent processing of NF-kappa B2 in maturing B cells. *Nat Immunol*, 3, 958-65.

- COLEMAN, R. A., SMITH, W. L. & NARUMIYA, S. 1994. International Union of Pharmacology classification of prostanoid receptors: properties, distribution, and structure of the receptors and their subtypes. *Pharmacol Rev*, 46, 205-29.
- COMINACINI, L., PASINI, A. F., GARBIN, U., DAVOLI, A., TOSETTI, M. L., CAMPAGNOLA, M., RIGONI, A., PASTORINO, A. M., LO CASCIO, V. & SAWAMURA, T. 2000. Oxidized low density lipoprotein (ox-LDL) binding to ox-LDL receptor-1 in endothelial cells induces the activation of NF-kappaB through an increased production of intracellular reactive oxygen species. *J Biol Chem*, 275, 12633-8.
- COOPE, H. J., ATKINSON, P. G., HUHSE, B., BELICH, M., JANZEN, J., HOLMAN, M. J., KLAUS, G. G., JOHNSTON, L. H. & LEY, S. C. 2002. CD40 regulates the processing of NF-kappaB2 p100 to p52. *EMBO J*, 21, 5375-85.
- CROMHEEKE, K. M., KOCKX, M. M., DE MEYER, G. R., BOSMANS, J. M., BULT, H., BEELAERTS, W. J., VRINTS, C. J. & HERMAN, A. G. 1999. Inducible nitric oxide synthase colocalizes with signs of lipid oxidation/peroxidation in human atherosclerotic plaques. *Cardiovasc Res*, 43, 744-54.
- CUSHING, S. D., BERLINER, J. A., VALENTE, A. J., TERRITO, M. C., NAVAB, M., PARHAMI, F., GERRITY, R., SCHWARTZ, C. J. & FOGELMAN, A. M. 1990.
 Minimally modified low density lipoprotein induces monocyte chemotactic protein 1 in human endothelial cells and smooth muscle cells. *Proc Natl Acad Sci U S A*, 87, 5134-8.
- DATTA, S. R., DUDEK, H., TAO, X., MASTERS, S., FU, H., GOTOH, Y. & GREENBERG, M. E. 1997. Akt phosphorylation of BAD couples survival signals to the cell-intrinsic death machinery. *Cell*, 91, 231-41.
- DAVIES, M. J., RICHARDSON, P. D., WOOLF, N., KATZ, D. R. & MANN, J. 1993. Risk of thrombosis in human atherosclerotic plaques: role of extracellular lipid, macrophage, and smooth muscle cell content. *Br Heart J*, 69, 377-81.
- DAVIS, N., GHOSH, S., SIMMONS, D. L., TEMPST, P., LIOU, H. C., BALTIMORE, D. & BOSE, H. R., JR. 1991. Rel-associated pp40: an inhibitor of the rel family of transcription factors. *Science*, 253, 1268-71.
- DE MARTIN, R., HOETH, M., HOFER-WARBINEK, R. & SCHMID, J. A. 2000. The transcription factor NF-kappa B and the regulation of vascular cell function. *Arterioscler Thromb Vasc Biol*, 20, E83-8.

- DE MARTIN, R., SCHMID, J. A. & HOFER-WARBINEK, R. 1999. The NF-kappaB/Rel family of transcription factors in oncogenic transformation and apoptosis. *Mutat Res*, 437, 231-43.
- DE VILLIERS, W. J. & SMART, E. J. 1999. Macrophage scavenger receptors and foam cell formation. *J Leukoc Biol*, 66, 740-6.
- DEBUSK, L. M., MASSION, P. P. & LIN, P. C. 2008. IkappaB kinase-alpha regulates endothelial cell motility and tumor angiogenesis. *Cancer Res*, 68, 10223-8.
- DEJARDIN, E., DROIN, N. M., DELHASE, M., HAAS, E., CAO, Y., MAKRIS, C., LI, Z. W., KARIN, M., WARE, C. F. & GREEN, D. R. 2002. The lymphotoxin-beta receptor induces different patterns of gene expression via two NF-kappaB pathways. *Immunity*, 17, 525-35.
- DEMICCO, E. G., KAVANAGH, K. T., ROMIEU-MOUREZ, R., WANG, X., SHIN, S. R., LANDESMAN-BOLLAG, E., SELDIN, D. C. & SONENSHEIN, G. E. 2005. RelB/p52 NF-kappaB complexes rescue an early delay in mammary gland development in transgenic mice with targeted superrepressor IkappaB-alpha expression and promote carcinogenesis of the mammary gland. *Mol Cell Biol*, 25, 10136-47.
- DENK, A., GOEBELER, M., SCHMID, S., BERBERICH, I., RITZ, O., LINDEMANN, D., LUDWIG, S. & WIRTH, T. 2001. Activation of NF-kappa B via the Ikappa B kinase complex is both essential and sufficient for proinflammatory gene expression in primary endothelial cells. *J Biol Chem*, 276, 28451-8.
- DEREGOWSKI, V., DELHALLE, S., BENOIT, V., BOURS, V. & MERVILLE, M. P. 2002. Identification of cytokine-induced nuclear factor-kappaB target genes in ovarian and breast cancer cells. *Biochem Pharmacol*, 64, 873-81.
- DEVARY, Y., ROSETTE, C., DIDONATO, J. A. & KARIN, M. 1993. NF-kappa B activation by ultraviolet light not dependent on a nuclear signal. *Science*, 261, 1442-5.
- DEVIN, A., LIN, Y., YAMAOKA, S., LI, Z., KARIN, M. & LIU, Z. 2001. The alpha and beta subunits of IkappaB kinase (IKK) mediate TRAF2-dependent IKK recruitment to tumor necrosis factor (TNF) receptor 1 in response to TNF. *Mol Cell Biol*, 21, 3986-94.
- DHANASEKARAN, D. N. & REDDY, E. P. 2008. JNK signaling in apoptosis. *Oncogene*, 27, 6245-51.

- DIDONATO, J. A., HAYAKAWA, M., ROTHWARF, D. M., ZANDI, E. & KARIN, M. 1997. A cytokine-responsive IkappaB kinase that activates the transcription factor NF-kappaB. *Nature*, 388, 548-54.
- DIMMELER, S., FLEMING, I., FISSLTHALER, B., HERMANN, C., BUSSE, R. & ZEIHER, A. M. 1999. Activation of nitric oxide synthase in endothelial cells by Aktdependent phosphorylation. *Nature*, 399, 601-5.
- DIMMELER, S., HAENDELER, J., GALLE, J. & ZEIHER, A. M. 1997. Oxidized lowdensity lipoprotein induces apoptosis of human endothelial cells by activation of CPP32-like proteases. A mechanistic clue to the 'response to injury' hypothesis. *Circulation*, 95, 1760-3.
- DOBRZANSKI, P., RYSECK, R. P. & BRAVO, R. 1993. Both N- and C-terminal domains of RelB are required for full transactivation: role of the N-terminal leucine zipper-like motif. *Mol Cell Biol*, 13, 1572-82.
- DOBRZANSKI, P., RYSECK, R. P. & BRAVO, R. 1995. Specific inhibition of RelB/p52 transcriptional activity by the C-terminal domain of p100. *Oncogene*, 10, 1003-7.
- DONG, Z. M., CHAPMAN, S. M., BROWN, A. A., FRENETTE, P. S., HYNES, R. O. & WAGNER, D. D. 1998. The combined role of P- and E-selectins in atherosclerosis. J *Clin Invest*, 102, 145-52.
- DONOVAN, C. E., MARK, D. A., HE, H. Z., LIOU, H. C., KOBZIK, L., WANG, Y., DE SANCTIS, G. T., PERKINS, D. L. & FINN, P. W. 1999. NF-kappa B/Rel transcription factors: c-Rel promotes airway hyperresponsiveness and allergic pulmonary inflammation. *J Immunol*, 163, 6827-33.
- DZAU, V. J., BRAUN-DULLAEUS, R. C. & SEDDING, D. G. 2002. Vascular proliferation and atherosclerosis: new perspectives and therapeutic strategies. *Nat Med*, 8, 1249-56.
- ELICEIRI, B. P., PAUL, R., SCHWARTZBERG, P. L., HOOD, J. D., LENG, J. & CHERESH, D. A. 1999. Selective requirement for Src kinases during VEGF-induced angiogenesis and vascular permeability. *Mol Cell*, 4, 915-24.
- ELICES, M. J., OSBORN, L., TAKADA, Y., CROUSE, C., LUHOWSKYJ, S., HEMLER, M. E. & LOBB, R. R. 1990. VCAM-1 on activated endothelium interacts with the leukocyte integrin VLA-4 at a site distinct from the VLA-4/fibronectin binding site. *Cell*, 60, 577-84.
- ERL, W., HANSSON, G. K., DE MARTIN, R., DRAUDE, G., WEBER, K. S. & WEBER,C. 1999. Nuclear factor-kappa B regulates induction of apoptosis and inhibitor of apoptosis protein-1 expression in vascular smooth muscle cells. *Circ Res*, 84, 668-77.

- FADINI, G. P., AGOSTINI, C., SARTORE, S. & AVOGARO, A. 2007. Endothelial progenitor cells in the natural history of atherosclerosis. *Atherosclerosis*, 194, 46-54.
- FAN, C. M. & MANIATIS, T. 1991. Generation of p50 subunit of NF-kappa B by processing of p105 through an ATP-dependent pathway. *Nature*, 354, 395-8.
- FAVOT, L., KERAVIS, T., HOLL, V., LE BEC, A. & LUGNIER, C. 2003. VEGF-induced HUVEC migration and proliferation are decreased by PDE2 and PDE4 inhibitors. *Thromb Haemost*, 90, 334-43.
- FAYARD, E., XUE, G., PARCELLIER, A., BOZULIC, L. & HEMMINGS, B. A. 2010. Protein kinase B (PKB/Akt), a key mediator of the PI3K signaling pathway. *Curr Top Microbiol Immunol*, 346, 31-56.
- FERNANDEZ-MAJADA, V., AGUILERA, C., VILLANUEVA, A., VILARDELL, F., ROBERT-MORENO, A., AYTES, A., REAL, F. X., CAPELLA, G., MAYO, M. W., ESPINOSA, L. & BIGAS, A. 2007. Nuclear IKK activity leads to dysregulated notchdependent gene expression in colorectal cancer. *Proc Natl Acad Sci U S A*, 104, 276-81.
- FIRESTEIN, G. S. 2003. Evolving concepts of rheumatoid arthritis. Nature, 423, 356-61.
- FISHMAN, A. P. 1982. Endothelium: a distributed organ of diverse capabilities. *Ann N Y Acad Sci*, 401, 1-8.
- FLAVAHAN, N. A. 1992. Atherosclerosis or lipoprotein-induced endothelial dysfunction. Potential mechanisms underlying reduction in EDRF/nitric oxide activity. *Circulation*, 85, 1927-38.
- FONG, A. & SUN, S. C. 2002. Genetic evidence for the essential role of beta-transducin repeat-containing protein in the inducible processing of NF-kappa B2/p100. *J Biol Chem*, 277, 22111-4.
- FONSECA, L. C., DADARKAR, S. S., LOBO, A. S., MISHRA, P. B., THAKKAR, A. D., CHANDRABABU, S. & PADIGARU, M. 2011. NF-kappaB-mediated antiinflammatory activity of the sesquiterpene lactone 7-hydroxyfrullanolide. *Eur J Pharmacol*, 657, 41-50.
- FORESTI, R., SARATHCHANDRA, P., CLARK, J. E., GREEN, C. J. & MOTTERLINI, R. 1999. Peroxynitrite induces haem oxygenase-1 in vascular endothelial cells: a link to apoptosis. *Biochem J*, 339 (Pt 3), 729-36.
- FREYSCHMIDT, E. J., MATHIAS, C. B., MACARTHUR, D. H., LAOUAR, A., NARASIMHASWAMY, M., WEIH, F. & OETTGEN, H. C. 2007. Skin

inflammation in RelB(-/-) mice leads to defective immunity and impaired clearance of vaccinia virus. *J Allergy Clin Immunol*, 119, 671-9.

- FROSTEGARD, J., ULFGREN, A. K., NYBERG, P., HEDIN, U., SWEDENBORG, J., ANDERSSON, U. & HANSSON, G. K. 1999. Cytokine expression in advanced human atherosclerotic plaques: dominance of pro-inflammatory (Th1) and macrophage-stimulating cytokines. *Atherosclerosis*, 145, 33-43.
- FRUEBIS, J., SILVESTRE, M., SHELTON, D., NAPOLI, C. & PALINSKI, W. 1999. Inhibition of VCAM-1 expression in the arterial wall is shared by structurally different antioxidants that reduce early atherosclerosis in NZW rabbits. *J Lipid Res*, 40, 1958-66.
- FUNAHASHI, Y., SHAWBER, C. J., SHARMA, A., KANAMARU, E., CHOI, Y. K. & KITAJEWSKI, J. 2011. Notch modulates VEGF action in endothelial cells by inducing Matrix Metalloprotease activity. *Vasc Cell*, 3, 2.
- GAMBLE, C., MCINTOSH, K., SCOTT, R., HO, K. H., PLEVIN, R. & PAUL, A. 2012. Inhibitory kappa B kinases as targets for pharmacological regulation. *Br J Pharmacol*, 165, 802-19.
- GANCHI, P. A., SUN, S. C., GREENE, W. C. & BALLARD, D. W. 1992. I kappa B/MAD-3 masks the nuclear localization signal of NF-kappa B p65 and requires the transactivation domain to inhibit NF-kappa B p65 DNA binding. *Mol Biol Cell*, 3, 1339-52.
- GANCHI, P. A., SUN, S. C., GREENE, W. C. & BALLARD, D. W. 1993. A novel NFkappa B complex containing p65 homodimers: implications for transcriptional control at the level of subunit dimerization. *Mol Cell Biol*, 13, 7826-35.
- GARCIA, M. G., ALANIZ, L., LOPES, E. C., BLANCO, G., HAJOS, S. E. & ALVAREZ,
 E. 2005. Inhibition of NF-kappaB activity by BAY 11-7082 increases apoptosis in multidrug resistant leukemic T-cell lines. *Leuk Res*, 29, 1425-34.
- GAREUS, R., KOTSAKI, E., XANTHOULEA, S., VAN DER MADE, I., GIJBELS, M. J., KARDAKARIS, R., POLYKRATIS, A., KOLLIAS, G., DE WINTHER, M. P. & PASPARAKIS, M. 2008. Endothelial cell-specific NF-kappaB inhibition protects mice from atherosclerosis. *Cell Metab*, 8, 372-83.
- GASPARIAN, A. V., GURYANOVA, O. A., CHEBOTAEV, D. V., SHISHKIN, A. A., YEMELYANOV, A. Y. & BUDUNOVA, I. V. 2009. Targeting transcription factor NFkappaB: comparative analysis of proteasome and IKK inhibitors. *Cell Cycle*, 8, 1559-66.

- GATTO, S., SCAPPINI, B., PHAM, L., ONIDA, F., MILELLA, M., BALL, G., RICCI, C., DIVOKY, V., VERSTOVSEK, S., KANTARJIAN, H. M., KEATING, M. J., CORTES-FRANCO, J. E. & BERAN, M. 2003. The proteasome inhibitor PS-341 inhibits growth and induces apoptosis in Bcr/Abl-positive cell lines sensitive and resistant to imatinib mesylate. *Haematologica*, 88, 853-63.
- GAUTREAU, A., POULLET, P., LOUVARD, D. & ARPIN, M. 1999. Ezrin, a plasma membrane-microfilament linker, signals cell survival through the phosphatidylinositol 3-kinase/Akt pathway. *Proc Natl Acad Sci U S A*, 96, 7300-5.
- GERONDAKIS, S., GROSSMANN, M., NAKAMURA, Y., POHL, T. & GRUMONT, R. 1999. Genetic approaches in mice to understand Rel/NF-kappaB and IkappaB function: transgenics and knockouts. *Oncogene*, 18, 6888-95.
- GHISELLI, G., SCHAEFER, E. J., GASCON, P. & BRESER, H. B., JR. 1981. Type III hyperlipoproteinemia associated with apolipoprotein E deficiency. *Science*, 214, 1239-41.
- GHOSH, S., GIFFORD, A. M., RIVIERE, L. R., TEMPST, P., NOLAN, G. P. & BALTIMORE, D. 1990. Cloning of the p50 DNA binding subunit of NF-kappa B: homology to rel and dorsal. *Cell*, 62, 1019-29.
- GHOSH, S. & KARIN, M. 2002. Missing pieces in the NF-kappaB puzzle. *Cell*, 109 Suppl, S81-96.
- GIET, R. & PRIGENT, C. 1999. Aurora/Ipl1p-related kinases, a new oncogenic family of mitotic serine-threonine kinases. *J Cell Sci*, 112 (Pt 21), 3591-601.
- GILROY, D. W., COLVILLE-NASH, P. R., WILLIS, D., CHIVERS, J., PAUL-CLARK, M. J. & WILLOUGHBY, D. A. 1999. Inducible cyclooxygenase may have antiinflammatory properties. *Nat Med*, 5, 698-701.
- GILSTON, V., JONES, H. W., SOO, C. C., COUMBE, A., BLADES, S., KALTSCHMIDT, C., BAEUERLE, P. A., MORRIS, C. J., BLAKE, D. R. & WINYARD, P. G. 1997. NF-kappa B activation in human knee-joint synovial tissue during the early stage of joint inflammation. *Biochem Soc Trans*, 25, 518S.
- GLOIRE, G., HORION, J., EL MJIYAD, N., BEX, F., CHARIOT, A., DEJARDIN, E. & PIETTE, J. 2007. Promoter-dependent effect of IKKalpha on NF-kappaB/p65 DNA binding. *J Biol Chem*, 282, 21308-18.
- GOH, F. G., NG, P. Y., NILSSON, M., KANKE, T. & PLEVIN, R. 2009. Dual effect of the novel peptide antagonist K-14585 on proteinase-activated receptor-2-mediated signalling. *Br J Pharmacol*, 158, 1695-704.

- GROSJEAN, J., KIRIAKIDIS, S., REILLY, K., FELDMANN, M. & PALEOLOG, E. 2006. Vascular endothelial growth factor signalling in endothelial cell survival: a role for NFkappaB. *Biochem Biophys Res Commun*, 340, 984-94.
- GROSS, S. S. & WOLIN, M. S. 1995. Nitric oxide: pathophysiological mechanisms. *Annu Rev Physiol*, 57, 737-69.
- GRUMONT, R. J. & GERONDAKIS, S. 1994. Alternative splicing of RNA transcripts encoded by the murine p105 NF-kappa B gene generates I kappa B gamma isoforms with different inhibitory activities. *Proc Natl Acad Sci U S A*, 91, 4367-71.
- GUPTA, S., SETH, A. & DAVIS, R. J. 1993. Transactivation of gene expression by Myc is inhibited by mutation at the phosphorylation sites Thr-58 and Ser-62. *Proc Natl Acad Sci U S A*, 90, 3216-20.
- HALL, G., SINGH, I. S., HESTER, L., HASDAY, J. D. & ROGERS, T. B. 2005. InhibitorkappaB kinase-beta regulates LPS-induced TNF-alpha production in cardiac myocytes through modulation of NF-kappaB p65 subunit phosphorylation. Am J Physiol Heart Circ Physiol, 289, H2103-11.
- HAN, Z., BOYLE, D. L., MANNING, A. M. & FIRESTEIN, G. S. 1998. AP-1 and NFkappaB regulation in rheumatoid arthritis and murine collagen-induced arthritis. *Autoimmunity*, 28, 197-208.
- HANAFUSA, H., TORII, S., YASUNAGA, T. & NISHIDA, E. 2002. Sprouty1 and Sprouty2 provide a control mechanism for the Ras/MAPK signalling pathway. *Nat Cell Biol*, 4, 850-8.
- HANDEL, M. L., MCMORROW, L. B. & GRAVALLESE, E. M. 1995. Nuclear factorkappa B in rheumatoid synovium. Localization of p50 and p65. *Arthritis Rheum*, 38, 1762-70.
- HANSEN, S. K., BAEUERLE, P. A. & BLASI, F. 1994. Purification, reconstitution, and I kappa B association of the c-Rel-p65 (RelA) complex, a strong activator of transcription. *Mol Cell Biol*, 14, 2593-603.
- HARRINGTON, J. R. 2000. The role of MCP-1 in atherosclerosis. Stem Cells, 18, 65-6.
- HARRISON, D. G. 1997. Cellular and molecular mechanisms of endothelial cell dysfunction. J Clin Invest, 100, 2153-7.
- HART, L. A., KRISHNAN, V. L., ADCOCK, I. M., BARNES, P. J. & CHUNG, K. F. 1998. Activation and localization of transcription factor, nuclear factor-kappaB, in asthma. *Am J Respir Crit Care Med*, 158, 1585-92.

- HATADA, E. N., NIETERS, A., WULCZYN, F. G., NAUMANN, M., MEYER, R., NUCIFORA, G., MCKEITHAN, T. W. & SCHEIDEREIT, C. 1992. The ankyrin repeat domains of the NF-kappa B precursor p105 and the protooncogene bcl-3 act as specific inhibitors of NF-kappa B DNA binding. *Proc Natl Acad Sci U S A*, 89, 2489-93.
- HEMMERICH, S., PAAVOLA, C., BLOOM, A., BHAKTA, S., FREEDMAN, R., GRUNBERGER, D., KRSTENANSKY, J., LEE, S., MCCARLEY, D., MULKINS, M., WONG, B., PEASE, J., MIZOUE, L., MIRZADEGAN, T., POLSKY, I., THOMPSON, K., HANDEL, T. M. & JARNAGIN, K. 1999. Identification of residues in the monocyte chemotactic protein-1 that contact the MCP-1 receptor, CCR2. *Biochemistry*, 38, 13013-25.
- HENKEL, T., ZABEL, U., VAN ZEE, K., MULLER, J. M., FANNING, E. & BAEUERLE,P. A. 1992. Intramolecular masking of the nuclear location signal and dimerization domain in the precursor for the p50 NF-kappa B subunit. *Cell*, 68, 1121-33.
- HINZ, M., KRAPPMANN, D., EICHTEN, A., HEDER, A., SCHEIDEREIT, C. & STRAUSS, M. 1999. NF-kappaB function in growth control: regulation of cyclin D1 expression and G0/G1-to-S-phase transition. *Mol Cell Biol*, 19, 2690-8.
- HIRATA, Y., MAEDA, S., OHMAE, T., SHIBATA, W., YANAI, A., OGURA, K., YOSHIDA, H., KAWABE, T. & OMATA, M. 2006. Helicobacter pylori induces IkappaB kinase alpha nuclear translocation and chemokine production in gastric epithelial cells. *Infect Immun*, 74, 1452-61.
- HOEBEN, A., LANDUYT, B., HIGHLEY, M. S., WILDIERS, H., VAN OOSTEROM, A. T. & DE BRUIJN, E. A. 2004. Vascular endothelial growth factor and angiogenesis. *Pharmacol Rev*, 56, 549-80.
- HOEFEN, R. J. & BERK, B. C. 2002. The role of MAP kinases in endothelial activation. *Vascul Pharmacol*, 38, 271-3.
- HOLTMANN, M. H. & NEURATH, M. F. 2004. Differential TNF-signaling in chronic inflammatory disorders. *Curr Mol Med*, 4, 439-44.
- HORIE, R., WATANABE, T. & UMEZAWA, K. 2006. Blocking NF-kappaB as a potential strategy to treat adult T-cell leukemia/lymphoma. *Drug News Perspect*, 19, 201-9.
- HORWITZ, B. H., SCOTT, M. L., CHERRY, S. R., BRONSON, R. T. & BALTIMORE, D. 1997. Failure of lymphopoiesis after adoptive transfer of NF-kappaB-deficient fetal liver cells. *Immunity*, 6, 765-72.

- HOU, B., EREN, M., PAINTER, C. A., COVINGTON, J. W., DIXON, J. D., SCHOENHARD, J. A. & VAUGHAN, D. E. 2004. Tumor necrosis factor alpha activates the human plasminogen activator inhibitor-1 gene through a distal nuclear factor kappaB site. *J Biol Chem*, 279, 18127-36.
- HU, Y., BAUD, V., DELHASE, M., ZHANG, P., DEERINCK, T., ELLISMAN, M., JOHNSON, R. & KARIN, M. 1999. Abnormal morphogenesis but intact IKK activation in mice lacking the IKKalpha subunit of IkappaB kinase. *Science*, 284, 316-20.
- HU, Y., BAUD, V., OGA, T., KIM, K. I., YOSHIDA, K. & KARIN, M. 2001. IKKalpha controls formation of the epidermis independently of NF-kappaB. *Nature*, 410, 710-4.
- HUANG, W. C., JU, T. K., HUNG, M. C. & CHEN, C. C. 2007. Phosphorylation of CBP by IKKalpha promotes cell growth by switching the binding preference of CBP from p53 to NF-kappaB. *Mol Cell*, 26, 75-87.
- HUXFORD, T., HUANG, D. B., MALEK, S. & GHOSH, G. 1998. The crystal structure of the IkappaBalpha/NF-kappaB complex reveals mechanisms of NF-kappaB inactivation. *Cell*, 95, 759-70.
- IADEMARCO, M. F., MCQUILLAN, J. J., ROSEN, G. D. & DEAN, D. C. 1992. Characterization of the promoter for vascular cell adhesion molecule-1 (VCAM-1). J Biol Chem, 267, 16323-9.
- IANARO, A., TERSIGNI, M., BELARDO, G., DI MARTINO, S., NAPOLITANO, M., PALMIERI, G., SINI, M., DE MAIO, A., OMBRA, M., GENTILCORE, G., CAPONE, M., ASCIERTO, M., SATRIANO, R. A., FARINA, B., FARAONE-MENNELLA, M., ASCIERTO, P. A. & IALENTI, A. 2009. NEMO-binding domain peptide inhibits proliferation of human melanoma cells. *Cancer Lett*, 274, 331-6.
- INOUE, J., KERR, L. D., KAKIZUKA, A. & VERMA, I. M. 1992a. I kappa B gamma, a 70 kd protein identical to the C-terminal half of p110 NF-kappa B: a new member of the I kappa B family. *Cell*, 68, 1109-20.
- INOUE, J., KERR, L. D., RASHID, D., DAVIS, N., BOSE, H. R., JR. & VERMA, I. M. 1992b. Direct association of pp40/I kappa B beta with rel/NF-kappa B transcription factors: role of ankyrin repeats in the inhibition of DNA binding activity. *Proc Natl Acad Sci U S A*, 89, 4333-7.
- JACQUE, E., TCHENIO, T., PITON, G., ROMEO, P. H. & BAUD, V. 2005. RelA repression of RelB activity induces selective gene activation downstream of TNF receptors. *Proc Natl Acad Sci U S A*, 102, 14635-40.

- JAFFE, E. A. & WEKSLER, B. B. 1979. Recovery of endothelial cell prostacyclin production after inhibition by low doses of aspirin. *J Clin Invest*, 63, 532-5.
- JAFFRAY, E., WOOD, K. M. & HAY, R. T. 1995. Domain organization of I kappa B alpha and sites of interaction with NF-kappa B p65. *Mol Cell Biol*, 15, 2166-72.
- JANABI, M., YAMASHITA, S., HIRANO, K., SAKAI, N., HIRAOKA, H., MATSUMOTO, K., ZHANG, Z., NOZAKI, S. & MATSUZAWA, Y. 2000. Oxidized LDL-induced NF-kappa B activation and subsequent expression of proinflammatory genes are defective in monocyte-derived macrophages from CD36-deficient patients. *Arterioscler Thromb Vasc Biol*, 20, 1953-60.
- JAWIEN, J., GAJDA, M., MATEUSZUK, L., OLSZANECKI, R., JAKUBOWSKI, A., SZLACHCIC, A., KORABIOWSKA, M. & KORBUT, R. 2005. Inhibition of nuclear factor-kappaB attenuates artherosclerosis in apoE/LDLR - double knockout mice. J Physiol Pharmacol, 56, 483-9.
- JIMI, E., IKEBE, T., TAKAHASHI, N., HIRATA, M., SUDA, T. & KOGA, T. 1996. Interleukin-1 alpha activates an NF-kappaB-like factor in osteoclast-like cells. *J Biol Chem*, 271, 4605-8.
- JOHNSON, J. L., DEVEL, L., CZARNY, B., GEORGE, S. J., JACKSON, C. L., ROGAKOS, V., BEAU, F., YIOTAKIS, A., NEWBY, A. C. & DIVE, V. 2011. A selective matrix metalloproteinase-12 inhibitor retards atherosclerotic plaque development in apolipoprotein E-knockout mice. *Arterioscler Thromb Vasc Biol*, 31, 528-35.
- KANAYAMA, A., SETH, R. B., SUN, L., EA, C. K., HONG, M., SHAITO, A., CHIU, Y.H., DENG, L. & CHEN, Z. J. 2004. TAB2 and TAB3 activate the NF-kappaB pathway through binding to polyubiquitin chains. *Mol Cell*, 15, 535-48.
- KANG, S. M., TRAN, A. C., GRILLI, M. & LENARDO, M. J. 1992. NF-kappa B subunit regulation in nontransformed CD4+ T lymphocytes. *Science*, 256, 1452-6.
- KANNO, S., ODA, N., ABE, M., TERAI, Y., ITO, M., SHITARA, K., TABAYASHI, K., SHIBUYA, M. & SATO, Y. 2000. Roles of two VEGF receptors, Flt-1 and KDR, in the signal transduction of VEGF effects in human vascular endothelial cells. *Oncogene*, 19, 2138-46.
- KANTERS, E., PASPARAKIS, M., GIJBELS, M. J., VERGOUWE, M. N., PARTOUNS-HENDRIKS, I., FIJNEMAN, R. J., CLAUSEN, B. E., FORSTER, I., KOCKX, M. M., RAJEWSKY, K., KRAAL, G., HOFKER, M. H. & DE WINTHER, M. P. 2003.

Inhibition of NF-kappaB activation in macrophages increases atherosclerosis in LDL receptor-deficient mice. *J Clin Invest*, 112, 1176-85.

- KAWAKAMI, K., SCHEIDEREIT, C. & ROEDER, R. G. 1988. Identification and purification of a human immunoglobulin-enhancer-binding protein (NF-kappa B) that activates transcription from a human immunodeficiency virus type 1 promoter in vitro. *Proc Natl Acad Sci U S A*, 85, 4700-4.
- KAYAGAKI, N., YAN, M., SESHASAYEE, D., WANG, H., LEE, W., FRENCH, D. M., GREWAL, I. S., COCHRAN, A. G., GORDON, N. C., YIN, J., STAROVASNIK, M. A. & DIXIT, V. M. 2002. BAFF/BLyS receptor 3 binds the B cell survival factor BAFF ligand through a discrete surface loop and promotes processing of NFkappaB2. *Immunity*, 17, 515-24.
- KHARBANDA, S., SAXENA, S., YOSHIDA, K., PANDEY, P., KANEKI, M., WANG, Q., CHENG, K., CHEN, Y. N., CAMPBELL, A., SUDHA, T., YUAN, Z. M., NARULA, J., WEICHSELBAUM, R., NALIN, C. & KUFE, D. 2000. Translocation of SAPK/JNK to mitochondria and interaction with Bcl-x(L) in response to DNA damage. J Biol Chem, 275, 322-7.
- KIM, B. J., RYU, S. W. & SONG, B. J. 2006. JNK- and p38 kinase-mediated phosphorylation of Bax leads to its activation and mitochondrial translocation and to apoptosis of human hepatoma HepG2 cells. *J Biol Chem*, 281, 21256-65.
- KIM, I., KIM, J. H., RYU, Y. S., LIU, M. & KOH, G. Y. 2000. Tumor necrosis factor-alpha upregulates angiopoietin-2 in human umbilical vein endothelial cells. *Biochem Biophys Res Commun*, 269, 361-5.
- KISHIDA, S., SANJO, H., AKIRA, S., MATSUMOTO, K. & NINOMIYA-TSUJI, J. 2005. TAK1-binding protein 2 facilitates ubiquitination of TRAF6 and assembly of TRAF6 with IKK in the IL-1 signaling pathway. *Genes Cells*, 10, 447-54.
- KISSELEVA, T., SONG, L., VORONTCHIKHINA, M., FEIRT, N., KITAJEWSKI, J. & SCHINDLER, C. 2006. NF-kappaB regulation of endothelial cell function during LPS-induced toxemia and cancer. *J Clin Invest*, 116, 2955-63.
- KO, H. M., JUNG, H. H., SEO, K. H., KANG, Y. R., KIM, H. A., PARK, S. J., LEE, H. K. & IM, S. Y. 2006. Platelet-activating factor-induced NF-kappaB activation enhances VEGF expression through a decrease in p53 activity. *FEBS Lett*, 580, 3006-12.
- KREUZ, S., SIEGMUND, D., SCHEURICH, P. & WAJANT, H. 2001. NF-kappaB inducers upregulate cFLIP, a cycloheximide-sensitive inhibitor of death receptor signaling. *Mol Cell Biol*, 21, 3964-73.

- KRISHNASWAMY, G., KELLEY, J., YERRA, L., SMITH, J. K. & CHI, D. S. 1999. Human endothelium as a source of multifunctional cytokines: molecular regulation and possible role in human disease. *J Interferon Cytokine Res*, 19, 91-104.
- KUBES, P., SUZUKI, M. & GRANGER, D. N. 1991. Nitric oxide: an endogenous modulator of leukocyte adhesion. *Proc Natl Acad Sci U S A*, 88, 4651-5.
- KUGIYAMA, K., YASUE, H., OKUMURA, K., OGAWA, H., FUJIMOTO, K., NAKAO, K., YOSHIMURA, M., MOTOYAMA, T., INOBE, Y. & KAWANO, H. 1996. Nitric oxide activity is deficient in spasm arteries of patients with coronary spastic angina. *Circulation*, 94, 266-71.
- KUNSCH, C., RUBEN, S. M. & ROSEN, C. A. 1992. Selection of optimal kappa B/Rel DNA-binding motifs: interaction of both subunits of NF-kappa B with DNA is required for transcriptional activation. *Mol Cell Biol*, 12, 4412-21.
- KUTUK, O. & BASAGA, H. 2003. Inflammation meets oxidation: NF-kappaB as a mediator of initial lesion development in atherosclerosis. *Trends Mol Med*, 9, 549-57.
- KWAK, Y. T., LI, R., BECERRA, C. R., TRIPATHY, D., FRENKEL, E. P. & VERMA, U.
 N. 2005. IkappaB kinase alpha regulates subcellular distribution and turnover of cyclin D1 by phosphorylation. *J Biol Chem*, 280, 33945-52.
- LAWRENCE, T., BEBIEN, M., LIU, G. Y., NIZET, V. & KARIN, M. 2005. IKKalpha limits macrophage NF-kappaB activation and contributes to the resolution of inflammation. *Nature*, 434, 1138-43.
- LERNBECHER, T., MULLER, U. & WIRTH, T. 1993. Distinct NF-kappa B/Rel transcription factors are responsible for tissue-specific and inducible gene activation. *Nature*, 365, 767-70.
- LEVINE, G. N., KEANEY, J. F., JR. & VITA, J. A. 1995. Cholesterol reduction in cardiovascular disease. Clinical benefits and possible mechanisms. N Engl J Med, 332, 512-21.
- LI, C. C., DAI, R. M. & LONGO, D. L. 1995. Inactivation of NF-kappa B inhibitor I kappa B alpha: ubiquitin-dependent proteolysis and its degradation product. *Biochem Biophys Res Commun*, 215, 292-301.
- LI, J., LIU, Z., JIANG, S., CORTESINI, R., LEDERMAN, S. & SUCIU-FOCA, N. 1999a. T suppressor lymphocytes inhibit NF-kappa B-mediated transcription of CD86 gene in APC. *J Immunol*, 163, 6386-92.
- LI, J., YEN, C., LIAW, D., PODSYPANINA, K., BOSE, S., WANG, S. I., PUC, J., MILIARESIS, C., RODGERS, L., MCCOMBIE, R., BIGNER, S. H.,

GIOVANELLA, B. C., ITTMANN, M., TYCKO, B., HIBSHOOSH, H., WIGLER, M. H. & PARSONS, R. 1997. PTEN, a putative protein tyrosine phosphatase gene mutated in human brain, breast, and prostate cancer. *Science*, 275, 1943-7.

- LI, Q., LU, Q., BOTTERO, V., ESTEPA, G., MORRISON, L., MERCURIO, F. & VERMA,
 I. M. 2005. Enhanced NF-kappaB activation and cellular function in macrophages lacking IkappaB kinase 1 (IKK1). *Proc Natl Acad Sci U S A*, 102, 12425-30.
- LI, Q., LU, Q., HWANG, J. Y., BUSCHER, D., LEE, K. F., IZPISUA-BELMONTE, J. C. & VERMA, I. M. 1999b. IKK1-deficient mice exhibit abnormal development of skin and skeleton. *Genes Dev*, 13, 1322-8.
- LI, Q., VAN ANTWERP, D., MERCURIO, F., LEE, K. F. & VERMA, I. M. 1999c. Severe liver degeneration in mice lacking the IkappaB kinase 2 gene. *Science*, 284, 321-5.
- LI, X., MASSA, P. E., HANIDU, A., PEET, G. W., ARO, P., SAVITT, A., MISCHE, S., LI, J. & MARCU, K. B. 2002. IKKalpha, IKKbeta, and NEMO/IKKgamma are each required for the NF-kappa B-mediated inflammatory response program. *J Biol Chem*, 277, 45129-40.
- LI, Y. Q., ZHANG, Z. X., XU, Y. J., NI, W., CHEN, S. X., YANG, Z. & MA, D. 2006. N-Acetyl-L-cysteine and pyrrolidine dithiocarbamate inhibited nuclear factor-kappaB activation in alveolar macrophages by different mechanisms. *Acta Pharmacol Sin*, 27, 339-46.
- LI, Z. & NABEL, G. J. 1997. A new member of the I kappaB protein family, I kappaB epsilon, inhibits RelA (p65)-mediated NF-kappaB transcription. *Mol Cell Biol*, 17, 6184-90.
- LI, Z. W., CHU, W., HU, Y., DELHASE, M., DEERINCK, T., ELLISMAN, M., JOHNSON, R. & KARIN, M. 1999d. The IKKbeta subunit of IkappaB kinase (IKK) is essential for nuclear factor kappaB activation and prevention of apoptosis. *J Exp Med*, 189, 1839-45.
- LIANG, C., ZHANG, M. & SUN, S. C. 2006. beta-TrCP binding and processing of NFkappaB2/p100 involve its phosphorylation at serines 866 and 870. *Cell Signal*, 18, 1309-17.
- LIANG, Z., BROOKS, J., WILLARD, M., LIANG, K., YOON, Y., KANG, S. & SHIM, H. 2007. CXCR4/CXCL12 axis promotes VEGF-mediated tumor angiogenesis through Akt signaling pathway. *Biochem Biophys Res Commun*, 359, 716-22.
- LIBBY, P. 2001. Managing the risk of atherosclerosis: the role of high-density lipoprotein. *Am J Cardiol*, 88, 3N-8N.

LIBBY, P. 2002. Inflammation in atherosclerosis. *Nature*, 420, 868-74.

- LIBBY, P., AIKAWA, M. & JAIN, M. K. 2006. Vascular endothelium and atherosclerosis. *Handb Exp Pharmacol*, 285-306.
- LIBBY, P., DICARLI, M. & WEISSLEDER, R. 2010a. The vascular biology of atherosclerosis and imaging targets. *J Nucl Med*, 51 Suppl 1, 33S-37S.
- LIBBY, P., OKAMOTO, Y., ROCHA, V. Z. & FOLCO, E. 2010b. Inflammation in atherosclerosis: transition from theory to practice. *Circ J*, 74, 213-20.
- LIDDLE, J., BAMBOROUGH, P., BARKER, M. D., CAMPOS, S., COUSINS, R. P., CUTLER, G. J., HOBBS, H., HOLMES, D. S., IOANNOU, C., MELLOR, G. W., MORSE, M. A., PAYNE, J. J., PRITCHARD, J. M., SMITH, K. J., TAPE, D. T., WHITWORTH, C. & WILLIAMSON, R. A. 2009. 4-Phenyl-7-azaindoles as potent and selective IKK2 inhibitors. *Bioorg Med Chem Lett*, 19, 2504-8.
- LIEKENS, S., SCHOLS, D. & HATSE, S. 2010. CXCL12-CXCR4 axis in angiogenesis, metastasis and stem cell mobilization. *Curr Pharm Des*, 16, 3903-20.
- LING, L., CAO, Z. & GOEDDEL, D. V. 1998. NF-kappaB-inducing kinase activates IKKalpha by phosphorylation of Ser-176. *Proc Natl Acad Sci U S A*, 95, 3792-7.
- LIU, B., PARK, E., ZHU, F., BUSTOS, T., LIU, J., SHEN, J., FISCHER, S. M. & HU, Y. 2006. A critical role for I kappaB kinase alpha in the development of human and mouse squamous cell carcinomas. *Proc Natl Acad Sci U S A*, 103, 17202-7.
- LIU, B., YANG, Y., CHERNISHOF, V., LOO, R. R., JANG, H., TAHK, S., YANG, R., MINK, S., SHULTZ, D., BELLONE, C. J., LOO, J. A. & SHUAI, K. 2007. Proinflammatory stimuli induce IKKalpha-mediated phosphorylation of PIAS1 to restrict inflammation and immunity. *Cell*, 129, 903-14.
- LIU, S. F., YE, X. & MALIK, A. B. 1999. Inhibition of NF-kappaB activation by pyrrolidine dithiocarbamate prevents In vivo expression of proinflammatory genes. *Circulation*, 100, 1330-7.
- LIZARD, G., LEMAIRE, S., MONIER, S., GUELDRY, S., NEEL, D. & GAMBERT, P. 1997. Induction of apoptosis and of interleukin-1beta secretion by 7betahydroxycholesterol and 7-ketocholesterol: partial inhibition by Bcl-2 overexpression. *FEBS Lett*, 419, 276-80.
- LUFTIG, M., YASUI, T., SONI, V., KANG, M. S., JACOBSON, N., CAHIR-MCFARLAND, E., SEED, B. & KIEFF, E. 2004. Epstein-Barr virus latent infection membrane protein 1 TRAF-binding site induces NIK/IKK alpha-dependent noncanonical NF-kappaB activation. *Proc Natl Acad Sci U S A*, 101, 141-6.

- LUOMA, J. S., STRALIN, P., MARKLUND, S. L., HILTUNEN, T. P., SARKIOJA, T. & YLA-HERTTUALA, S. 1998. Expression of extracellular SOD and iNOS in macrophages and smooth muscle cells in human and rabbit atherosclerotic lesions: colocalization with epitopes characteristic of oxidized LDL and peroxynitrite-modified proteins. *Arterioscler Thromb Vasc Biol*, 18, 157-67.
- MACKENZIE, C. J., RITCHIE, E., PAUL, A. & PLEVIN, R. 2007. IKKalpha and IKKbeta function in TNFalpha-stimulated adhesion molecule expression in human aortic smooth muscle cells. *Cell Signal*, 19, 75-80.
- MACMASTER, J. F., DAMBACH, D. M., LEE, D. B., BERRY, K. K., QIU, Y., ZUSI, F. C.
 & BURKE, J. R. 2003. An inhibitor of IkappaB kinase, BMS-345541, blocks endothelial cell adhesion molecule expression and reduces the severity of dextran sulfate sodium-induced colitis in mice. *Inflamm Res*, 52, 508-11.
- MADGE, L. A., KLUGER, M. S., ORANGE, J. S. & MAY, M. J. 2008. Lymphotoxin-alpha 1 beta 2 and LIGHT induce classical and noncanonical NF-kappa B-dependent proinflammatory gene expression in vascular endothelial cells. *J Immunol*, 180, 3467-77.
- MADGE, L. A. & MAY, M. J. 2010. Classical NF-kappaB activation negatively regulates noncanonical NF-kappaB-dependent CXCL12 expression. J Biol Chem, 285, 38069-77.
- MAHLEY, R. W. & RALL, S. C., JR. 2000. Apolipoprotein E: far more than a lipid transport protein. *Annu Rev Genomics Hum Genet*, 1, 507-37.
- MALEK, S., HUXFORD, T. & GHOSH, G. 1998. Ikappa Balpha functions through direct contacts with the nuclear localization signals and the DNA binding sequences of NF-kappaB. *J Biol Chem*, 273, 25427-35.
- MARAIS, R., WYNNE, J. & TREISMAN, R. 1993. The SRF accessory protein Elk-1 contains a growth factor-regulated transcriptional activation domain. *Cell*, 73, 381-93.
- MARLIN, S. D. & SPRINGER, T. A. 1987. Purified intercellular adhesion molecule-1 (ICAM-1) is a ligand for lymphocyte function-associated antigen 1 (LFA-1). *Cell*, 51, 813-9.
- MARUYAMA, T., FUKUSHIMA, H., NAKAO, K., SHIN, M., YASUDA, H., WEIH, F., DOI, T., AOKI, K., ALLES, N., OHYA, K., HOSOKAWA, R. & JIMI, E. 2010. Processing of the NF-kappa B2 precursor p100 to p52 is critical for RANKL-induced osteoclast differentiation. *J Bone Miner Res*, 25, 1058-67.

- MASSA, P. E., LI, X., HANIDU, A., SIAMAS, J., PARIALI, M., PAREJA, J., SAVITT, A. G., CATRON, K. M., LI, J. & MARCU, K. B. 2005. Gene expression profiling in conjunction with physiological rescues of IKKalpha-null cells with wild type or mutant IKKalpha reveals distinct classes of IKKalpha/NF-kappaB-dependent genes. J Biol Chem, 280, 14057-69.
- MAVRIA, G., VERCOULEN, Y., YEO, M., PATERSON, H., KARASARIDES, M., MARAIS, R., BIRD, D. & MARSHALL, C. J. 2006. ERK-MAPK signaling opposes Rho-kinase to promote endothelial cell survival and sprouting during angiogenesis. *Cancer Cell*, 9, 33-44.
- MAY, M. J., D'ACQUISTO, F., MADGE, L. A., GLOCKNER, J., POBER, J. S. & GHOSH,S. 2000. Selective inhibition of NF-kappaB activation by a peptide that blocks the interaction of NEMO with the IkappaB kinase complex. *Science*, 289, 1550-4.
- MAY, M. J., MARIENFELD, R. B. & GHOSH, S. 2002. Characterization of the Ikappa Bkinase NEMO binding domain. *J Biol Chem*, 277, 45992-6000.
- MCGORISK, G. M. & TREASURE, C. B. 1996. Endothelial dysfunction in coronary heart disease. *Curr Opin Cardiol*, 11, 341-50.
- MEILER, S. E., HUNG, R. R., GERSZTEN, R. E., GIANETTI, J., LI, L., MATSUI, T., GIMBRONE, M. A., JR. & ROSENZWEIG, A. 2002. Endothelial IKK beta signaling is required for monocyte adhesion under laminar flow conditions. *J Mol Cell Cardiol*, 34, 349-59.
- MERCURIO, F., DIDONATO, J., ROSETTE, C. & KARIN, M. 1992. Molecular cloning and characterization of a novel Rel/NF-kappa B family member displaying structural and functional homology to NF-kappa B p50/p105. *DNA Cell Biol*, 11, 523-37.
- MERCURIO, F., DIDONATO, J. A., ROSETTE, C. & KARIN, M. 1993. p105 and p98 precursor proteins play an active role in NF-kappa B-mediated signal transduction. *Genes Dev*, 7, 705-18.
- MOLITOR, J. A., WALKER, W. H., DOERRE, S., BALLARD, D. W. & GREENE, W. C. 1990. NF-kappa B: a family of inducible and differentially expressed enhancerbinding proteins in human T cells. *Proc Natl Acad Sci U S A*, 87, 10028-32.
- MOMBOULI, J. V. & VANHOUTTE, P. M. 1999. Endothelial dysfunction: from physiology to therapy. *J Mol Cell Cardiol*, 31, 61-74.
- MONCADA, S. & HIGGS, A. 1993. The L-arginine-nitric oxide pathway. N Engl J Med, 329, 2002-12.
- MORALES-RUIZ, M., FULTON, D., SOWA, G., LANGUINO, L. R., FUJIO, Y., WALSH, K. & SESSA, W. C. 2000. Vascular endothelial growth factor-stimulated actin reorganization and migration of endothelial cells is regulated via the serine/threonine kinase Akt. *Circ Res*, 86, 892-6.
- MORDMULLER, B., KRAPPMANN, D., ESEN, M., WEGENER, E. & SCHEIDEREIT, C. 2003. Lymphotoxin and lipopolysaccharide induce NF-kappaB-p52 generation by a co-translational mechanism. *EMBO Rep*, 4, 82-7.
- MORI, N., YAMADA, Y., IKEDA, S., YAMASAKI, Y., TSUKASAKI, K., TANAKA, Y., TOMONAGA, M., YAMAMOTO, N. & FUJII, M. 2002. Bay 11-7082 inhibits transcription factor NF-kappaB and induces apoptosis of HTLV-I-infected T-cell lines and primary adult T-cell leukemia cells. *Blood*, 100, 1828-34.
- MOROI, M., ZHANG, L., YASUDA, T., VIRMANI, R., GOLD, H. K., FISHMAN, M. C. & HUANG, P. L. 1998. Interaction of genetic deficiency of endothelial nitric oxide, gender, and pregnancy in vascular response to injury in mice. *J Clin Invest*, 101, 1225-32.
- MORTIER, J., MASEREEL, B., REMOUCHAMPS, C., GANEFF, C., PIETTE, J. & FREDERICK, R. 2010. NF-kappaB inducing kinase (NIK) inhibitors: identification of new scaffolds using virtual screening. *Bioorg Med Chem Lett*, 20, 4515-20.
- MOUNTAIN, D. J., SINGH, M., MENON, B. & SINGH, K. 2007. Interleukin-1beta increases expression and activity of matrix metalloproteinase-2 in cardiac microvascular endothelial cells: role of PKCalpha/beta1 and MAPKs. *Am J Physiol Cell Physiol*, 292, C867-75.
- MULLER, J. R. & SIEBENLIST, U. 2003. Lymphotoxin beta receptor induces sequential activation of distinct NF-kappa B factors via separate signaling pathways. *J Biol Chem*, 278, 12006-12.
- MUNOZ-CHAPULI, R., QUESADA, A. R. & ANGEL MEDINA, M. 2004. Angiogenesis and signal transduction in endothelial cells. *Cell Mol Life Sci*, 61, 2224-43.
- MURAKAMI, H., SHINOMIYA, N., KIKUCHI, T., FUJIKAWA, K. & NEMOTO, K. 2005. Differential sensitivity to NO-induced apoptosis between anterior cruciate and medial collateral ligament cells. *J Orthop Sci*, 10, 84-90.
- MYTAR, B., GAWLICKA, M., SZATANEK, R., WOLOSZYN, M., RUGGIERO, I., PIEKARSKA, B. & ZEMBALA, M. 2004. Induction of intracellular cytokine production in human monocytes/macrophages stimulated with ligands of pattern recognition receptors. *Inflamm Res*, 53, 100-6.

- NEURATH, M. F., FUSS, I., SCHURMANN, G., PETTERSSON, S., ARNOLD, K., MULLER-LOBECK, H., STROBER, W., HERFARTH, C. & BUSCHENFELDE, K.
 H. 1998. Cytokine gene transcription by NF-kappa B family members in patients with inflammatory bowel disease. *Ann N Y Acad Sci*, 859, 149-59.
- NEURATH, M. F., PETTERSSON, S., MEYER ZUM BUSCHENFELDE, K. H. & STROBER, W. 1996. Local administration of antisense phosphorothioate oligonucleotides to the p65 subunit of NF-kappa B abrogates established experimental colitis in mice. *Nat Med*, 2, 998-1004.
- NICHOLSON, A. C., FEBBRAIO, M., HAN, J., SILVERSTEIN, R. L. & HAJJAR, D. P. 2000. CD36 in atherosclerosis. The role of a class B macrophage scavenger receptor. *Ann N Y Acad Sci*, 902, 128-31; discussion 131-3.
- NICKLIN, S. A. & BAKER, A. H. 1999. Simple methods for preparing recombinant adenoviruses for high-efficiency transduction of vascular cells. *Methods Mol Med*, 30, 271-83.
- NIGG, E. A. 2001. Mitotic kinases as regulators of cell division and its checkpoints. *Nat Rev Mol Cell Biol*, 2, 21-32.
- NIKITENKO, L. L. 2009. Vascular endothelium in cancer. Cell Tissue Res, 335, 223-40.
- NISHIO, E., KANDA, Y., KUROKI, Y. & WATANABE, Y. 2000. The potentiation of ox-LDL induced apoptosis by inhibition of NF-kappaB. *Life Sci*, 67, 1983-92.
- NOLAN, G. P., GHOSH, S., LIOU, H. C., TEMPST, P. & BALTIMORE, D. 1991. DNA binding and I kappa B inhibition of the cloned p65 subunit of NF-kappa B, a relrelated polypeptide. *Cell*, 64, 961-9.
- NOR, J. E., CHRISTENSEN, J., MOONEY, D. J. & POLVERINI, P. J. 1999. Vascular endothelial growth factor (VEGF)-mediated angiogenesis is associated with enhanced endothelial cell survival and induction of Bcl-2 expression. *Am J Pathol*, 154, 375-84.
- NOVACK, D. V., YIN, L., HAGEN-STAPLETON, A., SCHREIBER, R. D., GOEDDEL, D. V., ROSS, F. P. & TEITELBAUM, S. L. 2003. The IkappaB function of NF-kappaB2 p100 controls stimulated osteoclastogenesis. *J Exp Med*, 198, 771-81.
- O'BRIEN, K. D., MCDONALD, T. O., CHAIT, A., ALLEN, M. D. & ALPERS, C. E. 1996. Neovascular expression of E-selectin, intercellular adhesion molecule-1, and vascular cell adhesion molecule-1 in human atherosclerosis and their relation to intimal leukocyte content. *Circulation*, 93, 672-82.

- O'CONNELL, M. A., BENNETT, B. L., MERCURIO, F., MANNING, A. M. & MACKMAN, N. 1998. Role of IKK1 and IKK2 in lipopolysaccharide signaling in human monocytic cells. *J Biol Chem*, 273, 30410-4.
- OZES, O. N., MAYO, L. D., GUSTIN, J. A., PFEFFER, S. R., PFEFFER, L. M. & DONNER, D. B. 1999. NF-kappaB activation by tumour necrosis factor requires the Akt serine-threonine kinase. *Nature*, 401, 82-5.
- PARHAMI, F., FANG, Z. T., FOGELMAN, A. M., ANDALIBI, A., TERRITO, M. C. & BERLINER, J. A. 1993. Minimally modified low density lipoprotein-induced inflammatory responses in endothelial cells are mediated by cyclic adenosine monophosphate. *J Clin Invest*, 92, 471-8.
- PARRY, G. C. & MACKMAN, N. 1994. A set of inducible genes expressed by activated human monocytic and endothelial cells contain kappa B-like sites that specifically bind c-Rel-p65 heterodimers. *J Biol Chem*, 269, 20823-5.
- PASCERI, V., WU, H. D., WILLERSON, J. T. & YEH, E. T. 2000. Modulation of vascular inflammation in vitro and in vivo by peroxisome proliferator-activated receptor-gamma activators. *Circulation*, 101, 235-8.
- PATEL, S. T. & KENT, K. C. 1998. Risk factors and their role in the diseases of the arterial wall. *Semin Vasc Surg*, 11, 156-68.
- PICOT, J., COOPER, K., BRYANT, J. & CLEGG, A. J. 2011. The clinical effectiveness and cost-effectiveness of bortezomib and thalidomide in combination regimens with an alkylating agent and a corticosteroid for the first-line treatment of multiple myeloma: a systematic review and economic evaluation. *Health Technol Assess*, 15, 1-204.
- PIERCE, J. W., SCHOENLEBER, R., JESMOK, G., BEST, J., MOORE, S. A., COLLINS, T. & GERRITSEN, M. E. 1997. Novel inhibitors of cytokine-induced IkappaBalpha phosphorylation and endothelial cell adhesion molecule expression show antiinflammatory effects in vivo. *J Biol Chem*, 272, 21096-103.
- PLAKSIN, D., BAEUERLE, P. A. & EISENBACH, L. 1993. KBF1 (p50 NF-kappa B homodimer) acts as a repressor of H-2Kb gene expression in metastatic tumor cells. J Exp Med, 177, 1651-62.
- PLOTNIKOV, A., ZEHORAI, E., PROCACCIA, S. & SEGER, R. 2011. The MAPK cascades: signaling components, nuclear roles and mechanisms of nuclear translocation. *Biochim Biophys Acta*, 1813, 1619-33.
- PLUMPE, J., MALEK, N. P., BOCK, C. T., RAKEMANN, T., MANNS, M. P. & TRAUTWEIN, C. 2000. NF-kappaB determines between apoptosis and proliferation

in hepatocytes during liver regeneration. Am J Physiol Gastrointest Liver Physiol, 278, G173-83.

- POLLMAN, M. J., NAUMOVSKI, L. & GIBBONS, G. H. 1999. Endothelial cell apoptosis in capillary network remodeling. *J Cell Physiol*, 178, 359-70.
- POPOV, Y., PATSENKER, E., BAUER, M., NIEDOBITEK, E., SCHULZE-KREBS, A. & SCHUPPAN, D. 2006. Halofuginone induces matrix metalloproteinases in rat hepatic stellate cells via activation of p38 and NFkappaB. *J Biol Chem*, 281, 15090-8.
- PRAJAPATI, S., TU, Z., YAMAMOTO, Y. & GAYNOR, R. B. 2006. IKKalpha regulates the mitotic phase of the cell cycle by modulating Aurora A phosphorylation. *Cell Cycle*, 5, 2371-80.
- PRICE, D. T. & LOSCALZO, J. 1999. Cellular adhesion molecules and atherogenesis. Am J Med, 107, 85-97.
- QI, M. & ELION, E. A. 2005. MAP kinase pathways. J Cell Sci, 118, 3569-72.
- QING, G., YAN, P., QU, Z., LIU, H. & XIAO, G. 2007. Hsp90 regulates processing of NFkappa B2 p100 involving protection of NF-kappa B-inducing kinase (NIK) from autophagy-mediated degradation. *Cell Res*, 17, 520-30.
- RA, H. J. & PARKS, W. C. 2007. Control of matrix metalloproteinase catalytic activity. *Matrix Biol*, 26, 587-96.
- RAHMAN, A., KEFER, J., BANDO, M., NILES, W. D. & MALIK, A. B. 1998. E-selectin expression in human endothelial cells by TNF-alpha-induced oxidant generation and NF-kappaB activation. *Am J Physiol*, 275, L533-44.
- RAJAVASHISTH, T. B., YAMADA, H. & MISHRA, N. K. 1995. Transcriptional activation of the macrophage-colony stimulating factor gene by minimally modified LDL. Involvement of nuclear factor-kappa B. *Arterioscler Thromb Vasc Biol*, 15, 1591-8.
- RANGASWAMI, H., BULBULE, A. & KUNDU, G. C. 2006. Nuclear factor inducing kinase: a key regulator in osteopontin- induced MAPK/IkappaB kinase dependent NF-kappaB-mediated promatrix metalloproteinase-9 activation. *Glycoconj J*, 23, 221-32.
- RAO, D. D., SENZER, N., CLEARY, M. A. & NEMUNAITIS, J. 2009. Comparative assessment of siRNA and shRNA off target effects: what is slowing clinical development. *Cancer Gene Ther*, 16, 807-9.
- RAUERT, H., WICOVSKY, A., MULLER, N., SIEGMUND, D., SPINDLER, V., WASCHKE, J., KNEITZ, C. & WAJANT, H. 2010. Membrane tumor necrosis factor (TNF) induces p100 processing via TNF receptor-2 (TNFR2). J Biol Chem, 285, 7394-404.

- RAY, P., ZHANG, D. H., ELIAS, J. A. & RAY, A. 1995. Cloning of a differentially expressed I kappa B-related protein. *J Biol Chem*, 270, 10680-5.
- RAZANI, B., ZARNEGAR, B., YTTERBERG, A. J., SHIBA, T., DEMPSEY, P. W., WARE, C. F., LOO, J. A. & CHENG, G. 2010. Negative feedback in noncanonical NF-kappaB signaling modulates NIK stability through IKKalpha-mediated phosphorylation. *Sci Signal*, 3, ra41.
- REDDY, S. T., GRIJALVA, V., NG, C., HASSAN, K., HAMA, S., MOTTAHEDEH, R., WADLEIGH, D. J., NAVAB, M. & FOGELMAN, A. M. 2002. Identification of genes induced by oxidized phospholipids in human aortic endothelial cells. *Vascul Pharmacol*, 38, 211-8.
- REMONDINO, A., KWON, S. H., COMMUNAL, C., PIMENTEL, D. R., SAWYER, D. B., SINGH, K. & COLUCCI, W. S. 2003. Beta-adrenergic receptor-stimulated apoptosis in cardiac myocytes is mediated by reactive oxygen species/c-Jun NH2-terminal kinase-dependent activation of the mitochondrial pathway. *Circ Res*, 92, 136-8.
- RICE, N. R., MACKICHAN, M. L. & ISRAEL, A. 1992. The precursor of NF-kappa B p50 has I kappa B-like functions. *Cell*, 71, 243-53.
- RICOTE, M., GARCIA-TUNON, I., BETHENCOURT, F. R., FRAILE, B., PANIAGUA, R. & ROYUELA, M. 2004. Interleukin-1 (IL-1alpha and IL-1beta) and its receptors (IL-1RI, IL-1RII, and IL-1Ra) in prostate carcinoma. *Cancer*, 100, 1388-96.
- ROBBESYN, F., SALVAYRE, R. & NEGRE-SALVAYRE, A. 2004. Dual role of oxidized LDL on the NF-kappaB signaling pathway. *Free Radic Res*, 38, 541-51.
- ROSS, R. 1999. Atherosclerosis--an inflammatory disease. N Engl J Med, 340, 115-26.
- ROSSI, A., KAPAHI, P., NATOLI, G., TAKAHASHI, T., CHEN, Y., KARIN, M. & SANTORO, M. G. 2000. Anti-inflammatory cyclopentenone prostaglandins are direct inhibitors of IkappaB kinase. *Nature*, 403, 103-8.
- ROTHE, M., SARMA, V., DIXIT, V. M. & GOEDDEL, D. V. 1995. TRAF2-mediated activation of NF-kappa B by TNF receptor 2 and CD40. *Science*, 269, 1424-7.
- ROTHWARF, D. M., ZANDI, E., NATOLI, G. & KARIN, M. 1998. IKK-gamma is an essential regulatory subunit of the IkappaB kinase complex. *Nature*, 395, 297-300.
- RUBEN, S. M., DILLON, P. J., SCHRECK, R., HENKEL, T., CHEN, C. H., MAHER, M., BAEUERLE, P. A. & ROSEN, C. A. 1991. Isolation of a rel-related human cDNA that potentially encodes the 65-kD subunit of NF-kappa B. *Science*, 254, 11.
- RUOCCO, M. G., MAEDA, S., PARK, J. M., LAWRENCE, T., HSU, L. C., CAO, Y., SCHETT, G., WAGNER, E. F. & KARIN, M. 2005. I{kappa}B kinase (IKK){beta},

but not IKK{alpha}, is a critical mediator of osteoclast survival and is required for inflammation-induced bone loss. *J Exp Med*, 201, 1677-87.

- RUSHE, M., SILVIAN, L., BIXLER, S., CHEN, L. L., CHEUNG, A., BOWES, S., CUERVO, H., BERKOWITZ, S., ZHENG, T., GUCKIAN, K., PELLEGRINI, M. & LUGOVSKOY, A. 2008. Structure of a NEMO/IKK-associating domain reveals architecture of the interaction site. *Structure*, 16, 798-808.
- RYSECK, R. P., BULL, P., TAKAMIYA, M., BOURS, V., SIEBENLIST, U., DOBRZANSKI, P. & BRAVO, R. 1992. RelB, a new Rel family transcription activator that can interact with p50-NF-kappa B. *Mol Cell Biol*, 12, 674-84.
- SAKAI, M., KOBORI, S., MIYAZAKI, A. & HORIUCHI, S. 2000. Macrophage proliferation in atherosclerosis. *Curr Opin Lipidol*, 11, 503-9.
- SAKURAI, H., CHIBA, H., MIYOSHI, H., SUGITA, T. & TORIUMI, W. 1999. IkappaB kinases phosphorylate NF-kappaB p65 subunit on serine 536 in the transactivation domain. *J Biol Chem*, 274, 30353-6.
- SAKURAI, H., HISADA, Y., UENO, M., SUGIURA, M., KAWASHIMA, K. & SUGITA,
 T. 1996. Activation of transcription factor NF-kappa B in experimental glomerulonephritis in rats. *Biochim Biophys Acta*, 1316, 132-8.
- SALAMEH, A., GALVAGNI, F., BARDELLI, M., BUSSOLINO, F. & OLIVIERO, S. 2005. Direct recruitment of CRK and GRB2 to VEGFR-3 induces proliferation, migration, and survival of endothelial cells through the activation of ERK, AKT, and JNK pathways. *Blood*, 106, 3423-31.
- SALCEDO, R., PONCE, M. L., YOUNG, H. A., WASSERMAN, K., WARD, J. M., KLEINMAN, H. K., OPPENHEIM, J. J. & MURPHY, W. J. 2000. Human endothelial cells express CCR2 and respond to MCP-1: direct role of MCP-1 in angiogenesis and tumor progression. *Blood*, 96, 34-40.
- SANDA, T., ASAMITSU, K., OGURA, H., IIDA, S., UTSUNOMIYA, A., UEDA, R. & OKAMOTO, T. 2006. Induction of cell death in adult T-cell leukemia cells by a novel IkappaB kinase inhibitor. *Leukemia*, 20, 590-8.
- SAREN, P., WELGUS, H. G. & KOVANEN, P. T. 1996. TNF-alpha and IL-1beta selectively induce expression of 92-kDa gelatinase by human macrophages. J Immunol, 157, 4159-65.
- SAXENA, U. & GOLDBERG, I. J. 1994. Endothelial cells and atherosclerosis: lipoprotein metabolism, matrix interactions, and monocyte recruitment. *Curr Opin Lipidol*, 5, 316-22.

- SCHAEFER, E. J., GREGG, R. E., GHISELLI, G., FORTE, T. M., ORDOVAS, J. M., ZECH, L. A. & BREWER, H. B., JR. 1986. Familial apolipoprotein E deficiency. J *Clin Invest*, 78, 1206-19.
- SCHERER, D. C., BROCKMAN, J. A., CHEN, Z., MANIATIS, T. & BALLARD, D. W. 1995. Signal-induced degradation of I kappa B alpha requires site-specific ubiquitination. *Proc Natl Acad Sci U S A*, 92, 11259-63.
- SCHMITZ, M. L. & BAEUERLE, P. A. 1991. The p65 subunit is responsible for the strong transcription activating potential of NF-kappa B. *EMBO J*, 10, 3805-17.
- SCHNEIDER, G., SAUR, D., SIVEKE, J. T., FRITSCH, R., GRETEN, F. R. & SCHMID, R.
 M. 2006. IKKalpha controls p52/RelB at the skp2 gene promoter to regulate G1- to S-phase progression. *EMBO J*, 25, 3801-12.
- SCHOPF, L., SAVINAINEN, A., ANDERSON, K., KUJAWA, J., DUPONT, M., SILVA, M., SIEBERT, E., CHANDRA, S., MORGAN, J., GANGURDE, P., WEN, D., LANE, J., XU, Y., HEPPERLE, M., HARRIMAN, G., OCAIN, T. & JAFFEE, B. 2006. IKKbeta inhibition protects against bone and cartilage destruction in a rat model of rheumatoid arthritis. *Arthritis Rheum*, 54, 3163-73.
- SCHRAMEK, H. 2002. MAP kinases: from intracellular signals to physiology and disease. *News Physiol Sci*, 17, 62-7.
- SCHRECK, R., ALBERMANN, K. & BAEUERLE, P. A. 1992. Nuclear factor kappa B: an oxidative stress-responsive transcription factor of eukaryotic cells (a review). *Free Radic Res Commun*, 17, 221-37.
- SCHRECK, R., RIEBER, P. & BAEUERLE, P. A. 1991. Reactive oxygen intermediates as apparently widely used messengers in the activation of the NF-kappa B transcription factor and HIV-1. *EMBO J*, 10, 2247-58.
- SCHWENKE, D. C. & BEHR, S. R. 1998. Vitamin E combined with selenium inhibits atherosclerosis in hypercholesterolemic rabbits independently of effects on plasma cholesterol concentrations. *Circ Res*, 83, 366-77.
- SCOTT, B. B., ZARATIN, P. F., GILMARTIN, A. G., HANSBURY, M. J., COLOMBO, A., BELPASSO, C., WINKLER, J. D. & JACKSON, J. R. 2005. TNF-alpha modulates angiopoietin-1 expression in rheumatoid synovial fibroblasts via the NF-kappa B signalling pathway. *Biochem Biophys Res Commun*, 328, 409-14.
- SCOTTO D'ABUSCO, A., POLITI, L., GIORDANO, C. & SCANDURRA, R. 2010. A peptidyl-glucosamine derivative affects IKKalpha kinase activity in human chondrocytes. *Arthritis Res Ther*, 12, R18.

- SEN, R. & BALTIMORE, D. 1986. Multiple nuclear factors interact with the immunoglobulin enhancer sequences. *Cell*, 46, 705-16.
- SENFTLEBEN, U., CAO, Y., XIAO, G., GRETEN, F. R., KRAHN, G., BONIZZI, G., CHEN, Y., HU, Y., FONG, A., SUN, S. C. & KARIN, M. 2001. Activation by IKKalpha of a second, evolutionary conserved, NF-kappa B signaling pathway. *Science*, 293, 1495-9.
- SHEMBADE, N., PUJARI, R., HARHAJ, N. S., ABBOTT, D. W. & HARHAJ, E. W. 2011. The kinase IKKalpha inhibits activation of the transcription factor NF-kappaB by phosphorylating the regulatory molecule TAX1BP1. *Nat Immunol*, 12, 834-43.
- SHIN, H. M., KIM, M. H., KIM, B. H., JUNG, S. H., KIM, Y. S., PARK, H. J., HONG, J. T., MIN, K. R. & KIM, Y. 2004. Inhibitory action of novel aromatic diamine compound on lipopolysaccharide-induced nuclear translocation of NF-kappaB without affecting IkappaB degradation. *FEBS Lett*, 571, 50-4.
- SHIRLEY, R. B., KADDOUR-DJEBBAR, I., PATEL, D. M., LAKSHMIKANTHAN, V., LEWIS, R. W. & KUMAR, M. V. 2005. Combination of proteasomal inhibitors lactacystin and MG132 induced synergistic apoptosis in prostate cancer cells. *Neoplasia*, 7, 1104-11.
- SIEBENLIST, U., FRANZOSO, G. & BROWN, K. 1994. Structure, regulation and function of NF-kappa B. *Annu Rev Cell Biol*, 10, 405-55.
- SIL, A. K., MAEDA, S., SANO, Y., ROOP, D. R. & KARIN, M. 2004. IkappaB kinasealpha acts in the epidermis to control skeletal and craniofacial morphogenesis. *Nature*, 428, 660-4.
- SILVERSTEIN, R. L. & FEBBRAIO, M. 2000. CD36 and atherosclerosis. *Curr Opin Lipidol*, 11, 483-91.
- SMOLEN, J. S. & STEINER, G. 2003. Therapeutic strategies for rheumatoid arthritis. *Nat Rev Drug Discov*, 2, 473-88.
- SNYDER, J. G., PREWITT, R., CAMPSEN, J. & BRITT, L. D. 2002. PDTC and Mg132, inhibitors of NF-kappaB, block endotoxin induced vasodilation of isolated rat skeletal muscle arterioles. *Shock*, 17, 304-7.
- SOLT, L. A., MADGE, L. A. & MAY, M. J. 2009. NEMO-binding domains of both IKKalpha and IKKbeta regulate IkappaB kinase complex assembly and classical NFkappaB activation. *J Biol Chem*, 284, 27596-608.
- SOMMERS, C. D., THOMPSON, J. M., GUZOVA, J. A., BONAR, S. L., RADER, R. K., MATHIALAGAN, S., VENKATRAMAN, N., HOLWAY, V. W., KAHN, L. E., HU,

G., GARNER, D. S., HUANG, H. C., CHIANG, P. C., SCHINDLER, J. F., HU, Y., MEYER, D. M. & KISHORE, N. N. 2009. Novel tight-binding inhibitory factorkappaB kinase (IKK-2) inhibitors demonstrate target-specific anti-inflammatory activities in cellular assays and following oral and local delivery in an in vivo model of airway inflammation. *J Pharmacol Exp Ther*, 330, 377-88.

- SONG, L., DONG, W., GAO, M., LI, J., HU, M., GUO, N. & HUANG, C. 2010. A novel role of IKKalpha in the mediation of UVB-induced G0/G1 cell cycle arrest response by suppressing Cyclin D1 expression. *Biochim Biophys Acta*, 1803, 323-32.
- SRINIVASAN, R., ZABUAWALA, T., HUANG, H., ZHANG, J., GULATI, P., FERNANDEZ, S., KARLO, J. C., LANDRETH, G. E., LEONE, G. & OSTROWSKI, M. C. 2009. Erk1 and Erk2 regulate endothelial cell proliferation and migration during mouse embryonic angiogenesis. *PLoS One*, 4, e8283.
- STANKEVICIUS, E., KEVELAITIS, E., VAINORIUS, E. & SIMONSEN, U. 2003. [Role of nitric oxide and other endothelium-derived factors]. *Medicina (Kaunas)*, 39, 333-41.
- STEFANOVSKY, V. Y., PELLETIER, G., HANNAN, R., GAGNON-KUGLER, T., ROTHBLUM, L. I. & MOSS, T. 2001. An immediate response of ribosomal transcription to growth factor stimulation in mammals is mediated by ERK phosphorylation of UBF. *Mol Cell*, 8, 1063-73.
- STEHLIK, C., DE MARTIN, R., BINDER, B. R. & LIPP, J. 1998a. Cytokine induced expression of porcine inhibitor of apoptosis protein (iap) family member is regulated by NF-kappa B. *Biochem Biophys Res Commun*, 243, 827-32.
- STEHLIK, C., DE MARTIN, R., KUMABASHIRI, I., SCHMID, J. A., BINDER, B. R. & LIPP, J. 1998b. Nuclear factor (NF)-kappaB-regulated X-chromosome-linked iap gene expression protects endothelial cells from tumor necrosis factor alpha-induced apoptosis. J Exp Med, 188, 211-6.
- STRATMAN, A. N., DAVIS, M. J. & DAVIS, G. E. 2011. VEGF and FGF prime vascular tube morphogenesis and sprouting directed by hematopoietic stem cell cytokines. *Blood*, 117, 3709-19.
- SUNWOO, J. B., CHEN, Z., DONG, G., YEH, N., CROWL BANCROFT, C., SAUSVILLE, E., ADAMS, J., ELLIOTT, P. & VAN WAES, C. 2001. Novel proteasome inhibitor PS-341 inhibits activation of nuclear factor-kappa B, cell survival, tumor growth, and angiogenesis in squamous cell carcinoma. *Clin Cancer Res*, 7, 1419-28.

- SUVORAVA, T., LAUER, N. & KOJDA, G. 2004. Physical inactivity causes endothelial dysfunction in healthy young mice. *J Am Coll Cardiol*, 44, 1320-7.
- SWIATKOWSKA, M., SZEMRAJ, J. & CIERNIEWSKI, C. S. 2005. Induction of PAI-1 expression by tumor necrosis factor alpha in endothelial cells is mediated by its responsive element located in the 4G/5G site. *FEBS J*, 272, 5821-31.
- TAK, P. P., GERLAG, D. M., AUPPERLE, K. R., VAN DE GEEST, D. A., OVERBEEK, M., BENNETT, B. L., BOYLE, D. L., MANNING, A. M. & FIRESTEIN, G. S. 2001. Inhibitor of nuclear factor kappaB kinase beta is a key regulator of synovial inflammation. *Arthritis Rheum*, 44, 1897-907.
- TAKAESU, G., SURABHI, R. M., PARK, K. J., NINOMIYA-TSUJI, J., MATSUMOTO, K. & GAYNOR, R. B. 2003. TAK1 is critical for IkappaB kinase-mediated activation of the NF-kappaB pathway. *J Mol Biol*, 326, 105-15.
- TAKEDA, K., TAKEUCHI, O., TSUJIMURA, T., ITAMI, S., ADACHI, O., KAWAI, T., SANJO, H., YOSHIKAWA, K., TERADA, N. & AKIRA, S. 1999. Limb and skin abnormalities in mice lacking IKKalpha. *Science*, 284, 313-6.
- TANAKA, E., SHIMOKAWA, H., KAMIUNETEN, H., ETO, Y., MATSUMOTO, Y., MORISHIGE, K., KOIKE, G., YOSHINAGA, M., EGASHIRA, K., TOKUNAGA, O., SHIOMI, M. & TAKESHITA, A. 2003. Disparity of MCP-1 mRNA and protein expressions between the carotid artery and the aorta in WHHL rabbits: one aspect involved in the regional difference in atherosclerosis. *Arterioscler Thromb Vasc Biol*, 23, 244-50.
- TANAKA, M., FUENTES, M. E., YAMAGUCHI, K., DURNIN, M. H., DALRYMPLE, S. A., HARDY, K. L. & GOEDDEL, D. V. 1999. Embryonic lethality, liver degeneration, and impaired NF-kappa B activation in IKK-beta-deficient mice. *Immunity*, 10, 421-9.
- TAS, S. W., DE JONG, E. C., HAJJI, N., MAY, M. J., GHOSH, S., VERVOORDELDONK, M. J. & TAK, P. P. 2005. Selective inhibition of NF-kappaB in dendritic cells by the NEMO-binding domain peptide blocks maturation and prevents T cell proliferation and polarization. *Eur J Immunol*, 35, 1164-74.
- TATETSU, H., OKUNO, Y., NAKAMURA, M., MATSUNO, F., SONOKI, T., TANIGUCHI, I., UNEDA, S., UMEZAWA, K., MITSUYA, H. & HATA, H. 2005.
 Dehydroxymethylepoxyquinomicin, a novel nuclear factor-kappaB inhibitor, induces apoptosis in multiple myeloma cells in an IkappaBalpha-independent manner. *Mol Cancer Ther*, 4, 1114-20.

TEITELBAUM, S. L. 2000. Bone resorption by osteoclasts. Science, 289, 1504-8.

- THOMPSON, J. E., PHILLIPS, R. J., ERDJUMENT-BROMAGE, H., TEMPST, P. & GHOSH, S. 1995. I kappa B-beta regulates the persistent response in a biphasic activation of NF-kappa B. *Cell*, 80, 573-82.
- TORRIE, L. J., MACKENZIE, C. J., PAUL, A. & PLEVIN, R. 2001. Hydrogen peroxidemediated inhibition of lipopolysaccharide-stimulated inhibitory kappa B kinase activity in rat aortic smooth muscle cells. *Br J Pharmacol*, 134, 393-401.
- TOWBIN, H., STAEHELIN, T. & GORDON, J. 1979. Electrophoretic transfer of proteins from polyacrylamide gels to nitrocellulose sheets: procedure and some applications. *Proc Natl Acad Sci U S A*, 76, 4350-4.
- TSAO, P. W., SUZUKI, T., TOTSUKA, R., MURATA, T., TAKAGI, T., OHMACHI, Y., FUJIMURA, H. & TAKATA, I. 1997. The effect of dexamethasone on the expression of activated NF-kappa B in adjuvant arthritis. *Clin Immunol Immunopathol*, 83, 173-8.
- TU, Z., KELLEY, V. R., COLLINS, T. & LEE, F. S. 2001. I kappa B kinase is critical for TNF-alpha-induced VCAM1 gene expression in renal tubular epithelial cells. J Immunol, 166, 6839-46.
- TUNG, W. H., SUN, C. C., HSIEH, H. L., WANG, S. W., HORNG, J. T. & YANG, C. M. 2007. EV71 induces VCAM-1 expression via PDGF receptor, PI3-K/Akt, p38 MAPK, JNK and NF-kappaB in vascular smooth muscle cells. *Cell Signal*, 19, 2127-37.
- TURNER, D. J., ALAISH, S. M., ZOU, T., RAO, J. N., WANG, J. Y. & STRAUCH, E. D. 2007. Bile salts induce resistance to apoptosis through NF-kappaB-mediated XIAP expression. *Ann Surg*, 245, 415-25.
- ULFHAMMER, E., LARSSON, P., KARLSSON, L., HRAFNKELSDOTTIR, T., BOKAREWA, M., TARKOWSKI, A. & JERN, S. 2006. TNF-alpha mediated suppression of tissue type plasminogen activator expression in vascular endothelial cells is NF-kappaB- and p38 MAPK-dependent. *J Thromb Haemost*, 4, 1781-9.
- URBAN, M. B., SCHRECK, R. & BAEUERLE, P. A. 1991. NF-kappa B contacts DNA by a heterodimer of the p50 and p65 subunit. *EMBO J*, 10, 1817-25.
- VAN BERLO, D., KNAAPEN, A. M., VAN SCHOOTEN, F. J., SCHINS, R. P. & ALBRECHT, C. 2010. NF-kappaB dependent and independent mechanisms of quartz-induced proinflammatory activation of lung epithelial cells. *Part Fibre Toxicol*, 7, 13.

- VAN DE STOLPE, A., CALDENHOVEN, E., STADE, B. G., KOENDERMAN, L., RAAIJMAKERS, J. A., JOHNSON, J. P. & VAN DER SAAG, P. T. 1994. 12-Otetradecanoylphorbol-13-acetate- and tumor necrosis factor alpha-mediated induction of intercellular adhesion molecule-1 is inhibited by dexamethasone. Functional analysis of the human intercellular adhesion molecular-1 promoter. *J Biol Chem*, 269, 6185-92.
- VAN DEN BRINK, G. R., TEN KATE, F. J., PONSIOEN, C. Y., RIVE, M. M., TYTGAT, G. N., VAN DEVENTER, S. J. & PEPPELENBOSCH, M. P. 2000. Expression and activation of NF-kappa B in the antrum of the human stomach. *J Immunol*, 164, 3353-9.
- VAN DER WAL, A. C., BECKER, A. E., VAN DER LOOS, C. M. & DAS, P. K. 1994. Site of intimal rupture or erosion of thrombosed coronary atherosclerotic plaques is characterized by an inflammatory process irrespective of the dominant plaque morphology. *Circulation*, 89, 36-44.
- VANE, J. R., ANGGARD, E. E. & BOTTING, R. M. 1990. Regulatory functions of the vascular endothelium. *N Engl J Med*, 323, 27-36.
- VERHASSELT, V., VANDEN BERGHE, W., VANDERHEYDE, N., WILLEMS, F., HAEGEMAN, G. & GOLDMAN, M. 1999. N-acetyl-L-cysteine inhibits primary human T cell responses at the dendritic cell level: association with NF-kappaB inhibition. *J Immunol*, 162, 2569-74.
- VOGEL, R. A. 1997. Coronary risk factors, endothelial function, and atherosclerosis: a review. *Clin Cardiol*, 20, 426-32.
- WANG, C., DENG, L., HONG, M., AKKARAJU, G. R., INOUE, J. & CHEN, Z. J. 2001. TAK1 is a ubiquitin-dependent kinase of MKK and IKK. *Nature*, 412, 346-51.
- WANG, H. Y., QU, P., LU, S., LIU, M. & JIANG, H. 2005. [The effect of TLR4/NF-kappaB activation and LOX-1 on monocyte adhesion to endothelium]. *Zhonghua Xin Xue Guan Bing Za Zhi*, 33, 827-31.
- WANG, W., MCLEOD, H. L. & CASSIDY, J. 2003. Disulfiram-mediated inhibition of NFkappaB activity enhances cytotoxicity of 5-fluorouracil in human colorectal cancer cell lines. *Int J Cancer*, 104, 504-11.
- WEIH, F., CARRASCO, D., DURHAM, S. K., BARTON, D. S., RIZZO, C. A., RYSECK,
 R. P., LIRA, S. A. & BRAVO, R. 1995. Multiorgan inflammation and hematopoietic abnormalities in mice with a targeted disruption of RelB, a member of the NF-kappa B/Rel family. *Cell*, 80, 331-40.

- WHARRY, C. E., HAINES, K. M., CARROLL, R. G. & MAY, M. J. 2009. Constitutive non-canonical NFkappaB signaling in pancreatic cancer cells. *Cancer Biol Ther*, 8, 1567-76.
- WILCOX, J. N., SUBRAMANIAN, R. R., SUNDELL, C. L., TRACEY, W. R., POLLOCK, J. S., HARRISON, D. G. & MARSDEN, P. A. 1997. Expression of multiple isoforms of nitric oxide synthase in normal and atherosclerotic vessels. *Arterioscler Thromb Vasc Biol*, 17, 2479-88.
- WILCZYNSKA, K. M., GOPALAN, S. M., BUGNO, M., KASZA, A., KONIK, B. S., BRYAN, L., WRIGHT, S., GRISWOLD-PRENNER, I. & KORDULA, T. 2006. A novel mechanism of tissue inhibitor of metalloproteinases-1 activation by interleukin-1 in primary human astrocytes. *J Biol Chem*, 281, 34955-64.
- WILSON, S. H., BEST, P. J., EDWARDS, W. D., HOLMES, D. R., JR., CARLSON, P. J., CELERMAJER, D. S. & LERMAN, A. 2002. Nuclear factor-kappaB immunoreactivity is present in human coronary plaque and enhanced in patients with unstable angina pectoris. *Atherosclerosis*, 160, 147-53.
- WONG, B. R., JOSIEN, R. & CHOI, Y. 1999. TRANCE is a TNF family member that regulates dendritic cell and osteoclast function. *J Leukoc Biol*, 65, 715-24.
- WORONICZ, J. D., GAO, X., CAO, Z., ROTHE, M. & GOEDDEL, D. V. 1997. IkappaB kinase-beta: NF-kappaB activation and complex formation with IkappaB kinase-alpha and NIK. *Science*, 278, 866-9.
- WU, C. J. & ASHWELL, J. D. 2008. NEMO recognition of ubiquitinated Bcl10 is required for T cell receptor-mediated NF-kappaB activation. *Proc Natl Acad Sci U S A*, 105, 3023-8.
- WU, L. W., MAYO, L. D., DUNBAR, J. D., KESSLER, K. M., BAERWALD, M. R., JAFFE, E. A., WANG, D., WARREN, R. S. & DONNER, D. B. 2000. Utilization of distinct signaling pathways by receptors for vascular endothelial cell growth factor and other mitogens in the induction of endothelial cell proliferation. *J Biol Chem*, 275, 5096-103.
- XIA, P., GAMBLE, J. R., RYE, K. A., WANG, L., HII, C. S., COCKERILL, P., KHEW-GOODALL, Y., BERT, A. G., BARTER, P. J. & VADAS, M. A. 1998. Tumor necrosis factor-alpha induces adhesion molecule expression through the sphingosine kinase pathway. *Proc Natl Acad Sci U S A*, 95, 14196-201.
- XIA, Y., CHEN, S., WANG, Y., MACKMAN, N., KU, G., LO, D. & FENG, L. 1999. RelB modulation of IkappaBalpha stability as a mechanism of transcription suppression of

interleukin-1alpha (IL-1alpha), IL-1beta, and tumor necrosis factor alpha in fibroblasts. *Mol Cell Biol*, 19, 7688-96.

- XIA, Y., PAUZA, M. E., FENG, L. & LO, D. 1997. RelB regulation of chemokine expression modulates local inflammation. *Am J Pathol*, 151, 375-87.
- XIAO, G., FONG, A. & SUN, S. C. 2004. Induction of p100 processing by NF-kappaBinducing kinase involves docking IkappaB kinase alpha (IKKalpha) to p100 and IKKalpha-mediated phosphorylation. *J Biol Chem*, 279, 30099-105.
- XIAO, G., HARHAJ, E. W. & SUN, S. C. 2001. NF-kappaB-inducing kinase regulates the processing of NF-kappaB2 p100. *Mol Cell*, 7, 401-9.
- XIAO, Q., HSU, C. Y., CHEN, H., MA, X., XU, J. & LEE, J. M. 2005. Characterization of cis-regulatory elements of the vascular endothelial growth inhibitor gene promoter. *Biochem J*, 388, 913-20.
- YAMAMOTO, K., ICHIJO, H. & KORSMEYER, S. J. 1999. BCL-2 is phosphorylated and inactivated by an ASK1/Jun N-terminal protein kinase pathway normally activated at G(2)/M. *Mol Cell Biol*, 19, 8469-78.
- YAMAMOTO, Y., VERMA, U. N., PRAJAPATI, S., KWAK, Y. T. & GAYNOR, R. B. 2003. Histone H3 phosphorylation by IKK-alpha is critical for cytokine-induced gene expression. *Nature*, 423, 655-9.
- YAMAOKA, S., COURTOIS, G., BESSIA, C., WHITESIDE, S. T., WEIL, R., AGOU, F., KIRK, H. E., KAY, R. J. & ISRAEL, A. 1998. Complementation cloning of NEMO, a component of the IkappaB kinase complex essential for NF-kappaB activation. *Cell*, 93, 1231-40.
- YANG, F., TANG, E., GUAN, K. & WANG, C. Y. 2003. IKK beta plays an essential role in the phosphorylation of RelA/p65 on serine 536 induced by lipopolysaccharide. J Immunol, 170, 5630-5.
- YANG, S. R., VALVO, S., YAO, H., KODE, A., RAJENDRASOZHAN, S., EDIRISINGHE, I., CAITO, S., ADENUGA, D., HENRY, R., FROMM, G., MAGGIRWAR, S., LI, J. D., BULGER, M. & RAHMAN, I. 2008. IKK alpha causes chromatin modification on pro-inflammatory genes by cigarette smoke in mouse lung. *Am J Respir Cell Mol Biol*, 38, 689-98.
- YANG, Y. L., CHUANG, L. Y., GUH, J. Y., LIU, S. F., HUNG, M. Y., LIAO, T. N. & HUANG, Y. L. 2004. Thrombospondin-1 mediates distal tubule hypertrophy induced by glycated albumin. *Biochem J*, 379, 89-97.

- YING, H., ELPEK, K. G., VINJAMOORI, A., ZIMMERMAN, S. M., CHU, G. C., YAN,
 H., FLETCHER-SANANIKONE, E., ZHANG, H., LIU, Y., WANG, W., REN, X.,
 ZHENG, H., KIMMELMAN, A. C., PAIK, J. H., LIM, C., PERRY, S. R., JIANG, S.,
 MALINN, B., PROTOPOPOV, A., COLLA, S., XIAO, Y., HEZEL, A. F.,
 BARDEESY, N., TURLEY, S. J., WANG, Y. A., CHIN, L., THAYER, S. P. &
 DEPINHO, R. A. 2011. Pten is a major tumor suppressor in pancreatic ductal
 adenocarcinoma and regulates an NF-kappaB-cytokine network. *Cancer Discov*, 1, 158-169.
- ZABEL, U. & BAEUERLE, P. A. 1990. Purified human I kappa B can rapidly dissociate the complex of the NF-kappa B transcription factor with its cognate DNA. *Cell*, 61, 255-65.
- ZABEL, U., SCHRECK, R. & BAEUERLE, P. A. 1991. DNA binding of purified transcription factor NF-kappa B. Affinity, specificity, Zn2+ dependence, and differential half-site recognition. *J Biol Chem*, 266, 252-60.
- ZANDI, E., ROTHWARF, D. M., DELHASE, M., HAYAKAWA, M. & KARIN, M. 1997. The IkappaB kinase complex (IKK) contains two kinase subunits, IKKalpha and IKKbeta, necessary for IkappaB phosphorylation and NF-kappaB activation. *Cell*, 91, 243-52.
- ZEIHER, A. M. 1996. Endothelial vasodilator dysfunction: pathogenetic link to myocardial ischaemia or epiphenomenon? *Lancet*, 348 Suppl 1, s10-2.
- ZERNECKE, A., ERL, W., FRAEMOHS, L., LIETZ, M. & WEBER, C. 2003. Suppression of endothelial adhesion molecule up-regulation with cyclopentenone prostaglandins is dissociated from IkappaB-alpha kinase inhibition and cell death induction. *FASEB J*, 17, 1099-101.
- ZHANG, F., YU, W., HARGROVE, J. L., GREENSPAN, P., DEAN, R. G., TAYLOR, E.W. & HARTLE, D. K. 2002. Inhibition of TNF-alpha induced ICAM-1, VCAM-1 and E-selectin expression by selenium. *Atherosclerosis*, 161, 381-6.
- ZHANG, W. J. & FREI, B. 2002. Albumin selectively inhibits TNF alpha-induced expression of vascular cell adhesion molecule-1 in human aortic endothelial cells. *Cardiovasc Res*, 55, 820-9.
- ZHENG, B., ZHOU, R., GONG, Y., YANG, X. & SHAN, Q. 2011. Proteasome inhibitor bortezomib overcomes P-gp-mediated multidrug resistance in resistant leukemic cell lines. *Int J Lab Hematol.*

- ZHENG, C. F. & GUAN, K. L. 1993. Cloning and characterization of two distinct human extracellular signal-regulated kinase activator kinases, MEK1 and MEK2. J Biol Chem, 268, 11435-9.
- ZHU, Y., LIAO, H. L., LIN, J. H., VERNA, L. & STEMERMAN, M. B. 1999. Low-density lipoprotein augments interleukin-1-induced vascular adhesion molecule expression in human endothelial cells. *Atherosclerosis*, 144, 357-65.
- ZIEGLER-HEITBROCK, H. W., STERNSDORF, T., LIESE, J., BELOHRADSKY, B., WEBER, C., WEDEL, A., SCHRECK, R., BAUERLE, P. & STROBEL, M. 1993. Pyrrolidine dithiocarbamate inhibits NF-kappa B mobilization and TNF production in human monocytes. *J Immunol*, 151, 6986-93.